Max-Planck-Institut für Mathematik Bonn

On the Kontsevich geometry of the combinatorial Teichmüller space

by

Jørgen Ellegaard Andersen Gaëtan Borot Séverin Charbonnier Alessandro Giacchetto Danilo Lewański Campbell Wheeler



Max-Planck-Institut für Mathematik Preprint Series 2020 (55)

Date of submission: September 22, 2020

On the Kontsevich geometry of the combinatorial Teichmüller space

by

Jørgen Ellegaard Andersen Gaëtan Borot Séverin Charbonnier Alessandro Giacchetto Danilo Lewański Campbell Wheeler

Max-Planck-Institut für Mathematik Vivatsgasse 7 53111 Bonn Germany Center for Quantum Mathematics Danish Institute for Advanced Study Campusvej 55 5230 Odense Denmark

Institut für Mathematik Humboldt-Universität zu Berlin Rudower Chaussee 25 12489 Berlin Germany

Université Paris-Saclay, CNRS, CEA Institut du Physique Théorique 91191 Gif-sur-Yvette France

Institut des Hautes Études Scientifiques le Bois-Marie 35 route de Chartres 91440 Bures-sur-Yvette France MPIM 20-55

ON THE KONTSEVICH GEOMETRY OF THE COMBINATORIAL TEICHMÜLLER SPACE

Jørgen Ellegaard Andersen*, Gaëtan Borot *[†], Séverin Charbonnier, Alessandro Giacchetto*, Danilo Lewański *^{‡§}, Campbell Wheeler*

Abstract

We study the combinatorial Teichmüller space and construct on it global coordinates, analogous to the Fenchel–Nielsen coordinates on the ordinary Teichmüller space. We prove that these coordinates form an atlas with piecewise linear transition functions, and constitute global Darboux coordinates for the Kontsevich symplectic structure on top-dimensional cells.

We then set up the geometric recursion in the sense of Andersen–Borot–Orantin adapted to the combinatorial setting, which naturally produces mapping class group invariant functions on the combinatorial Teichmüller spaces. We establish a combinatorial analogue of the Mirzakhani–McShane identity fitting this framework.

As applications, we obtain geometric proofs of Witten conjecture/Kontsevich theorem (Virasoro constraints for ψ -classes intersections) and of Norbury's topological recursion for the lattice point count in the combinatorial moduli spaces. These proofs arise now as part of a unified theory and proceed in perfect parallel to Mirzakhani's proof of topological recursion for the Weil–Petersson volumes.

We move on to the study of the spine construction and the associated rescaling flow on the Teichmüller space. We strengthen former results of Mondello and Do on the convergence of this flow. In particular, we prove convergence of hyperbolic Fenchel–Nielsen coordinates to the combinatorial ones with some uniformity. This allows us to effectively carry natural constructions on the Teichmüller space to their analogues in the combinatorial spaces. For instance, we obtain the piecewise linear structure on the combinatorial Teichmüller space as the limit of the smooth structure on the Teichmüller space.

To conclude, we provide further applications to the enumerative geometry of multicurves, Masur–Veech volumes and measured foliations in the combinatorial setting.

^{**}Center for Quantum Mathematics, Danish Institute for Advanced Study, Campusvej 55, 5230 Odense, Denmark.
*(on leave from) Max Planck Institut für Mathematik, Vivatsgasse 7, 53111 Bonn, Germany

[†]Institut für Mathematik, Humboldt Universität zu Berlin, Rudower Chaussee 25, 12489 Berlin, Germany.

[‡]Université Paris-Saclay, CNRS, CEA, Institut de Physique Théorique, 91191 Gif-sur-Yvette, France.

[§]Institut des Hautes Études Scientifiques, le Bois-Marie, 35 route de Chartres, 91440 Bures-sur-Yvette, France.

Contents

1	Introduction	3
2	Topology of combinatorial Teichmüller spaces 2.1 Combinatorial moduli spaces	 11 13 18 24 26 30
3	The symplectic structure 3.1 Kontsevich's form	41 41 43 46
4	Functions from geometric recursion 4.1 Definition of the geometric recursion 4.2 (Discrete) integration and topological recursion 4.3 Remark: inducing D from C 4.4 Mirzakhani–McShane identities 4.5 Combinatorial length statistics of multicurves 4.6 Kontsevich amplitudes for hyperbolic GR	50 53 60 60 63 65
5	Rescaling flow: from hyperbolic to combinatorial geometry5.1Convergence of lengths and twists	70 70 75 77 81
6	Enumerative geometry of multicurves6.1Parametrising multicurves6.2Unit balls in measured foliations: the hyperbolic case6.3Unit balls in measured foliations: the combinatorial case6.4Geometric recursion for combinatorial length statistics of multicurves6.5Combinatorial formulae for Masur-Veech volumes	 83 83 85 86 89 91
Α	Topology on the combinatorial Teichmüller spaces	93
B	Examples of cutting, gluing, and combinatorial Fenchel–Nielsen coordinates	94
C	Integral points in the moduli space and factors of 2	101

1 Introduction

The Teichmüller space \mathfrak{T}_{Σ} of a bordered surface Σ of genus g with n boundary components can be described in many ways. Although they all lead to the same topological space, they highlight different geometries. Its description via hyperbolic structures leads to the Weil–Petersson geometry: there exists a natural mapping class group invariant symplectic form ω_{WP} for which Fenchel–Nielsen coordinates turn out to be, by a formula of Wolpert [44], global Darboux coordinates. The identification of the Teichmüller space with a component of the PSL(2, \mathbb{R}) character variety, on the other hand, gives access to the transformation of the Fenchel–Nielsen coordinates under change of pants decompositions (*cf.* Okai's formulae in [37]). Furthermore, Mirzakhani–McShane identities [32] give a partition of unity on \mathfrak{T}_{Σ} that is recursive in the topology of Σ as established in [2]. The integration of this recursive formula with respect to the Weil–Petersson volume form recovers, again by Wolpert's formula, the topological recursion in the sense of Eynard and Orantin for the Weil–Petersson volumes of the moduli space $\mathfrak{M}_{g,n}(L)$ of bordered surfaces of genus g with fixed boundary lengths $L \in \mathbb{R}^n_+$. A certain class of such recursive formulae has been recently extended to handle statistics of multicurves [2]. In particular, a recursion for Masur–Veech volumes of the top stratum of the moduli space of quadratic differentials [1] has been derived.

The topological space $\mathcal{M}_{g,n}(L)$ (resp. its orbifold universal cover $\mathcal{T}_{\Sigma}(L)$) can also be described in a combinatorial way as a space of (resp. isotopy classes of embedded) metric ribbon graphs. Such a description turned out to provide a very useful point of view in the study of the topology of the moduli spaces, see *e.g.* [23, 24, 39]. An important result in this direction is provided by Kontsevich [28] as part of the proof of Witten's conjecture, in which he equips the combinatorial moduli space with a 2-form ω_{K} , symplectic away from certain strata of positive codimension.

The purpose of this work is to develop the combinatorial geometry of T_{Σ} associated with ω_{K} in complete parallel to its well-known Weil–Petersson geometry. In particular, by means of the measured foliation description of the combinatorial Teichmüller spaces, we construct the combinatorial analogue of Fenchel–Nielsen coordinates. We then establish the combinatorial analogue of Wolpert formula, of Okai's formulae, and of Mirzakhani–McShane identities. Finally, we strengthen previously known results of Mondello and Do by comparing the combinatorial geometry with the hyperbolic geometry in the limit of large boundary lengths, so that it becomes possible to show convergence of other interesting quantities under this flow.

Harvesting the fruits of these results and of the general formalism of geometric recursion of [2], we obtain geometric proofs of Witten's conjecture/Kontsevich theorem and of Norbury's topological recursion for the lattice point count. This approach – in which the combinatorial analogue of Fenchel–Nielsen coordinates and of Wolpert formula are essential tools – revisits the proofs of these results given respectively in [5] and [36] in a way that is now uniform and parallel to Mirzakhani's proof of topological recursion for the Weil–Petersson volume. We also get access to (asymptotic) statistics of multicurves with respect to their combinatorial length, and establish a link between Masur–Veech volumes and combinatorial geometry.

In all these examples, we see in action the following principle: that topological recursion formulae for integrals on moduli spaces are more conveniently derived by geometric recursion formulae for the integrand themselves, lifted to functions or forms on Teichmüller spaces.

Let us turn to a more detailed account of the main results and of the general organisation of the paper.

Combinatorial spaces and their geometry. We recall the definition of the moduli space of metric ribbon graphs $\mathcal{M}_{g,n}^{\text{comb}}(L)$ (Section 2.1) and of the associated combinatorial Teichmüller spaces $\mathcal{T}_{\Sigma}^{\text{comb}}(L)$ (Section 2.2). In a nutshell, for a connected bordered surface Σ of genus g with n boundaries,

$$\mathfrak{T}^{comb}_{\Sigma}(L) = \left\{ \begin{array}{l} \text{embedded metric ribbon graphs of genus g with n faces} \\ \text{of lengths } L = (L_1, \ldots, L_n) \text{ that are retract of } \Sigma \end{array} \right\} \middle/ \text{isotopy }.$$



Figure 1: Two examples of combinatorial structures on a sphere with four boundary components.

Elements of $\mathcal{T}_{\Sigma}^{comb}(L)$ are called combinatorial structures and $\mathcal{M}_{g,n}^{comb}(L)$ is the quotient of $\mathcal{T}_{\Sigma}^{comb}(L)$ by the pure mapping class group of Σ , which depends only on the genus and the number of boundary components of the surface.

As a set, the space $\mathfrak{T}_{\Sigma}^{comb}$ is given by the union of $\mathfrak{T}_{\Sigma}^{comb}(L)$ over $L \in \mathbb{R}^{n}_{+}$. It is in fact nothing but the (geometric realisation of the) metrised proper arc complex on Σ [35] and it is homeomorphic to the usual Teichmüller space \mathfrak{T}_{Σ} parametrising hyperbolic metrics on Σ making the boundaries geodesic. Furthermore, it can be seen as a certain subset of MF_{Σ}^{\star} , the space of measured foliations on bordered surfaces introduced in [22]. Though the idea of describing $\mathfrak{T}_{\Sigma}^{comb}$ via measured foliations is not new, it turns out to be key for many of our results.

Many aspects of the geometry of hyperbolic structures have an analogue for combinatorial structures. First of all, we can define the length of simple closed curves with respect to a combinatorial structure, and this gives an alternative description of the topology on $\mathcal{T}_{\Sigma}^{comb}$ as follows. Let S_{Σ} be the set of simple closed curves – including boundary components – on Σ . Our first result is Theorem 2.20 in the text, restated here.

Theorem A.1. The combinatorial length map $\ell_* : \mathfrak{T}_{\Sigma}^{comb} \to \mathbb{R}_+^{S_{\Sigma}}$ is a homeomorphism onto its image.

It is possible to cut combinatorial structures along simple closed curves, and it is possible to (self-)glue combinatorial structures along boundary components of the same lengths, once a choice of marked points on the boundary components which are being glued has been made and the value of a generic twist $\tau \in \mathbb{R}$ has been chosen (Section 2.4). The main difference with the hyperbolic world is in the gluing: for some values of the twist τ , it is not possible to glue the combinatorial structures; in the measured foliation description this correspond to the creation of saddle connections that cannot be removed by Whitehead equivalences. However, we show that the set of twists for which we cannot perform the gluing is a countable subset of \mathbb{R} with open dense complement. These notions are the necessary ingredients for the definition of the combinatorial analogue of Fenchel–Nielsen coordinates (Section 2.5). We give here a concise form of the stronger Theorem 2.41.

Theorem A.2. *Given a seamed pants decomposition for a connected bordered surface* Σ *of genus* g *with* n *boundary components, we have an open map*

$$\begin{aligned} \mathfrak{T}_{\Sigma}^{\operatorname{comb}}(\mathsf{L}) &\longrightarrow \mathbb{R}^{3g-3+n}_{+} \times \mathbb{R}^{3g-3+n} \\ \mathbb{G} &\longmapsto \left(\ell(\mathbb{G}), \tau(\mathbb{G})\right) \end{aligned}$$

that is a homeomorphism onto its image, which has a complement of zero measure.

R. Kaufmann pointed out to us that certain aspects of gluing combinatorial structures in terms of the arc complex are discussed in his work with Livernet and Penner [27]. However, to the best of our knowledge, the construction of combinatorial Fenchel–Nielsen coordinates and Theorem A.2 are new.

Like in the hyperbolic case, we show that the twist parameters can be recovered from the data of 9g-9+3n lengths of simple closed curves, determined by a fixed seamed pants decomposition.

Theorem A.3. *Given a seamed pants decomposition for a connected bordered surface* Σ *of genus* g *with* n *boundary components, there exist simple closed curves* $(\gamma_1, \ldots, \gamma_{3g-3+n}), (\delta_1, \ldots, \delta_{3g-3+n}), (\eta_1, \ldots, \eta_{3g-3+n})$ *such that*

$$\begin{split} \mathfrak{T}^{comb}_{\Sigma}(\mathsf{L}) &\longrightarrow \mathbb{R}^{9g-9+3\mathfrak{n}}_{+} \\ \mathbb{G} &\longmapsto \left(\ell_{\mathbb{G}}(\gamma), \ell_{\mathbb{G}}(\delta), \ell_{\mathbb{G}}(\eta)\right) \end{split}$$

is continuous and injective.

Although [22] contains some partial computations going in this direction, it seems to us that Theorem A.3 cannot be directly extracted from there, as we work with a restricted set of measured foliations. We therefore provide a proof (Section 2.6). This result is used in the derivation of Theorems C.1 and C.2.

In Section 3.1 we recall the definition of Kontsevich 2-form ω_K [28] that gives a symplectic structure on the top-dimensional stratum of $\mathcal{M}_{g,n}^{comb}(L)$. The volume of $\mathcal{M}_{g,n}^{comb}(L)$ with respect to the Liouville volume form $d\mu_K = \omega_K^{3g-3+n}/(3g-3+n)!$, denoted

$$V_{g,n}^{K}(L) = \int_{\mathcal{M}_{g,n}^{comb}(L)} d\mu_{K},$$

is finite. As functions of $L \in \mathbb{R}^n_+$, they are polynomials whose coefficients compute ψ -classes intersections on the Deligne–Mumford compactification of the moduli space of curves, see [28] completed by [46]. After we lift ω_K to a mapping class group invariant 2-form on $\mathcal{T}_{\Sigma}^{\text{comb}}$, the main result of Section 3 is a combinatorial analogue of Wolpert's formula [45], expressing Kontsevich's form in terms of combinatorial Fenchel–Nielsen coordinates (Theorem 3.9 in the text).

Theorem A.4. Let Σ be a connected bordered surface of genus g with n boundary components, and fix any combinatorial Fenchel–Nielsen coordinates (ℓ_i, τ_i) for $\mathcal{T}_{\Sigma}^{\text{comb}}(L)$. Then

$$\omega_K = \sum_{i=1}^{3g-3+n} d\ell_i \wedge d\tau_i.$$

We note that another, *a priori* different set of Darboux coordinates for ω_K have been constructed by Bertola and Korotkin [6] from periods of quadratic differentials. An advantage of our combinatorial Fenchel– Nielsen coordinates is their compatibility with the cutting and gluing operations. This enables us to deduce from Theorem A.4 an integration result for mapping class group invariant functions with respect to the measure μ_K (Section 3.3), which is the analogue of Mirzakhani's integration lemma [32] known in the hyperbolic context.

Geometric recursion (GR) and topological recursion (TR). The Weil–Petersson volumes of $\mathcal{M}_{g,n}(L)$, the Kontsevich volumes, and the counting of lattice points in $\mathcal{M}_{g,n}^{comb}(L)$ all satisfy topological recursion in the sense of [20]. In particular, they can be computed recursively in 2g - 2 + n. We emphasize that this type of recursive relations can be rather systematically obtained from recursions at the level of functions (hence, before integration) on the Teichmüller space. More precisely, building on the framework of geometric recursion in the sense of [2] (which in the original paper has been applied to \mathcal{T}_{Σ}), we set up in Section 4 the geometric recursion to construct mapping class group invariant functions on $\mathcal{T}_{\Sigma}^{comb}$ as well. Let us denote by P a pair of pants and by T a torus with one boundary component. The following result is Theorem 4.4 in the text.

Theorem B.1 (*Combinatorial GR is well-defined*). Let A, B, C be measurable functions on \mathbb{R}^3_+ with A and C symmetric under exchange of their last two variables, D_T be a measurable function on \mathcal{T}^{comb}_T which is mapping class group invariant, and assume they all satisfy the bounds of Definition 4.3. The following definitions are well-posed, and assign functorially to any bordered surface Σ a measurable function Ξ_{Σ} on $\mathcal{T}^{comb}_{\Sigma}$, called the GR amplitude.

- $\Xi_{\rm P}(\mathbb{G}) = A(\vec{\ell}_{\mathbb{G}}(\partial {\rm P}))$, where $\vec{\ell}_{\mathbb{G}}(\partial {\rm P})$ is the triple of combinatorial boundary lengths of P.
- $\Xi_T = D_T$.
- If Σ is a disjoint union of $\Sigma_1, \ldots, \Sigma_k, \Xi_{\Sigma_1 \sqcup \cdots \sqcup \Sigma_k}(\mathbb{G}_1, \ldots, \mathbb{G}_k) = \prod_{i=1}^k \Xi_{\Sigma_i}(\mathbb{G}_i)$.
- If Σ is connected and has Euler characteristic $\chi_{\Sigma} \leq -2$,

$$\Xi_{\Sigma}(\mathbb{G}) = \sum_{m=2}^{n} \sum_{[P] \in \mathcal{P}_{\Sigma}^{m}} B(\vec{\ell}_{\mathbb{G}}(\partial P)) \Xi_{\Sigma-P}(\mathbb{G}|_{\Sigma-P}) + \frac{1}{2} \sum_{[P] \in \mathcal{P}_{\Sigma}^{\varnothing}} C(\vec{\ell}_{\mathbb{G}}(\partial P)) \Xi_{\Sigma-P}(\mathbb{G}|_{\Sigma-P})$$

where $\mathcal{P}_{\Sigma}^{\mathfrak{m}}$ and $\mathcal{P}_{\Sigma}^{\varnothing}$ (cf. Definition 2.22) are certain sets of homotopy classes of embedded pairs of pants in Σ , such that $\Sigma - P$ is stable, and $\mathbb{G}|_{\Sigma - P}$ is the result of cutting the combinatorial structure \mathbb{G} and restricting it to $\Sigma - P$.

Further, the function Ξ_{Σ} *is invariant under mapping classes of* Σ *preserving* $\partial_1 \Sigma$ *.*

By means of Theorem A.4, we show that integrating GR amplitudes automatically yields functions of boundary lengths that satisfy topological recursion. If Σ is a connected bordered surface of type (g, n), let us denote by $\Xi_{g,n}$ the function induced by Ξ_{Σ} on $\mathcal{M}_{g,n}^{comb}$. The following result is Theorem 4.7 in the text.

Theorem B.2 (*TR from GR*). *If* A, B, C *are measurable functions on* \mathbb{R}^3_+ *and* D_T *is a measurable function on* \mathcal{T}^{comb}_T *satisfying the conditions of Definition* 4.5, *then the integrals*

$$V\Xi_{g,n}(L) = \int_{\mathcal{M}_{g,n}^{comb}(L)} \Xi_{g,n} \, d\mu_K$$

exist, define measurable functions on $L \in \mathbb{R}^{n}_{+}$ *, and we have that*

$$\begin{split} & V\Xi_{g,n}(L_{1},...,L_{n}) = \\ &= \sum_{m=2}^{n} \int_{\mathbb{R}_{+}} d\ell \, \ell \, B(L_{1},L_{m},\ell) \, V\Xi_{g,n-1}(\ell,L_{2},...,\widehat{L_{m}},...,L_{n}) \\ &\quad + \frac{1}{2} \int_{\mathbb{R}_{+}^{2}} d\ell d\ell' \, \ell\ell' \, C(L_{1},\ell,\ell') \bigg(V\Xi_{g-1,n+1}(\ell,\ell',L_{2},...,L_{n}) + \sum_{\substack{h+h'=g\\ J \sqcup J' = \{L_{2},...,L_{n}\}}} V\Xi_{h,1+\#J}(\ell,J) \, V\Xi_{h',1+\#J'}(\ell',J') \bigg). \end{split}$$

where by convention $V\Xi_{0,1} = 0$ and $V\Xi_{0,2} = 0$.

A similar result holds if, instead of integrating against μ_K , we sum over the lattice $\mathcal{M}_{g,n}^{comb}\mathbb{Z}(L) \subset \mathcal{M}_{g,n}^{comb}(L)$, which consists of metric ribbon graphs with integral edge lengths. This has no counterpart in the hyperbolic world. Due to the existence of pathological twists for the gluing – which, although they are rare in the whole space, could, and in fact do, hit the lattice – this is only possible under extra conditions for the initial data B and C.

Theorem B.3 (*Discrete TR from GR*). Let A, B, C be functions on \mathbb{R}^3_+ and D_T be a function on \mathcal{T}^{comb}_T such that

$$L_1 + L_2 < \ell \quad \Rightarrow \quad B(L_1, L_2, \ell) = 0, \qquad and \qquad L_1 < \ell + \ell' \quad \Rightarrow \quad C(L_1, \ell, \ell') = 0.$$

For all 2g - 2 + n > 0, the lattice sum

$$\mathsf{N}\Xi_{g,\mathfrak{n}}(\mathsf{L}) = \sum_{\mathbf{G}\in\mathcal{M}_{g,\mathfrak{n}}^{\mathrm{comb},\mathbb{Z}}(\mathsf{L})} \frac{\Xi_{g,\mathfrak{n}}(\mathbf{G})}{\#\mathrm{Aut}(\mathbf{G})}$$

defines a function of $L \in \mathbb{Z}^n_+$ *which is zero whenever* $\sum_{i=1}^n L_i$ *is odd, and otherwise satisfies the recursion*

$$\begin{split} &\mathsf{N}\Xi_{g,n}(\mathsf{L}_{1},\ldots,\mathsf{L}_{n}) = \\ &= \sum_{m=2}^{n} \sum_{\ell>0} \ell \, \mathsf{B}_{\mathbb{Z}}(\mathsf{L}_{1},\mathsf{L}_{m},\ell) \, \mathsf{N}\Xi_{g,n-1}(\ell,\mathsf{L}_{2},\ldots,\widehat{\mathsf{L}}_{m},\ldots,\mathsf{L}_{n}) \\ &+ \frac{1}{2} \sum_{\ell,\ell'>0} \ell \ell' \, \mathsf{C}_{\mathbb{Z}}(\mathsf{L}_{1},\ell',\ell') \bigg(\mathsf{N}\Xi_{g-1,n+1}(\ell,\ell',\mathsf{L}_{2},\ldots,\mathsf{L}_{n}) + \sum_{\substack{\mathsf{h}+\mathsf{h}'=\mathsf{g}\\\mathsf{I}\sqcup \mathsf{I}'=\{\mathsf{L}_{2},\ldots,\mathsf{L}_{n}\}} \mathsf{N}\Xi_{\mathsf{h},1+\#\mathsf{J}}(\ell,\mathsf{J}) \, \mathsf{N}\Xi_{\mathsf{h}',1+\#\mathsf{J}'}(\ell',\mathsf{J}') \bigg), \end{split}$$

where $X_{\mathbb{Z}}(L_1, L_2, L_3)$ is equal to $X(L_1, L_2, L_3)$ if $L_1 + L_2 + L_3$ is an even integer and 0 otherwise.

As applications of this more general theory we can re-prove known results in a completely geometric and uniform way, as well as obtaining new results. A key role for applications is played by the combinatorial analogue of the Mirzakhani–McShane identity, whose proof transposes the original strategy of Mirzakhani [32] to the combinatorial world (where it is much simpler).

Theorem B.4. Denote $[x]_+ = \max\{x, 0\}$ and define the Kontsevich initial data

$$\begin{split} A^{K}(L_{1}, L_{2}, L_{3}) &= 1, \\ B^{K}(L_{1}, L_{2}, \ell) &= \frac{1}{2L_{1}} \left([L_{1} - L_{2} - \ell]_{+} - [-L_{1} + L_{2} - \ell]_{+} + [L_{1} + L_{2} - \ell]_{+} \right), \\ C^{K}(L_{1}, \ell, \ell') &= \frac{1}{L_{1}} [L_{1} - \ell - \ell']_{+}, \\ D^{K}_{T}(\mathbb{G}) &= \sum_{\gamma \in S^{\circ}_{T}} C^{K} \big(\ell_{\mathbb{G}}(\partial T), \ell_{\mathbb{G}}(\gamma), \ell_{\mathbb{G}}(\gamma) \big), \end{split}$$

where S°_{T} is the set of isotopy classes of simple closed curves which are not boundary parallel on T. The corresponding *GR* amplitudes are $\Xi^{K}_{\Sigma}(\mathbb{G}) = 1$ for any $\mathbb{G} \in \mathbb{T}^{\text{comb}}_{\Sigma}$ and any bordered surface Σ .

Combining this result with Theorem B.2 gives a new proof of the topological recursion for Kontsevich volumes, whereas Theorem B.3 gives a new proof of the topological recursion for the lattice point count. The former is equivalent to a proof of Witten's conjecture, which originally followed from Kontsevich theorem [28] supplemented by [18]. The latter is a result known since Norbury [36]. The enumeration of the numbers $\#\mathcal{M}_{g,n}^{omb,\mathbb{Z}}(L)$ has been connected to matrix integrals in the early works of Chekhov and Makeenko [12, 14, 13] and further related to enumeration of chord diagrams in [3, 4]. At that point, Schwinger–Dyson equations for such models give rise to equations that are eventually but non-obviously equivalent to [36]. The scheme of proofs we put forward transcends the algebraic manipulation – whose geometric meaning is unclear – pertaining to the realm of matrix integrals and which was necessary in both Kontsevich's original proof and in Chekhov–Makeenko's works.

A geometric proof of Witten's conjecture that is in a way similar to ours was proposed by Bennett, Cochran, Safnuk and Woskoff in [5]. In this regard, the novel element of our work is firstly to make evident the connection of the partition of unity of [5, Section 4] with a Mirzakhani–McShane identity. Then the mechanism of integration in [5] – that relies on a *local* torus action and was valid only for functions of restricted support such as the Kontsevich initial data – gets realised as a special case of the more general Theorem B.2, by means of the *global* Fenchel–Nielsen coordinates and of Theorem A.4.

Flowing from hyperbolic to combinatorial spaces. The hyperbolic and combinatorial Teichmüller spaces can be identified via the spine homeomorphism sp: $\mathcal{T}_{\Sigma} \to \mathcal{T}_{\Sigma}^{\text{comb}}$ originating from the work of Penner [38] and Bowditch–Epstein [9] and Luo [29]. One can in fact interpolate between their respective geometries with the rescaling flow, defined for $\beta \ge 1$ by

$$\sigma \in \mathfrak{T}_{\Sigma}(\mathsf{L}), \qquad \sigma^{\beta} = \mathrm{sp}^{-1}(\beta \operatorname{sp}(\sigma)) \in \mathfrak{T}_{\Sigma}(\beta \mathsf{L}),$$

where the operation β consists in multiplying all edge lengths in sp(σ) by β .

It is known from the works of Mondello [35] and Do [19] that, for each $\sigma \in \mathfrak{T}_{\Sigma}$ as $\beta \to \infty$, the metric $\beta^{-1}\sigma^{\beta}$ converges in the Gromov–Hausdorff sense to $\mathfrak{sp}(\sigma)$, $\beta^{-1}\ell_{\sigma^{\beta}}(\gamma)$ converges to $\ell_{\mathfrak{sp}(\sigma)}(\gamma)$ for each simple closed curve γ , and that the Poisson structure $\beta^2 R_{\beta}^* \pi_{WP}$ converges to π_K on the open cells. For the comparison we have used the map $R_{\beta}: \mathfrak{T}_{\Sigma}^{\mathrm{comb}}(L) \to \mathfrak{T}_{\Sigma}(\beta L)$ which is obtained by composing \mathfrak{sp}^{-1} with β . In Section 5.1, we complete this description by giving uniform bounds for the convergence of lengths and twists. Note that to access twists, we rely on the (9g - 9 + 3n)-theorem A.3.

Theorem C.1. Let Σ be a connected bordered surfaces of type (g, n). For any γ simple closed curve in Σ and $\sigma \in T_{\Sigma}$, we have that

$$\lim_{\beta \to \infty} \frac{\ell_{\sigma^{\beta}}(\gamma)}{\beta} = \ell_{sp(\sigma)}(\gamma)$$

and the convergence is uniform on the thick parts of T_{Σ} . Moreover, consider a seamed pants decomposition, that defines 3g - 3 + n hyperbolic and combinatorial twist parameter functions τ_i . Then for any $\sigma \in T_{\Sigma}$, we have that

$$\lim_{\beta \to \infty} \frac{\tau_i(\sigma^\beta)}{\beta} = \tau_i \big(sp(\sigma) \big),$$

and the convergence is uniform on every compact of T_{Σ} .

As a consequence, we can flow quite systematically results in hyperbolic geometry to results in combinatorial geometry, and natural functions on \mathcal{T}_{Σ} to natural functions on $\mathcal{T}_{\Sigma}^{comb}$. In this regard, $\mathcal{T}_{\Sigma}^{comb}$ behaves like a tropicalisation of \mathcal{T}_{Σ} . We use this method in Section 5.2 to describe the action of the generators of the mapping class group on the combinatorial Fenchel–Nielsen coordinates, from which we conclude the following statement.

Theorem C.2. The combinatorial Fenchel–Nielsen coordinates equip T_{Σ}^{comb} with a piecewise linear structure.

As a second application, we show in Sections 5.3–5.4 that the flow in the $\beta \rightarrow \infty$ limit takes the hyperbolic GR to the combinatorial GR, and does the same for TR after integration.

Theorem C.3. Let $(A_{\beta}, B_{\beta}, C_{\beta})_{\beta \ge 1}$ be a 1-parameter family of initial data, satisfying the conditions of Theorem 5.16 and converging uniformly on any compact to a limit $(\hat{A}, \hat{B}, \hat{C})$ as $\beta \to \infty$. Denote by $\Omega_{\Sigma;\beta}$ the GR amplitudes on T_{Σ} , which results from the initial data $(A_{\beta}, B_{\beta}, C_{\beta})$, and by $\widehat{\Xi}_{\Sigma}$ the GR amplitudes on T_{Σ}^{comb} , which results from the initial data $(\hat{A}, \hat{B}, \hat{C})$. We have for any bordered surface Σ and $\sigma \in T_{\Sigma}$

$$\lim_{\beta\to\infty}\Omega_{\Sigma;\beta}(\sigma^{\beta})=\widehat{\Xi}_{\Sigma}(\operatorname{sp}(\sigma)),$$

and the convergence is uniform on any compact of T_{Σ} .

Theorem C.4. Let $(A_{\beta}, B_{\beta}, C_{\beta})_{\beta \ge 1}$ be a 1-parameter family of initial data, satisfying the conditions of Theorem 5.18 and converging uniformly on any compact to a limit $(\hat{A}, \hat{B}, \hat{C})$ as $\beta \to \infty$. Denote by $\Omega_{\Sigma;\beta}$ the GR amplitudes on \mathfrak{T}_{Σ} , which results from the initial data $(A_{\beta}, B_{\beta}, C_{\beta})$, and by $\widehat{\Xi}_{\Sigma}$ the GR amplitudes on $\mathfrak{T}_{\Sigma}^{\text{comb}}$, which results form the initial data $(\hat{A}, \hat{B}, \hat{C})$. For 2g - 2 + n > 0 and any $L \in \mathbb{R}^{n}_{+}$, we have

$$\lim_{\beta\to\infty}\frac{V\Omega_{\Sigma;\beta}(\beta L)}{\beta^{6g-6+2n}}=V\widehat{\Xi}_{\Sigma}(L),$$

and the convergence is uniform for L in any set of the form $(0, M]^n$ with M > 0.

This gives a second proof of the combinatorial Mirzakhani–McShane identity of Theorem B.4, as a result of applying the flow to the original hyperbolic Mirzakhani–McShane identity of [32].

Application to statistics of measured foliations with respect to combinatorial structures. Along the lines of [2], we prove in Section 4.5 a generalisation of Theorem B.4 that gives access to statistics of combinatorial lengths of multicurves. Denote by M'_{Σ} the set of primitive multicurves on Σ – which include the empty multicurve and exclude the boundary components.

Theorem D.1. Let $f: \mathbb{R}_+ \to \mathbb{R}$ be a measurable function such that $|f(\ell)|$ decays faster than ℓ^{-s} , for any s > 0. Let (A, B, C, D) be initial data satisfying the conditions of Theorem B.1. Denote by Ξ_{Σ} the corresponding GR amplitudes, and assume they are invariant under braidings of all boundary components. Then, the GR amplitudes $\Xi_{\Sigma}[f]$ associated with the initial data

$$\begin{split} A[f](L_1, L_2, L_3) &= A(L_1, L_2, L_3), \\ B[f](L_1, L_2, \ell) &= B(L_1, L_2, \ell) + f(\ell)A(L_1, L_2, \ell), \\ C[f](L_1, \ell, \ell') &= C(L_1, \ell, \ell') + f(\ell)B(L_1, \ell, \ell') + f(\ell')B(L_1, \ell', \ell) + f(\ell)f(\ell')A(L_1, \ell, \ell'), \\ D_T[f](\mathbb{G}) &= D_T(\mathbb{G}) + \sum_{\gamma \in S^\circ_T} f(\ell_{\mathbb{G}}(\gamma)) A(\ell_{\mathbb{G}}(\partial T), \ell_{\mathbb{G}}(\gamma), \ell_{\mathbb{G}}(\gamma)), \end{split}$$

yield

$$\Xi_{\Sigma}[f](\mathbb{G}) = \sum_{c \in \mathcal{M}'_{\Sigma}} \Xi_{\Sigma-c}(\mathbb{G}|_{\Sigma-c}) \prod_{\gamma \in \pi_0(c)} f(\ell_{\mathbb{G}}(\gamma)).$$

On this basis, we analyse in Section 6 various questions related to the asymptotic growth of the number of multicurves of length smaller than or equal to t when $t \to \infty$, obtaining an application to the enumerative geometry of flat surfaces. In [1], the authors and Delecroix have described the Masur–Veech volumes of the moduli space of quadratic differential as the constant term $MV_{g,n}(0)$ of a family of polynomials $MV_{g,n}(L_1, \ldots, L_n)$ satisfying the topological recursion. An immediate consequence of Theorem D.1 is that the Masur–Veech polynomials are also integrals over $\mathcal{M}_{g,n}^{comb}(L)$ against μ_K of statistics of combinatorial lengths of multicurves.

Theorem D.2. *For* 2g - 2 + n > 0*, we have*

$$MV_{g,n}(L) = \frac{2^{4g-2+n}(4g-4+n)!}{(6g-7+2n)!} V\Xi_{g,n}^{MV}(L),$$

where

$$\Xi_{\Sigma}^{\mathrm{MV}}(\mathbb{G}) = \sum_{\mathbf{c} \in \mathcal{M}_{\Sigma}} e^{-\ell_{\mathbb{G}}(\mathbf{c})}$$

are the combinatorial GR amplitudes obtained by twisting the Kontsevich initial data by $f(\ell) = \frac{1}{e^{\ell}-1}$, and M_{Σ} is the set of multicurves on Σ – which include the empty multicurve and exclude the boundary components.

Since the work of Mirzakhani [34], we know that Masur–Veech volumes $MV_{g,n}(0)$ can be accessed through the enumerative geometry of measured foliations with respect to hyperbolic length functions. We show that the role of hyperbolic geometry is not essential. Namely, a similar relation holds with respect to combinatorial length functions. Let MF_{Σ} be the set of measured foliations specified in Section 6.2, equipped with its Thurston measure μ_{Th} coming from integer point counting.

Theorem D.3. For any bordered surface Σ of type (g, n), there exists a continuous function $\mathscr{B}_{\Sigma}^{comb} : \mathscr{T}_{\Sigma}^{comb} \to \mathbb{R}_+$ such that

$$\forall \mathbb{G} \in \mathbb{T}_{\Sigma}^{comb}, \qquad \mathscr{B}_{\Sigma}^{comb}(\mathbb{G}) = \lim_{t \to \infty} \frac{\#\{c \in M_{\Sigma} \mid \ell_{\mathbb{G}}(c) \leqslant t\}}{t^{6g-6+2n}} = \mu_{Th}\big(\{\lambda \in MF_{\Sigma} \mid \ell_{\mathbb{G}}(\lambda) \leqslant 1\}\big).$$

Further, for any $L \in \mathbb{R}^{n}_{+}$, $\mathscr{B}^{comb}_{\Sigma}$ *is integrable on* $\mathcal{M}^{comb}_{g,n}(L)$ *with respect to* μ_{K} *, and*

$$MV_{g,n}(0) = 2^{4g-2+n}(4g-4+n)! \cdot (6g-6+2n) \cdot \int_{\mathcal{M}_{g,n}^{comb}(L)} \mathcal{B}_{\Sigma}^{comb} \, d\mu_{K}.$$

We conclude in Section 6.5 by giving nine equivalent ways to compute $MV_{g,n}(0)$ and discuss the logical relations between them, adding to the contributions of earlier works the perspectives offered by combinatorial geometry.

Appendices. The paper is supplemented with three appendices and an index of notations. In Appendix A we recall the original definition of the topological space T_{Σ}^{comb} from the arc complex, which is shown in Corollary 2.21 to be equivalent to the one induced by combinatorial length functions. In Appendix B we give detailed examples of cutting, gluing and Fenchel–Nielsen coordinates. In Appendix C we present self-contained proofs of facts – known as folklore but for which we could not identify a clear reference – related to the appearance of factors of 2 in the study of the integral structure of $\mathcal{M}_{q,n}^{comb}(L)$.

Notation for statements and proofs. The symbol \Box is used at the end of a proof as usual. The symbol \blacksquare is used at the end of those statements that do not come with a formal proof in the text: they are either proved in other sources in the literature or the argument before their statement already constitutes a proof for them. Those statements that are not followed immediately by a proof or a \blacksquare are proved later in the text.

Acknowledgements. We thank Vincent Delecroix for numerous discussions, Leonid Chekhov and Ralph Kaufmann for references. J.E.A. was supported in part by the Danish National Sciences Foundation Centre of Excellence grant "Quantum Geometry of Moduli Spaces" (DNRF95) at which part of this work was carried out. JEA and D.L. are currently funded in part by the ERC Synergy grant "ReNewQuantum". G.B., S.C., A.G., D.L. and C.W. are supported by the Max-Planck-Gesellschaft. D.L. is supported by the Institut de Physique Théorique (IPhT), CEA, CNRS, and by the Institut des Hautes Études Scientifiques (IHES).

2 Topology of combinatorial Teichmüller spaces

We begin this section with a basic introduction to the combinatorial moduli space $\mathcal{M}_{g,n}^{comb}$, then we study the topology of the combinatorial Teichmüller space $\mathcal{T}_{\Sigma}^{comb}$, which is the orbifold universal cover of $\mathcal{M}_{g,n}^{comb}$. We review its relation with the ordinary Teichmüller space of hyperbolic structures, we propose a description in terms of measured foliations, and then we introduce the associated length functions. We discuss cutting and gluing after a twist, and finally introduce the combinatorial analogue of Fenchel–Nielsen coordinates, that are shown in Section 5.2 to endow $\mathcal{T}_{\Sigma}^{comb}$ with an atlas having piecewise linear coordinate transformations.

2.1 Combinatorial moduli spaces

Definition 2.1. A *graph* is a triple (\vec{E}, i, \sim) , where \vec{E} is a finite set (oriented edges), i: $\vec{E} \rightarrow \vec{E}$ is a fixed-point-free involution and \sim is an equivalence relation on \vec{E} . The set of i-orbits is denoted by E and describes the unoriented edges; the equivalence class of an oriented edge \vec{e} is simply denoted by e. The set $V = \vec{E} / \sim$ describes the vertices. For a fixed $\nu \in V$, $\#\{\vec{e} \in \vec{E} \mid [\vec{e}] = \nu\}$ is called the valency of ν . Connectedness of graphs is defined in the natural way.

Definition 2.2. A *ribbon graph* is a triple $G = (\vec{E}, i, s)$ where \vec{E} is a finite set (oriented edges), $i: \vec{E} \to \vec{E}$ is a fixed-point-free involution and $s: \vec{E} \to \vec{E}$ is a permutation. The set of i-orbits is denoted by $E=E_G$ and describes the unoriented edges; the equivalence class of an oriented edge \vec{e} is simply denoted by e. The set of s-orbits is denoted by $V = V_G$ and describes the vertices. Let $\phi = i \circ s^{-1}$; the set of ϕ -orbits is denoted by $F = F_G$ and describes the faces (or boundaries) of G. Given a ribbon graph (\vec{E}, i, s), the underlying graph is (\vec{E}, i, \sim), where we declare that $\vec{e} \sim \vec{e'}$ if they are in the same s-orbit. A ribbon graph is connected if the underlying graph is.

The genus of a connected ribbon graph G, denoted by g_G , is defined by

$$2 - 2g_G = \#V_G - \#E_G + \#F_G.$$
(2.1)

If furthermore it has $n = \#F_G$ boundaries, it is said to be of type (g, n). We call G

- reduced if all its vertices have valency ≥ 3 ,
- labelled if its boundaries are labelled $\partial_1 G, \ldots, \partial_n G$.

We define $\mathcal{R}_{q,n}$ to be the set of connected, reduced, labelled ribbon graphs of type (g, n).

Remark 2.3. One can think of a ribbon graph as a graph with a cyclic order on the half-edges incident to each vertex. Using this, one can thicken the edges into a ribbons, which at each vertex glues together to form a smooth surface with boundary (see Figure 3), hence the name ribbon graph.

A metric ribbon graph **G** consists of a ribbon graph $G = (\vec{E}, i, s)$, together with the assignment of $\ell_G : E_G \rightarrow \mathbb{R}_+$. The perimeter of a boundary component $\partial_m G$ consisting of the cyclic sequence of edges (e_1, \ldots, e_k) is defined by

$$\ell_{\mathbf{G}}(\partial_{\mathbf{m}}\mathbf{G}) = \sum_{i=1}^{k} \ell_{\mathbf{G}}(e_i).$$
(2.2)

Definition 2.4. An automorphism of a ribbon graph $G = (\vec{E}, i, s)$ is a permutation $\varphi : \vec{E} \to \vec{E}$ that commutes with i and s and acts trivially on the set of faces F. We denote by Aut(G) the automorphism group of G. An automorphism of a metric ribbon graph **G** is an automorphism of the underlying ribbon graph preserving ℓ_{G} . We denote by Aut(G) the automorphism group of G; it is a subgroup of Aut(G).

For any ribbon graph G, its automorphism group Aut(G) acts on the space of metrics on G, by edge permutation. Given a point in $\mathbb{R}^{E_G}_+$, namely a metric ribbon graph G, its stabiliser under the action of Aut(G) is precisely Aut(G). It is then natural to make the following definition.

Definition 2.5. For 2g-2+n > 0, the *combinatorial moduli space* $\mathcal{M}_{g,n}^{\text{comb}}$ is the orbicell-complex parametrising metric ribbon graphs of type (g, n), *i.e.*

$$\mathcal{M}_{g,n}^{\text{comb}} = \bigcup_{G \in \mathcal{R}_{g,n}} \frac{\mathbb{R}_{+}^{\mathsf{L}_{G}}}{\operatorname{Aut}(G)},$$
(2.3)

where the cells naturally glue together via edge degeneration: when an edge length goes to zero, the edge contracts to give a metric ribbon graph with fewer edges as discussed in [28]. Define the perimeter map $p: \mathcal{M}_{g,n}^{comb} \to \mathbb{R}_{+}^{n}$ by setting

$$\mathbf{p}(\mathbf{G}) = \left(\ell_{\mathbf{G}}(\partial_{1}\mathbf{G}), \dots, \ell_{\mathbf{G}}(\partial_{n}\mathbf{G})\right).$$
(2.4)

We also denote $\mathcal{M}_{q,n}^{\text{comb}}(L) = p^{-1}(L)$ for $L \in \mathbb{R}_+^n$.

From the above definition, it follows that the combinatorial moduli space $\mathcal{M}_{g,n}^{comb}$ has a natural real orbifold structure of dimension 6g - 6 + 3n. As we remarked before, for a point **G** in the moduli space its orbifold stabiliser is Aut(**G**). Further, we have an orbicell decomposition given by the sets $\mathbb{R}_{+}^{E_G}$ /Aut(**G**) consisting of those metric ribbon graphs whose underlying ribbon graph is **G**. Notice that the dimension of such a cell is $\#E_G$, and in particular the top-dimensional cells are the ones associated to trivalent metric ribbon graphs, for which $\#E_G = 6g - 6 + 3n$.

Example 2.6. The moduli space $\mathcal{M}_{0,3}^{\text{comb}}$ is homeomorphic through the perimeter map to the open cone \mathbb{R}^3_+ , obtained as the union of seven cells corresponding to the seven ribbon graphs of type (0,3) (see Figure 2a). In this case there are no orbifold points and $\mathcal{M}_{0,3}^{\text{comb}}(L_1, L_2, L_3)$ is just a point.

A more complicated example is that of $\mathcal{M}_{1,1}^{\text{comb}}$. The space is obtained by gluing two orbicells, as depicted in Figure 2b.

• The first orbicell is ℝ³₊ quotiented by ℤ/6ℤ, that is the automorphism group of the unique ribbon graph of type (1,1) with three edges:



All metric ribbon graphs G_1 whose underlying graph is G_1 have $Aut(G_1) = \mathbb{Z}/2\mathbb{Z}$ except for the metric ribbon graph where all three edges have the same length which has $Aut(G_1) = \mathbb{Z}/6\mathbb{Z}$ instead.

• The second orbicell is ℝ²₊ quotiented by ℤ/4ℤ, the automorphism group of the unique ribbon graph of type (1,1) with two edges:



All metric ribbon graphs G_2 whose underlying graph is G_2 have $Aut(G_2) = \mathbb{Z}/2\mathbb{Z}$, except for the metric ribbon graph where both edges have the same length which rather has $Aut(G_1) = \mathbb{Z}/4\mathbb{Z}$.





(a) The 7 cells composing $\mathcal{M}_{0,3}^{comb}$. The picture represents the cone \mathbb{R}^3_+ together with a slice { $L_1 + L_2 + L_3 = \text{const}$ }. The dotted lines are not part of the space.

(b) The two orbicells composing $\mathcal{M}_{1,1}^{\text{comb}}$. On the right, the red point has stabiliser $\mathbb{Z}/6\mathbb{Z}$, the orange point has stabiliser $\mathbb{Z}/4\mathbb{Z}$, and all other points have stabiliser $\mathbb{Z}/2\mathbb{Z}$. The white points are not part of the space.

Figure 2: The combinatorial moduli spaces of type (0,3) and (1,1).

The main reason for considering $\mathcal{M}_{g,n}^{\text{comb}}(L)$ is the following result.

Theorem 2.7. The moduli spaces $\mathfrak{M}_{g,n}$ and $\mathcal{M}_{g,n}^{comb}(L)$ are orbifold-homeomorphic.

There are at least two different ways to define the above homeomorphism. The first, originating from Jenkins and Strebel [26, 41], relies on the geometry of meromorphic quadratic differentials. The second one uses hyperbolic geometry and is due to Penner and Bowditch–Epstein [38, 9]. In the next section we discuss the latter and its generalisation to combinatorial Teichmüller spaces, due to Luo and Mondello [29, 35].

2.2 Combinatorial Teichmüller spaces

One can think of points in the combinatorial Teichmüller space as points in the combinatorial moduli space together with a marking, the advantage being that there is now a well-defined notion of length of curves. We review its known relation with hyperbolic surfaces via the spine construction. Although the topological properties of such spaces have already been studied in the literature, the main focus has been on the proper arc complex description. Here we put forward an equivalent description via measured foliations which later facilitates our constructions.

2.2.1 Primary definitions

Definition 2.8. A *bordered surface* Σ is a non-empty topological, compact, oriented surface with labelled boundary components $\partial_1 \Sigma, \ldots, \partial_n \Sigma$. We assume that each connected component has non-empty boundary and is stable, *i.e.* its Euler characteristic is negative. If furthermore Σ is connected of genus g, we call (g, n) the type of Σ . We use P (resp. T) to refer to bordered surfaces with the topology of a pair of pants (resp. of a torus with one boundary component).

Denote by Mod_{Σ} the *mapping class group* of Σ :

$$\operatorname{Mod}_{\Sigma} = \operatorname{Diff}^{+}(\Sigma) / \operatorname{Diff}_{0}^{+}(\Sigma),$$
 (2.5)

where $\text{Diff}^+(\Sigma)$ is the group of orientation-preserving diffeomorphisms of the surface Σ and $\text{Diff}_0^+(\Sigma)$ denotes its subgroup consisting of those diffeomorphisms isotopic to the identity. The *pure mapping class group*



Figure 3: The geometric realisation of a graph Γ (left), of a metric ribbon graph **G** (center), and the local picture around a vertex of $|\mathbf{G}|$ (right). In red we draw **G** embedded into $|\mathbf{G}|$ and the segments along which the stripes are glued are depicted in blue.

 Mod_{Σ}^{∂} is the subgroup of Mod_{Σ} consisting of mapping classes which preserve the labellings of boundary component of Σ .

To describe the combinatorial Teichmüller space it is convenient to introduce some explicit geometric structures associated with a metric ribbon graph.

Definition 2.9. The *geometric realisation of a graph* Γ is the 1-dimensional CW-complex

$$|\Gamma| = \bigsqcup_{e \in \mathsf{E}} [0, 1] \Big/ \sim$$

obtained by identifying endpoints corresponding to the same vertex in Γ .

Definition 2.10. The *geometric realisation of a metric ribbon graph* \mathbf{G} of type (g, n) is the oriented, compact, genus g surface with n boundary components

$$|\mathbf{G}| = \left\{ (\mathbf{u}, \mathbf{t}, \vec{e}) \in [-1, 1] \times \mathbb{R} \times \vec{E} \mid \mathbf{t} \in [0, \ell_{\mathbf{G}}(e)] \right\} / \sim$$

where the equivalence relation ~ is given by

$$(u, t, \vec{e}) \sim (u, \ell_{G}(e) - t, i(\vec{e}))$$
 for $u \in [-1, 1], \vec{e} \in \vec{E}, t \in [0, \ell_{G}(e)]$

and

$$(\mathfrak{u}, \ell_{\mathbf{G}}(e), \vec{e}) \sim (-\mathfrak{u}, 0, \mathfrak{s}(\vec{e}))$$
 for $\mathfrak{u} \in [0, 1], \ \vec{e} \in \vec{E}$.

The geometric realisation of the underlying graph can be seen as a subset of $|\mathbf{G}|$, and the inclusion is a deformation retract; see Figure 3 for an example.

Definition 2.11. A *combinatorial marking* of a bordered surface Σ is an ordered pair (**G**, f) where **G** is a metric ribbon graph and f: $\Sigma \rightarrow |\mathbf{G}|$ is a homeomorphism respecting the labellings of boundaries of **G** and Σ . The *combinatorial Teichmüller space* is defined as

$$\mathcal{T}_{\Sigma}^{\text{comb}} = \left\{ \left(\mathbf{G}, f \right) \middle| \left(\mathbf{G}, f \right) \text{ is a combinatorial marking on } \Sigma \right\} \middle/ \sim .$$
(2.6)

Here, $(\mathbf{G}, \mathbf{f}) \sim (\mathbf{G}', \mathbf{f}')$ if and only if there exists a metric ribbon graph isomorphism $\varphi \colon \mathbf{G} \to \mathbf{G}'$ such that $\varphi \circ \mathbf{f}$ is isotopic to \mathbf{f}' . We call such equivalence classes *combinatorial structures* and denote by $\mathbb{G} = [\mathbf{G}, \mathbf{f}]$ the points in $\mathcal{T}_{\Sigma}^{\text{comb}}$.

Notice that, for a fixed $\mathbb{G} \in \mathbb{T}_{\Sigma}^{comb}$ each boundary component $\partial_m \Sigma$ corresponds to a unique face of the embedded graph. Thus, we can talk about the length (with respect to \mathbb{G}) of the boundary $\partial_m \Sigma$, denoted by $\ell_{\mathbb{G}}(\partial_m \Sigma)$. In particular, we can define the perimeter map $p: \mathbb{T}_{\Sigma}^{comb} \to \mathbb{R}^n_+$ by setting

$$\mathbf{p}(\mathbb{G}) = \left(\ell_{\mathbb{G}}(\partial_{1}\Sigma), \dots, \ell_{\mathbb{G}}(\partial_{n}\Sigma)\right). \tag{2.7}$$

We also set $\mathcal{T}_{\Sigma}^{comb}(L) = p^{-1}(L)$.

We remark that, for a fixed representative (\mathbf{G}, f) of $\mathbb{G} \in \mathbb{T}_{\Sigma}^{comb}$, we have a retraction from the surface to the geometric realisation of the graph underlying \mathbf{G} . Thus, we can picture elements of $\mathbb{T}_{\Sigma}^{comb}$ as metric ribbon graphs embedded into Σ up to isotopy, such that the embedded graph is a retract of the surface.



Figure 4: An example of combinatorial marking on a torus with one boundary component.

We recall in Appendix A that the combinatorial Teichmüller space can be endowed with a natural topology via the so-called proper arc complex that makes it into a cell complex. The cells, denoted $\mathfrak{Z}_{\Sigma,G}$, are indexed by isotopy classes of embedding of G into Σ , onto which Σ retracts, and they parametrise the metrics of G. The practical consequence of this topological discussion is that the edge lengths form a coordinate system in each cell. In particular, this makes it easier to check whether a function defined on $\mathcal{T}_{\Sigma}^{\text{comb}}$ is piecewise continuous, once it is expressed in terms of edge lengths.

The mapping class group of Σ naturally acts on $\mathcal{T}_{\Sigma}^{comb}$ by setting

$$[\phi].[\mathbf{G},\mathbf{f}] = [\mathbf{G},\mathbf{f}\circ\phi],$$

for $[\phi] \in Mod_{\Sigma}$ and $[\mathbf{G}, f] \in \mathbb{T}_{\Sigma}^{comb}$. When Σ has type (g, n), by forgetting the marking we have the isomorphism

$$\mathfrak{M}_{q,n}^{\operatorname{comb}}(L) \cong \mathfrak{T}_{\Sigma}^{\operatorname{comb}}(L) / \operatorname{Mod}_{\Sigma}^{\mathfrak{d}}$$

We denote the quotient $\mathcal{T}_{\Sigma}^{comb}(L)/Mod_{\Sigma}^{\partial}$ by $\mathcal{M}_{\Sigma}^{comb}(L)$ when we want to refer to the actual surface Σ .

Example 2.12. For a pair of pants P, the perimeter map gives the isomorphism $\mathcal{T}_{P}^{\text{comb}} \cong \mathbb{R}_{+}^{3}$. The pure mapping class group is trivial, so that $\mathcal{M}_{03}^{\text{comb}} \cong \mathbb{R}_{+}^{3}$, see Figure 5a.

For a torus T with one boundary component, \mathfrak{T}_T^{comb} is the union of infinitely many cells homeomorphic to \mathbb{R}^3_+ , glued together through infinitely many cells homeomorphic to \mathbb{R}^2_+ . In Figure 5b we presented some adjacent cells of \mathfrak{T}_T^{comb} . In the quotient by Mod^3_T , all the 3- and 2-cells are identified, and we are left with a further action of $\mathbb{Z}/6\mathbb{Z}$ for the top-dimensional cell and an action of $\mathbb{Z}/4\mathbb{Z}$ for the codimension 1 cell. The result is the combinatorial moduli space $\mathfrak{M}_{1,1}^{comb}$ described in Example 2.6.

2.2.2 Relation with ordinary Teichmüller spaces

For a fixed connected bordered surface Σ of type (g, n), we can consider the ordinary Teichmüller space. Recall that a hyperbolic marking on Σ is a pair (X, f) where X is a hyperbolic surface with labelled geodesic boundaries and f: $\Sigma \rightarrow X$ is an orientation-preserving diffeomorphism respecting the labelling. Define the Teichmüller space as

$$\mathfrak{T}_{\Sigma} = \left\{ \left(X, f \right) \middle| \left(X, f \right) \text{ is a hyperbolic marking on } \Sigma \right\} \middle/ \sim,$$
(2.8)

where $(X, f) \sim (X', f')$ if and only if there exists an isometry $\varphi: X \to X'$ respecting the labelling of the boundaries and such that $\varphi \circ f$ is isotopic to f'. We denote points in \mathcal{T}_{Σ} by $\sigma = [X, f]$, and we call them



Figure 5: The combinatorial Teichmüller spaces of type (0,3) and (1,1).

hyperbolic structures on Σ . By considering hyperbolic length of the boundaries, we have a perimeter map $p: \mathfrak{T}_{\Sigma} \to \mathbb{R}^n_+$ and we set $\mathfrak{T}_{\Sigma}(L) = p^{-1}(L)$ for $L \in \mathbb{R}^n_+$.

It is well-known that the quotient of $T_{\Sigma}(L)$ by the pure mapping class group Mod_{Σ}^{∂} is orbifold-homeomorphic to

$$\mathcal{M}_{g,n}(L) = \left\{ X \middle| \begin{array}{c} X \text{ is a hyperbolic surface of type } (g,n) \\ \text{with labelled boundary of lengths } L \end{array} \right\} \middle/ \sim,$$
(2.9)

where $X \sim X'$ if and only if there exists an isometry from X to X' preserving the labelling of the boundary components. Further, for each $L \in \mathbb{R}^{n}_{+}$, $\mathcal{M}_{g,n}(L)$ is orbifold-homeomorphic to the moduli space of complex curves $\mathfrak{M}_{g,n}$. For later use, we review the description of a mapping class group equivariant homeomorphism between the combinatorial Teichmüller space and the ordinary one, due to Luo and Mondello [29, 35]. It lifts to the level of Teichmüller spaces the construction of Penner and Bowditch–Epstein [38, 9].

Let $\sigma \in \mathcal{T}_{\Sigma}(L)$ and fix a representative (X, f). Define the valency $\nu_{\sigma}(q)$ of a point q in the interior of Σ as the number of shortest geodesics joining q to $\partial \Sigma$ that realise the distance $\operatorname{dist}_{\sigma}(q, \partial \Sigma)$. Clearly $\nu_{\sigma}(q) \ge 1$. Define the loci $A_{\sigma} = \{q \in \Sigma \mid \nu_{\sigma}(q) = 2\}$ and $V_{\sigma} = \{q \in \Sigma \mid \nu_{\sigma}(q) \ge 3\}$. We have that A_{σ} is the disjoint union of simple, open geodesic arcs α_e indexed by $E = \pi_0(A_{\sigma})$, the set of edges, and V_{σ} is a finite collection of points, the vertices. We define the *spine* $\operatorname{sp}(\sigma)$ of σ as the 1-dimensional CW-complex embedded in Σ given by $V_{\sigma} \cup A_{\sigma}$.

We can naturally assign to $sp(\sigma)$ a metric $\ell_{sp(\sigma)}$: $E \to \mathbb{R}_+$ in the following way. For each vertex q of $sp(\sigma)$, consider the $\nu_{\sigma}(q)$ shortest geodesics from q to the boundary – which are called *ribs*. Cutting Σ along its ribs yields a union of hexagons; the diagonal of each hexagon whose endpoints are the vertices of the spine corresponds to the edges of the spine (see Figure 6). We assign to it the length of the side of the hexagon which lies along the boundary of Σ ; there are two such sides, but they have the same length since the reflection with respect to the edge is a hyperbolic isometry of the hexagon. In this way, $sp(\sigma)$ induces a combinatorial marking on Σ and the perimeters of $sp(\sigma)$ correspond precisely to the hyperbolic lengths of the boundaries of Σ . Further, isotopy classes of hyperbolic markings correspond to isotopy classes of combinatorial markings. Thus, we are led to the following definition.

Definition 2.13. There exists a well-defined map

$$sp: \mathfrak{T}_{\Sigma}(L) \longrightarrow \mathfrak{T}_{\Sigma}^{comb}(L), \tag{2.10}$$

Figure 6: Example of the spine construction. In red, the spine $sp(\sigma)$. In blue, the ribs emanating from two vertices.

called the spine map.

It is possible (although more difficult) to construct the inverse map and actually show that it is a homeomorphism, equivariant with respect to the action of Mod_{Σ}^{∂} .

Theorem 2.14. [29, 35] The spine map sp: $T_{\Sigma}(L) \to T_{\Sigma}^{comb}(L)$ is a homeomorphism, equivariant under the action of the pure mapping class group.

In [29, 35] the theorem is stated in terms of the proper arc complex, rather than the combinatorial Teichmüller space, but their relation is recalled in Appendix A. As a direct consequence, we find that for each fixed $L \in \mathbb{R}^{n}_{+}$, there is an orbifold homeomorphism $\mathcal{M}_{g,n}(L) \cong \mathcal{M}_{g,n}^{comb}(L)$ and thus with $\mathfrak{M}_{g,n}$ (Theorem 2.7).

2.2.3 Relation with measured foliations

For a given metric ribbon graph, its geometric realisation is naturally endowed with a measured foliation, as we now explain. We refer to [22, Section 5.1] for a complete discussion about measured foliations, but to be self-contained we recall here the basic definitions.

Definition 2.15. Let Σ be a bordered surface, and \mathcal{F} a *foliation* on Σ with isolated singularities. A *transverse invariant measure* on \mathcal{F} is a measure μ defined on each arc transverse to the foliation, invariant under isotopy of arcs through transverse arcs whose endpoints remain in the same leaf. If the arc passes through a singularity, the transversality pertains to all points of the arc belonging to a regular leaf.

In what follows, we also suppose that:

- the measure is regular with respect to the Lebesgue one: every regular point of Σ admits a smooth chart U ∋ (x, y) where the foliation is defined by dy and the measure on each transverse arc is induced by |dy|;
- each point of Σ has a neighbourhood that is the domain of a chart isomorphically foliated as one of the models of Figure 7.

Definition 2.16. We say that two measured foliations on Σ are *Whitehead equivalent* if they differ by isotopy or a Whitehead move, (see Figure 8). We denote by MF_{Σ}^{\star} the set of Whitehead equivalence classes of measured foliations on Σ .

We can now discuss the relation between foliations and metric ribbon graphs.

Definition 2.17. Given a metric ribbon graph **G** of type (g, n), the geometric realisation $|\mathbf{G}|$ has a unique measured foliation $(\mathcal{F}_{\mathbf{G}}, \mu_{\mathbf{G}})$ such that:

• the singularities of $(\mathcal{F}_{G}, \mu_{G})$ are the vertices of the embedded graph,

- the measured foliation is transverse to the embedded graph,
- (𝔅_G, μ_G) on the hexagon around each edge e ∈ E agrees with |dt|, where t is the natural coordinate on [0, ℓ_G(e)] as in Definition 2.10.

The singular leaves of the measured foliation cut $|\mathbf{G}|$ into hexagons, each with two opposite edges consisting of arcs along the boundary of $|\mathbf{G}|$, and the remaining four edges consisting of singular leaves (see Figure 9). The diagonals parallel to the boundary arcs are nothing but the edges of the embedded graph. In the description of § 2.2.2, the singular leaves correspond to the ribs, and the singular points to the vertices of the spine. Notice that such a hexagon decomposition of $|\mathbf{G}|$, together with the assignment of a positive length to each diagonal, is sufficient to reconstruct the metric ribbon graph **G**. For an element $\mathbb{G} = [\mathbf{G}, \mathbf{f}]$ in $\mathcal{T}_{\Sigma}^{\text{comb}}$, we get a natural isotopy class of measured foliations ($\mathcal{F}_{\mathbb{G}}, \mu_{\mathbb{G}}$) on Σ by pushing ($\mathcal{F}_{\mathbf{G}}, \mu_{\mathbf{G}}$) forward along f. In the following, we omit the transverse measure $\mu_{\mathbb{G}}$ when there is no ambiguity.

The above construction defines a map

$$\mathcal{F}_*: \mathcal{T}_{\Sigma}^{\text{comb}} \longrightarrow \text{MF}_{\Sigma}^{\star}, \tag{2.11}$$

whose image is the set of classes of measured foliations admitting a representative, with respect to Whitehead equivalence, whose leaves are all transverse to the boundary of Σ (*cf.* Figure 7b and 7e), and there is no singular leaf connecting two singular points – such singular leaf is called a *saddle connection*. Moreover, the measured foliation can be used to give a hexagonal decomposition of the surface, and reconstruct the embedded metric ribbon graph. To summarise, we have the following lemma.

Lemma 2.18. The map \mathcal{F}_* is injective. Its image consists of measured foliations admitting a representative with respect to Whitehead equivalence whose leaves are all transverse to the boundary of the surface and with no saddle connections.

2.3 Length functions and topology

We introduce now combinatorial length functions for simple closed curves, and show that combinatorial structures are (locally in $\mathcal{T}_{\Sigma}^{\text{comb}}$) completely determined by the knowledge of finitely many of these lengths. This gives an alternative description of the topology on $\mathcal{T}_{\Sigma}^{\text{comb}}$.

2.3.1 Notations for curves

We denote by

- S_Σ the set of homotopy classes of simple closed curves on Σ, and its subset S^o_Σ consisting of non boundary parallel (also called *essential*) curves,
- M_Σ the set of multicurves, *i.e.* homotopy classes of finite unions of pairwise disjoint essential simple closed curves on Σ,
- M[']_Σ the set of primitive multicurves, *i.e.* those multicurves whose components are pairwise non-homotopic.

From now on, curves are always considered up to homotopy. By convention M_{Σ} and M'_{Σ} contain the empty multicurve, whereas S_{Σ} and S_{Σ}° do not.

If Σ is a bordered surface and $\gamma \in S_{\Sigma}^{\circ}$ (or more generally in M'_{Σ}), we can consider the closed surface Σ_{γ} defined as the result of cutting Σ along a chosen representative of γ . The assumptions on γ imply that every connected component of Σ_{γ} is stable. Among such connected components, there is one containing $\partial_{1}\Sigma$. We label the boundaries of such surface by putting the components of $\partial\Sigma$ first (in the order they appear in Σ), followed by those of γ (in some order). For the connected components of Σ_{γ} that do not contain $\partial_{1}\Sigma$.

we label the boundaries by putting the components of γ first (in some order), followed by those of $\partial \Sigma$ (in the order they appear in Σ). In the following we specify the choice of order only in case it has an actual relevance in the argument.

2.3.2 Combinatorial length functions

Consider a combinatorial structure $\mathbb{G} \in \mathbb{T}_{\Sigma}^{\text{comb}}$ and a simple closed curve $\gamma \in S_{\Sigma}$. As discussed in the previous section, \mathbb{G} induces an isotopy class of measured foliations ($\mathcal{F}_{\mathbb{G}}, \mu_{\mathbb{G}}$) on Σ , so we have a well-defined length function $\ell_{\mathbb{G}}$ defined on S_{Σ} (*cf.* [22, Section 5.3])¹

$$\ell_{\mathbb{G}}(\gamma) = \inf_{\gamma_0} \sup_{\alpha_i} \left(\sum_{i=1}^k \mu_{\mathbb{G}}(\alpha_i) \right),$$
(2.12)

where the infimum is taken over all representatives γ_0 in the homotopy class γ , and the supremum is taken over all the possible ways of writting γ_0 as a sum of arcs $\alpha_1, \ldots, \alpha_k$, mutually disjoint and transverse to $\mathcal{F}_{\mathbb{G}}$.

It can be shown [22, Proposition 5.7] that the infimum is reached by quasitransverse representatives of γ .

Definition 2.19. We say that a curve γ_0 is *quasitransverse* to a foliation \mathcal{F} if each connected component of γ_0 minus the singularities of \mathcal{F} is either a leaf or is transverse to \mathcal{F} . Further, in a neighbourhood of a singularity, we require that no transverse arc lies in a sector adjacent to an arc contained in a leaf, and that consecutive transverse arcs lie in distinct sectors.

In terms of the embedded metric ribbon graph, we have the following effective way to compute the length $\ell_{\mathbb{G}}(\gamma)$: consider the (unique) representative γ_0 of γ that has been homotoped to the embedded graph and is non-backtracking. The length of the curve can now be computed as the sum of the lengths of the edges visited by the curve, since such γ_0 is a quasitransverse representative, perhaps after performing a sequence of Whitehead moves in a small disc neighbourhood of the vertices of \mathbb{G} , making a measured foliation $\mathcal{F}'_{\mathbb{G}}$ Whitehead equivalent to $\mathcal{F}_{\mathbb{G}}$ and thus representing the same point in MF^{*}_{\Sigma}. One can also conclude from [22, Proposition 5.9] that the length is always positive.

Notice that the assignment $\mathbb{G} \mapsto \ell_{\mathbb{G}}(\gamma)$ is continuous on $\mathbb{T}_{\Sigma}^{\text{comb}}$, as it is a sum of edge lengths on the closure of each open cell. If the curve is homotopic to one of the boundary components, the notion of boundary length described before (Equation (2.7)) agrees with this more general definition. The notion of length naturally extends to M_{Σ} and M'_{Σ} by adding the length of components. Using Equation (2.12), we can also define the length of homotopy classes (relative to $\partial \Sigma$) of arcs between boundaries. This is again a continuous assignment, but it can now take zero values.

In § 2.3.3-2.3.4 we prove the following result regarding the combinatorial length spectrum, which is the combinatorial analogue of [22, Theorem 1.4].

Theorem 2.20. Let us equip $\mathbb{R}^{S_{\Sigma}}_{+}$ with the product topology. The combinatorial length of simple closed curves gives a map $\ell_*: \mathbb{T}_{\Sigma}^{\text{comb}} \to \mathbb{R}^{S_{\Sigma}}_{+}$ which is a homeomorphism onto its image.

Corollary 2.21. The topology on $\mathcal{T}_{\Sigma}^{\text{comb}}$ defined via the arc-complex (Appendix A) and the initial topology induced by the combinatorial length map $\ell_* : \mathcal{T}_{\Sigma}^{\text{comb}} \to \mathbb{R}_+^{S_{\Sigma}}$ coincide.

When Σ is a closed surface, [22, Proposition 6.12] shows that the length map $\ell_*: MF_{\Sigma}^* \to \mathbb{R}^{S_{\Sigma}}_+$ is a bijection onto its image, and equips MF_{Σ}^* with the initial topology induced by this map, making it a homeomorphism onto its image. Such a result – and a topology for MF_{Σ}^* – has however not been spelled out in the case of bordered surfaces, which would be necessary if one wished to prove Theorem 2.20 by injecting $\mathcal{T}_{\Sigma}^{\text{comb}}$ into MF_{Σ}^* with the map \mathcal{F}_* of Equation (2.11). Here we prove the theorem by a direct construction of the inverse

¹In [22] this quantity is denoted by $I(\mathcal{F}, \mu; \gamma)$.

map from the image of l_* , via the study of lengths of curves and their relation with embedded pairs of pants. These computations are used again in Section 4.

In Theorem 2.42 we moreover justify the fact that, as in the hyperbolic case, the data of 9g - 9 + 3n simple closed curves is enough for the reconstruction.

2.3.3 Embedded pairs of pants

We are interested in embedded pairs of pants and the way the lengths of their boundary components (with respect to a combinatorial structure) determine the combinatorial structure itself.

Definition 2.22. When Σ is a bordered surface of type (g, n) such that 2g - 2 + n > 1, and $m_0 \in \{1, ..., n\}$, we let $\mathcal{P}_{\Sigma, m_0}$ be the set of homotopy classes of embeddings $\varphi \colon P \hookrightarrow \Sigma$ of pairs of pants P such that

- $\varphi(\partial_1 P) = \partial_{m_0} \Sigma$,
- $\overline{\Sigma \setminus \varphi(P)}$ (where the closure is taken in Σ) is stable,
- if $\varphi(\partial_i P) = \partial_m \Sigma$ for some $m \neq m_0$, then i = 2.

We denote by $\mathcal{P}_{\Sigma,m_0}^{\varnothing}$ the set of homotopy classes of pair of pants [P]'s for which $\partial_i P \subset \Sigma^\circ$ for i = 2, 3, and by $\mathcal{P}_{\Sigma,m_0}^m$ the set of such [P]'s for which $\partial_2 P = \partial_m \Sigma$ for some $m \neq m_0$. We have a partition

$$\mathcal{P}_{\Sigma,\mathfrak{m}_0} = \left(\bigsqcup_{\mathfrak{m} \neq \mathfrak{m}_0} \mathcal{P}^{\mathfrak{m}}_{\Sigma,\mathfrak{m}_0} \right) \sqcup \mathcal{P}^{\varnothing}_{\Sigma,\mathfrak{m}_0}.$$

These pairs of pants can be described alternatively in terms of arcs between boundaries [2], which we recall here for the convenience of the reader.

Definition 2.23. Denote by \mathcal{A}_{Σ,m_0} the set of non-trivial homotopy classes of proper embeddings² \mathfrak{a} : $[0,1] \hookrightarrow \Sigma$ with $\mathfrak{a}(0) \in \mathfrak{d}_{m_0}\Sigma$. We denote the class of a proper embedding \mathfrak{a} with the letter α . There is a partition

$$\mathcal{A}_{\Sigma,\mathfrak{m}_0} = \left(\bigsqcup_{\mathfrak{m} \neq \mathfrak{m}_0} \mathcal{A}_{\Sigma,\mathfrak{m}_0}^{\mathfrak{m}}
ight) \sqcup \mathcal{A}_{\Sigma,\mathfrak{m}_0}^{\varnothing}$$

where the elements of $\mathcal{A}_{\Sigma,m_0}^m$ are those with a representative a such that $\mathfrak{a}(1) \in \mathfrak{d}_m \Sigma$, and the elements of $\mathcal{A}_{\Sigma,m_0}^{\varnothing}$ are those with a representative a such that $\mathfrak{a}(1) \in \mathfrak{d}_{m_0} \Sigma$.

We obtain a surjective map

$$Q_{\mathfrak{m}_0} \colon \mathcal{A}_{\Sigma,\mathfrak{m}_0} \longrightarrow \mathcal{P}_{\Sigma,\mathfrak{m}_0} \tag{2.13}$$

by assigning to an arc $\alpha = [a]$ the homotopy class of pairs of pants [P] whose boundaries are formed by the boundary of a closed tubular neighbourhood of a and curves homotopic to the boundaries joined by a (Figure 11). The boundaries of P are labelled as follows. We always set $\partial_1 P = \partial_{m_0} \Sigma$. Then if $\varphi(\partial_i P) = \partial_m \Sigma$ for some $m \neq m_0$, then i = 2 and $\partial_3 P$ is determined; otherwise, we define $\partial_2 P$ (resp. $\partial_3 P$) to be the boundary component on the left-hand side (resp. right-hand side) of the curve α oriented from 0 to 1.

Remark 2.24. The restriction of Q_{m_0} to \mathcal{A}_{Σ,m_0} is not injective. More precisely, $Q_{m_0}^{-1}([P])$ contains a single element when $[P] \in \mathcal{P}_{\Sigma,m_0}^{\mathfrak{m}}$, while it contains three elements when $[P] \in \mathcal{P}_{\Sigma,m_0}^{\mathfrak{m}}$ for any $\mathfrak{m} \neq \mathfrak{m}_0$. Indeed any given $[P] \in \mathcal{P}_{\Sigma,m_0}^{\mathfrak{m}}$ can be obtained by an arc α from $\partial_{\mathfrak{m}_0}\Sigma$ to $\partial_{\mathfrak{m}}\Sigma$, but also by an arc α' from $\partial_{\mathfrak{m}_0}\Sigma$ to itself and its inverse $-\alpha'$ (Figure 11). Notice that in this case α is the only arc in $\mathcal{A}_{\Sigma,\mathfrak{m}_0}^{\mathfrak{m}}$, while $\alpha', -\alpha' \in \mathcal{A}_{\Sigma,\mathfrak{m}_0}^{\varnothing}$.

We introduce the notion of small pairs of pants, which play a role in the rest of the paper.

²If X and Y are topological manifolds with boundaries, a continuous map $f: X \to Y$ is called a proper embedding if $f^{-1}(\partial Y) = \partial X$, and we use here the natural notion of homotopies among such.

Definition 2.25. Let $\mathbb{G} \in \mathcal{T}_{\Sigma}^{\text{comb}}$. We say that $[P] \in \mathcal{P}_{\Sigma,m_0}$ is \mathbb{G} -small if

$$\ell_{\mathbb{G}}(\varphi(\partial P) \cap \partial \Sigma) \geqslant \ell_{\mathbb{G}}(\varphi(\partial P) \cap \Sigma^{\circ}). \tag{2.14}$$

The definition does not depend on the representative chosen for [P]. If the inequalities are strict, then we say the pair of pants is strictly G-small.

In other words, dropping the embedding from the notation:

- when $[P] \in \mathcal{P}_{\Sigma,m_0}^m$, we have $\ell_{\mathbb{G}}(\partial_3 P) \leq \ell_{\mathbb{G}}(\partial_1 P) + \ell_{\mathbb{G}}(\partial_2 P)$,
- when $[P] \in \mathcal{P}_{\Sigma_{max}}^{\varnothing}$ we have $\ell_{\mathbb{G}}(\mathfrak{d}_2 P) + \ell_{\mathbb{G}}(\mathfrak{d}_3 P) \leq \ell_{\mathbb{G}}(\mathfrak{d}_1 P)$.

We can characterise small pairs of pants in terms of the corresponding arcs as follows.

Lemma 2.26. Let $[P] \in \mathcal{P}_{\Sigma, \mathfrak{m}_0}$.

- $[P] \in \mathcal{P}_{\Sigma,m_0}^m$ is \mathbb{G} -small if and only if for the unique $\alpha \in Q_{m_0}^{-1}([P]) \cap \mathcal{A}_{\Sigma,m_0}^m$ we have $\ell_{\mathbb{G}}(\alpha) = 0$,
- $[P] \in \mathcal{P}^{\varnothing}_{\Sigma,m_0}$ is G-small if and only if for the unique $\alpha \in Q_{m_0}^{-1}([P])$ we have $\ell_{\mathbb{G}}(\alpha) = 0$.

Proof. Let $\alpha \in \mathcal{A}_{\Sigma,m_0}^m$ for $m \neq m_0$ and assume that $\ell_{\mathbb{G}}(\alpha) > 0$. It is then uniquely represented by some non backtracking edgepath α in \mathbb{G} with initial and final vertices adjacent to $\partial_{m_0}\Sigma$ and $\partial_m\Sigma$. Then, $\partial_3 Q_{m_0}(\alpha)$ can be homotoped to a non backtracking edgepath consisting in travelling along α , then going around $\partial_m\Sigma$, then travelling backwards along α , and going around $\partial_{m_0}\Sigma$. This implies that

$$\ell_{\mathbb{G}}(\mathfrak{d}_{\mathfrak{m}_{0}}\Sigma) + \ell_{\mathbb{G}}(\mathfrak{d}_{\mathfrak{m}}\Sigma) < \ell_{\mathbb{G}}(\mathfrak{d}_{3}Q_{\mathfrak{m}_{0}}(\alpha)),$$

hence $Q_{\mathfrak{m}_0}(\alpha)$ is not G-small, and we conclude by the contrapositive. A similar argument works when $\alpha \in \mathcal{A}_{\Sigma,\mathfrak{m}_0}^{\varnothing}$.

Remark 2.27. Notice that α can only have length zero with respect to \mathbb{G} if it has a quasitransverse representative given by an oriented non-singular leaf in the foliation associated to \mathbb{G} . There are finitely many homotopy classes of oriented leaves relative to the boundary, and they are in bijection with the oriented edges of the metric ribbon graph. Since \mathcal{A}_{Σ,m_0} surjects onto \mathcal{P}_{Σ,m_0} for Σ of type (g,n) and $\mathbb{G} \in \mathcal{T}_{\Sigma}^{\text{comb}}$, there are at most 2(6g - 6 + 3n) \mathbb{G} -small pairs of pants in \mathcal{P}_{Σ,m_0} .

Small pairs of pants can be characterised in terms of the support of two functions that play an important role in Section 4.4. Let $[x]_+ = \max\{x, 0\}$ and consider the functions on $\mathcal{T}_P^{\text{comb}} \cong \mathbb{R}^3_+$ defined by

$$B^{K}(L_{1}, L_{2}, \ell) = \frac{1}{2L_{1}} \Big([L_{1} - L_{2} - \ell]_{+} - [-L_{1} + L_{2} - \ell]_{+} + [L_{1} + L_{2} - \ell]_{+} \Big),$$

$$C^{K}(L_{1}, \ell, \ell') = \frac{1}{L_{1}} [L_{1} - \ell - \ell']_{+}.$$
(2.15)

It is easy to check that these functions only take non-negative values. They encode aspects of the geometry of combinatorial pairs of pants, as described in the following lemma, and are the combinatorial analogs of the functions \mathcal{D} and \mathcal{R} introduced by Mirzakhani in [32] in the hyperbolic context.

Lemma 2.28. The function B^{K} associates to $(\ell_{\mathbb{G}}(\partial_{1}P), \ell_{\mathbb{G}}(\partial_{2}P), \ell_{\mathbb{G}}(\partial_{3}P)) \in \mathbb{R}^{3}_{+} \cong \mathbb{T}_{P}^{comb}$ the fraction of the $\partial_{1}P$ that is not common with $\partial_{3}P$ (once retracted to the graph). Similarly, C^{K} associates to a point $\vec{\ell}_{\mathbb{G}}(\partial P) \in \mathbb{R}^{3}_{+} \cong \mathbb{T}_{P}^{comb}$ the fraction of $\partial_{1}P$ that is not common with $\partial_{2}P \cup \partial_{3}P$.

Proof. The result follows from direct computations in the closure of each of the four open cells of \mathcal{T}_{P}^{comb} . The various inequalities that define the cells are used to simplify B^{K} and C^{K} . Consider for example Figure 12a: we have $\ell_{\mathbb{G}}(\partial_{3}P) \ge \ell_{\mathbb{G}}(\partial_{1}P) + \ell_{\mathbb{G}}(\partial_{2}P)$ so that $B^{K}(L_{1}, L_{2}, \ell) = 0$, and the portion of $\partial_{1}P$ that is not common with $\partial_{3}P$ is zero (all of $\partial_{1}P$ intersects with $\partial_{3}P$). Similarly, in the case of Figure 12b we find $\ell_{\mathbb{G}}(\partial_{2}P) \ge \ell_{\mathbb{G}}(\partial_{1}P) + \ell_{\mathbb{G}}(\partial_{3}P)$, so that

$$B^{K}(L_{1},L_{2},\ell) = \frac{1}{2\ell_{\mathbb{G}}(\mathfrak{d}_{1}\mathsf{P})} \Big(\ell_{\mathbb{G}}(\mathfrak{d}_{1}\mathsf{P}) + \ell_{\mathbb{G}}(\mathfrak{d}_{2}\mathsf{P}) - \ell_{\mathbb{G}}(\mathfrak{d}_{3}\mathsf{P}) - \big(-\ell_{\mathbb{G}}(\mathfrak{d}_{1}\mathsf{P}) + \ell_{\mathbb{G}}(\mathfrak{d}_{2}\mathsf{P}) - \ell_{\mathbb{G}}(\mathfrak{d}_{3}\mathsf{P})\big) \Big) = 1,$$

and indeed all of $\partial_1 P$ intersects with $\partial_2 P$. Finally, in Figure 12c we find $\ell_G(\partial_i P) \leq \ell_G(\partial_j P) + \ell_G(\partial_k P)$ for all $i, j, k \in \{1, 2, 3\}$, which is the condition to be in this cell. We also see that the length of the edge adjacent to $\partial_1 P$ and $\partial_2 P$ is given by $\frac{1}{2} (\ell_G(\partial_1 P) + \ell_G(\partial_2 P) - \ell_G(\partial_3 P))$ and therefore its fraction of the first boundary is given by exactly

$$\mathsf{B}^{\mathsf{K}}(\mathsf{L}_{1},\mathsf{L}_{2},\boldsymbol{\ell}) = \frac{1}{2\ell_{\mathbb{G}}(\vartheta_{1}\mathsf{P})} \big(\ell_{\mathbb{G}}(\vartheta_{1}\mathsf{P}) + \ell_{\mathbb{G}}(\vartheta_{2}\mathsf{P}) - \ell_{\mathbb{G}}(\vartheta_{3}\mathsf{P})\big).$$

The computation in other cells is similar, and analogously for C^K.

Notice that if B^K or C^K are non-zero, then there is at least an edge with a corresponding arc, that defines the pair of pants passing through the edge transversely, and therefore of zero length. As an immediate consequence this or simply by considering the support of B^K and C^K , we deduce the following characterisation of small pairs of pants.

Corollary 2.29. Let $\mathbb{G} \in \mathcal{T}_{\Sigma}^{\text{comb}}$:

- $[P] \in \mathcal{P}_{\Sigma,m_0}^m$ is G-small if and only if $B^K(\vec{\ell}_G(\partial P)) > 0$,
- $[P] \in \mathfrak{P}^{\varnothing}_{\Sigma, \mathfrak{m}_{0}}$ is \mathbb{G} -small if and only if $C^{K}(\vec{\ell}_{\mathbb{G}}(\partial P)) > 0$.

Here $\vec{l}_{\mathbb{G}}(\partial P)$ *is the ordered triple of combinatorial lengths of the boundary components of* P.

2.3.4 A partial inverse of the combinatorial length spectrum map

In this paragraph we exhibit an inverse on the image of the combinatorial length map of Theorem 2.20. We first observe that, given $\mathbb{G} = [\mathbf{G}, \mathbf{f}] \in \mathbb{T}_{\Sigma}^{\text{comb}}$ and an oriented edge \vec{e} of the embedded graph, we can take the dual arc $\alpha_{\vec{e}}$: its starting point (resp. ending point) lies on the component of $\partial \Sigma$ adjacent to $\mathbf{f}(\vec{e})$ on the right (resp. left) and intersects the embedded graph exactly once through this edge. Note that the two boundary components adjacent to \vec{e} may coincide. We thus obtain a map $\vec{E}_{\mathbf{G}} \to \mathcal{A}_{\Sigma}^{\text{all}} := \bigsqcup_{m_0=1}^n \mathcal{A}_{\Sigma,m_0}$. Composing with Q_{m_0} , we can associate to each oriented edge of \mathbb{G} a class of embedded pair of pants in $\mathcal{P}_{\Sigma}^{\text{all}} = \bigsqcup_{m_0=1}^n \mathcal{P}_{\Sigma,m_0}$.

Lemma 2.30. Let e be an edge in $\mathbb{G} \in \mathbb{T}_{\Sigma}^{\text{comb}}$ and fix an arbitrary orientation \vec{e} .

If e is adjacent to ∂_{m1}Σ ≠ ∂_{m2}Σ, let [P₁] ∈ P^{m2}_{Σ,m1} and [P₂] ∈ P^{m1}_{Σ,m2} be the pairs of pants corresponding to e and i(e) respectively. Then

$$\begin{split} \ell_{\mathbb{G}}(e) &= \ell_{\mathbb{G}}(\vartheta_{\mathfrak{m}_{1}}\Sigma) \Big(B^{K}(\vec{\ell}_{\mathbb{G}}(\vartheta P_{1})) - C^{K}(\vec{\ell}_{\mathbb{G}}(\vartheta P_{1})) \Big) \\ &= \ell_{\mathbb{G}}(\vartheta_{\mathfrak{m}_{2}}\Sigma) \Big(B^{K}(\vec{\ell}_{\mathbb{G}}(\vartheta P_{2})) - C^{K}(\vec{\ell}_{\mathbb{G}}(\vartheta P_{2})) \Big). \end{split}$$

If e is adjacent to ∂_{m₀}Σ on both sides, let [P] ∈ P_{Σ,m₀} be the pair of pants corresponding to e or i(e) (the two pairs of pants coincide). Then

$$\ell_{\mathbb{G}}(e) = \frac{1}{2} \ell_{\mathbb{G}}(\partial_{\mathfrak{m}_0} \Sigma) \operatorname{C}^{\mathrm{K}}(\vec{\ell}_{\mathbb{G}}(\partial \mathsf{P})).$$

Proof. Given a boundary component, we can represent the edges around it by a polygon where some edges and vertices are identified. The sequence of edges around the boundary is non-backtracking.

If *e* is adjacent to $\partial_{m_1}\Sigma \neq \partial_{m_2}\Sigma$, we have two polygons around $\partial_{m_1}\Sigma$ and $\partial_{m_2}\Sigma$. When neither of these are 1-gons, then we can represent $\gamma = \partial_3 P_1 = \partial_3 P_2$ as in Figure 13a. Due to the absence of bivalent vertices, this is again non backtracking. We therefore have $\ell_{\mathbb{G}}(\gamma) = \ell_{\mathbb{G}}(\partial_{m_1}\Sigma) + \ell_{\mathbb{G}}(\partial_{m_2}\Sigma) - 2\ell_{\mathbb{G}}(e)$. Notice that we also have $|\ell_{\mathbb{G}}(\partial_{m_1}\Sigma) - \ell_{\mathbb{G}}(\partial_{m_2}\Sigma)| < \ell_{\mathbb{G}}(\gamma)$. Therefore, comparing the expression for $\ell_{\mathbb{G}}(e)$ with the expression in the statement, we see they agree.

Now suppose without loss of generality that m_2 is a 1-gon. This implies that m_1 is not a 1-gon, as gluing two 1-gons together would produce a cylinder. It is also clear that $\ell_{\mathbb{G}}(e) = \ell_{\mathbb{G}}(\partial_{m_2}\Sigma)$. We can represent $\gamma = \partial_3 P_1 = \partial_3 P_2$ by Figure 13b. This implies that $\ell_{\mathbb{G}}(\partial_{m_1}\Sigma) \ge \ell_{\mathbb{G}}(\partial_{m_2}\Sigma) + \ell_{\mathbb{G}}(\gamma)$ and therefore, using this to calculate the expression in the statement, we see that they agree.

If *e* is adjacent to $\partial_{m_0}\Sigma$ on both sides, we have one polygon around $\partial_{m_0}\Sigma$. We can then represent $\partial_2 P$ and $\partial_3 P$ as in Figure 13c. In absence of bivalent vertices, backtracking cannot occur around either side of *e*. Therefore we have $\ell_{\mathbb{G}}(\partial_{m_0}\Sigma) = \ell_{\mathbb{G}}(\partial_2 P) + \ell_{\mathbb{G}}(\partial_3 P) + 2\ell_{\mathbb{G}}(e)$. Comparing the expression for $\ell_{\mathbb{G}}(e)$ with the expression in the statement, we see they agree.

This lemma shows that we can reconstruct edge lengths from lengths of simple closed curves, which in turn proves that Theorem 2.20 holds when restricted to the closure of a cell. To fully reconstruct the ribbon graph (*i.e.* determine in which cell we are) from just the lengths of simple closed curves, we use the characterisations of small pairs of pants and their relation to edges in the embedded graph.

Proof of Theorem 2.20. We can now exhibit a global inverse on the image of the combinatorial length spectrum map ℓ_* : $\mathfrak{T}_{\Sigma}^{\text{comb}} \to \mathbb{R}_+^{S_{\Sigma}}$. We first consider the following composition:

$$\begin{split} \mathbb{R}^{S_{\Sigma}}_{+} &\to (\mathbb{R}^{3}_{+})^{\mathcal{P}^{all}_{\Sigma}} \to \mathbb{R}^{\mathcal{A}^{all}_{\Sigma}}_{\geqslant 0} \\ \lambda &\longmapsto l_{\lambda}. \end{split}$$

The first arrow associates to a length functional $\lambda \in \mathbb{R}^{S_{\Sigma}}_{+}$ the functional on $\mathcal{P}_{\Sigma}^{\text{all}}$ defined by $[P] \mapsto \vec{\lambda}(\partial P)$, where $\vec{\lambda}(\partial P)$ is the ordered triples of boundary lengths. The second arrow associates to a functional $\vec{\lambda}$ on $\mathcal{P}_{\Sigma}^{\text{all}}$ a functional on $\mathcal{A}_{\Sigma}^{\text{all}}$ given by

$$\alpha \longmapsto \begin{cases} \lambda_1(\mathsf{P}_{\alpha}) \Big(\mathsf{B}^{\mathsf{K}}(\vec{\lambda}(\mathsf{P}_{\alpha})) - \mathsf{C}^{\mathsf{K}}(\vec{\lambda}(\mathsf{P}_{\alpha})) \Big) & \text{ if } \alpha \in \mathcal{A}_{\Sigma,\mathfrak{m}_0}^{\mathfrak{m}} \\ \frac{1}{2} \lambda_1(\mathsf{P}_{\alpha}) \operatorname{C}^{\mathsf{K}}(\vec{\lambda}(\mathsf{P}_{\alpha})) & \text{ if } \alpha \in \mathcal{A}_{\Sigma,\mathfrak{m}_0}^{\varnothing} \end{cases}$$

where $[P_{\alpha}] = Q_{m_0}(\alpha)$ and λ_1 is the first component of the triple $\overline{\lambda}$. Note that these are exactly the expressions appearing in Lemma 2.30.

Consider now the restriction to $\operatorname{Im}(\ell_*)$. If $\lambda = \ell_*(\mathbb{G})$ for some $\mathbb{G} \in \mathfrak{T}_{\Sigma}^{\operatorname{comb}}$, we know from Lemma 2.26 and Remark 2.27 that l_{λ} , as a functional on $\mathcal{A}_{\Sigma}^{\operatorname{all}}$, is supported on the complement of those oriented arcs α homotopic to non-singular oriented leaves in the foliation associated to \mathbb{G} . These are in bijection with the oriented edges of \mathbb{G} . Forgetting about the orientation, consider the finite collection $(\alpha_1, \ldots, \alpha_k)$ of all such arcs, choosing representatives that are pairwise non-intersecting in Σ . This defines a proper simplex in the arc complex \mathcal{A}_{Σ} (see Appendix A), and taking the dual we obtain a ribbon graph **G** with an embedding into Σ . We can equip it with a metric, assigning the length $l_{\lambda}(\alpha_i)$ to the edge dual to α_i . This represents a point in $\mathfrak{T}_{\Sigma}^{\operatorname{comb}}$, so that we have a map

$$l_*: \operatorname{Im}(\ell_*) \longrightarrow \mathfrak{T}_{\Sigma}^{\operatorname{comb}}.$$

By construction, $l_* \circ \ell_* = id$. Thus, ℓ_* is injective.

The map ℓ_* is clearly continuous, since the lengths of simple closed curves are linear combinations of lengths of edges. The inverse map l_* is also continuous on the ℓ_* -image of each cell, since we realised the edge lengths as piecewise linear (and thus continuous) functions of the length of locally finitely many simple closed curves. This completes the proof.

2.4 Cutting and gluing

Before describing a combinatorial version of Fenchel–Nielsen coordinates, we need the notion of cutting a combinatorial structure along an essential simple closed curve, and the reciprocal notion of gluing combinatorial structures along boundary components of the same length.

2.4.1 Basic definitions

CUTTING. Consider a bordered surface Σ , fix $\mathbb{G} \in \mathbb{T}_{\Sigma}^{comb}$ and γ an essential simple closed curve. We want to define a combinatorial structure on the surface Σ_{γ} obtained by cutting Σ along a chosen representative of γ . To this end, choose a representative (**G**, f), so that we have an induced structure of measured foliation on Σ . If necessary, perform a minimal sequence of local Whitehead moves in small disc neighbourhoods of the vertices, in such a way that γ is transversal to the resulting foliation. We then restrict the measured foliation to Σ_{γ} , which is induced from a unique metric ribbon graph \mathbf{G}_{γ} with an embedding which up to isotopy does not depend on the choices made. This defines a combinatorial structure $\mathbb{G}_{\gamma} \in \mathbb{T}_{\Sigma_{\gamma}}^{comb}$.

Cutting also makes sense when γ is a primitive multicurve, and it is equivalent to cutting along each component of γ in an arbitrary order. Note that the lengths of edges after cutting are again linear combinations of the edge lengths which agree on the closure of the open cells. This shows that the cutting, viewed as a map $\mathcal{T}_{\Sigma}^{\text{comb}} \rightarrow \mathcal{T}_{\Sigma_{\gamma}}^{\text{comb}}$, is continuous. See Figure 14 and Figure 15 for a local illustration of the cutting, and Appendix B for some global examples.

GLUING. Consider a bordered surface Σ , possibly disconnected, with a choice of two boundary components γ_- and γ_+ . Let $\mathbb{G} \in \mathbb{T}_{\Sigma}^{\text{comb}}$ be such that $\ell_{\mathbb{G}}(\gamma_-) = \ell_{\mathbb{G}}(\gamma_+)$. We want to define a combinatorial structure on the surface obtained by topologically gluing γ_- and γ_+ . Fix a representative (**G**, f), so that we have an induced structure of a measured foliation \mathcal{F} on Σ . First, we observe that once we pick a point p_- on γ_- , there is a unique action of \mathbb{R} on γ_- which preserves the induced measure and orientation on γ_- . We let p_-^{τ} be the image of p_- under the action of $\tau \in \mathbb{R}$. Pick now a point p_+ on γ_+ , and identify γ_- with γ_+ in a measure preserving way, such that p_-^{τ} is identified with p_+ in an orientation reversing way. This means that we have a unique measured foliation \mathcal{F}^{τ} induced on the glued surface, which we denote Σ^{τ} .

2.4.2 Admissible gluings

What is not clear from the above construction is whether the measured foliation \mathcal{F}^{τ} is associated to a combinatorial structure on Σ^{τ} . If this is true, we call such τ an *admissible twist*. We refer to Figure 14 and Figure 15 – read from right to left – for a local illustration of the gluing, Appendix B for some global examples, and Figure 16 for an example of \mathcal{F}^{τ} that is not associated to a combinatorial structure.

Proposition 2.31. There exists a unique metric ribbon graph \mathbf{G}^{τ} and a unique marking $f: \Sigma^{\tau} \to |\mathbf{G}^{\tau}|$ up to isotopy, such that the measured foliation induced on Σ^{τ} agrees with \mathcal{F}^{τ} if and only if \mathcal{F}^{τ} has a representative without saddle connections, i.e. no leaf between two singularities.

Proof. Perform a maximal sequence of Whitehead moves, *i.e.* that reduces the connected components of the the compact singular leaves to a graph with one vertex. Let $\Lambda(\mathcal{F}^{\tau})$ be the set of leaves of \mathcal{F}^{τ} , and define $\hat{\Sigma}^{\tau} = \{\lambda \in \Lambda(\mathcal{F}^{\tau}) \mid \lambda \cap \partial \Sigma^{\tau} \neq \emptyset\}$ (*cf.* Figure 16). Then from Poincaré recurrence [22, Theorem 5.2] this is nothing but the union of all leaves which go from boundary to boundary together with the finitely many leaves which connect the boundary to a singular point of \mathcal{F}^{τ} (*i.e.* no leaves starting from the boundary

spiral in the surface). If $\hat{\Sigma}^{\tau} = \Sigma^{\tau}$, then \mathcal{F}^{τ} has no closed singular leaves and we see that the singular leaves of \mathcal{F}^{τ} split the surface into hexagons, which in turn determines \mathbf{G}^{τ} uniquely and its marking up to isotopy. If not, choose a good atlas for \mathcal{F}^{τ} as defined in [22, Section 5.2], and observe that the complement of the singular leaves in $\hat{\Sigma}^{\tau}$ is a finite disjoint union of squares S_i , each with a non-singular foliation transverse to two open arcs of the boundary of Σ^{τ} , running between endpoints of the singular leaves in $\hat{\Sigma}^{\tau}$, and such that $\bar{S}_i - S_i \subset \Sigma^{\tau}$ are made up of a finite number of compact singular leaves of \mathcal{F}^{τ} . But since we are assuming $\hat{\Sigma}^{\tau} \neq \Sigma^{\tau}$, there must be at least one of these which connects two singular points in the interior of Σ^{τ} . As we took a representative of \mathcal{F}^{τ} with one singular point for each connected component of the compact singular leaves, we see that this implies there must be a cycle of singular leaves which cannot define a combinatorial structure.

We observe that, even though the foliation associated to \mathbb{G} has no saddle connections, they may occur for \mathcal{F}^{τ} as a result of the gluing process. However, Proposition 2.31 together with the next result imply that this is generically not the case, and that it is never the case for \mathbb{G} -strictly small pairs of pants.

Lemma 2.32. Let \mathbb{G} be such that every vertex around γ_{-} has exactly one singular leaf reaching γ_{-} and no singular leaf reaching γ_{+} . Then all twists $\tau \in \mathbb{R}$ are admissible. The same statement holds if we exchange γ_{-} and γ_{+} .

Proof. Under such hypothesis, all smooth leaves starting from γ_{-} end at a boundary component of Σ that is neither γ_{-} nor γ_{+} . Therefore for every twist τ , we glue the singular leaves to leaves of γ_{+} which immediately reach a boundary component of $\partial \Sigma^{\tau}$. Gluing the singular leaves of γ_{-} can result in the creation of a leaf which returns back to γ_{+} again. This however corresponds two gluing to leaves of γ_{+} which immediately reach a boundary component of $\partial \Sigma^{\tau}$.

Corollary 2.33. Let Σ be a bordered surface of Euler characteristic $\leqslant -2$, take $[P] \in \mathcal{P}_{\Sigma,\mathfrak{m}_0}$ for some $\mathfrak{m}_0 \in \{1, \ldots, n\}$, $\mathbb{G} \in \mathbb{T}_{\Sigma}^{\text{comb}}$, and consider the operation of cutting along $\partial P \cap \Sigma^\circ$, twisting and gluing back. If P is \mathbb{G} -strictly small, then any twist $\tau \in \mathbb{R}^{\pi_0(\partial P \cap \Sigma^\circ)}$ is admissible. The same is true if $\Sigma = T$ and we self-glue after twisting the pair of pants obtained from T by cutting along $\gamma \in S_T^\circ$.

Proposition 2.34. For $\mathbb{G} \in \mathcal{T}_{\Sigma}^{\text{comb}}$ with $l = \ell_{\mathbb{G}}(\gamma_{-}) = \ell_{\mathbb{G}}(\gamma_{+})$ the set of admissible twists is an open dense subset of \mathbb{R} with countable complement.

We need the result of the following lemma before proving the proposition. Let $S^- = \{\ell_1^-, \dots, \ell_M^-\}$ and $S^+ = \{\ell_1^+, \dots, \ell_N^+\}$ be the finite sets of lengths of edges in \mathbb{G} into which γ_- and γ_+ decompose into respectively. Without loss of generality, assume that p_- and p_+ are contained in singular leaves of \mathbb{G} .

Lemma 2.35. If we choose the points p_{\pm} such that $\tau = 0$ identifies two singular leaves, then for $\tau \notin \text{span}_{\mathbb{Q}}(S^- \cup S^+)$, \mathfrak{F}^{τ} has a Whitehead representative with no singular leaf between two singularities.

Proof. Denote by γ the curve in Σ^{τ} which is the image of γ_{\pm} , and identify $\gamma \sim \mathbb{R}/\mathbb{IZ}$ where $l = \ell_{\mathbb{G}}(\gamma_{-}) = \ell_{\mathbb{G}}(\gamma_{+})$ and 0 corresponds to a singular leaf on the γ_{+} side. Consider a point $p \in \gamma$. If we follow along the leaf passing through p (in either the γ_{+} side or the γ_{-} side of the glued surface) until it gets back to γ at a point p', we find that there exists some $\mathbb{R} \in \operatorname{span}_{\mathbb{Z}}(\mathbb{S}^{-} \cup \mathbb{S}^{+})$ for each of the following cases, such that

- p' = -p + R, for a leaf going from γ_+ to γ_+ ;
- $p' = -p + 2\tau + R$, for a leaf going from γ_- to γ_- ;
- $p' = p + \tau + R$, for a leaf going from γ_+ to γ_- ;
- $p' = p \tau + R$, for a leaf going from γ_- to γ_+ .

Indeed, we firstly notice that all singular leaves on the γ_+ side are identified as some points in span_Z(S⁺), while on the γ_- side they are identified as some points in $\tau + \text{span}_{Z}(S^-)$.

Suppose first that p' is obtained from p by following a leaf going from γ_+ to γ_+ (see Figure 17a). We notice that p is given by $p = R_0 + a$, where $R_0 \in \text{span}_{\mathbb{Z}}(S^+)$ is the distance from the chosen singular leaf at 0 to the singular leaf just before p on the γ_+ side, and $a \ge 0$. Then, following the leaf, we find that the singular leaf just before p becomes the singular leaf just after p', and $p' = R_1 - a = -p + (R_0 + R_1)$ where $R_1 \in \text{span}_{\mathbb{Z}}(S^+)$ is the distance from the chosen singular leaf at 0 to the leaf just after p' (following the orientation of γ_+). In particular, we obtain the claim with $R = R_0 + R_1 \in \text{span}_{\mathbb{Z}}(S^+)$.

Similarly, suppose now that p' is obtained from p by following a leaf going from γ_- to γ_+ (see Figure 17b). Now we have $p = R_0 + \tau + a$, where $R_0 \in \text{span}_{\mathbb{Z}}(S^-)$ and a > 0 (here the singular leaf just before p on the γ_- side is at distance $R_0 + \tau$ from the chosen singular leaf at 0). Then, following the leaf, we find that the singular leaf at $R_0 + \tau$ is identified with a singular leaf at distance $R_1 \in \text{span}_{\mathbb{Z}}(S^+)$ from the chosen singular leaf at 0. Therefore $p' = R_1 + a = p - \tau + (R_0 - R_1)$. Thus, the claim with $R = R_0 - R_1 \in \text{span}_{\mathbb{Z}}(S^- \cup S^+)$.

The other cases follow similarly. Now, if p is a point at a singular leaf on the γ_{\pm} side, we see by induction that, after gluing, the singular leaf passes through γ at some other points of the form $\pm(p \mp n\tau) + R$ for some $n \in \mathbb{Z}_+$ and $R \in \text{span}_{\mathbb{Z}}(S^- \cup S^+)$. This implies that, if \mathbb{G}^{τ} has two singular points connected by a leaf, then a non-zero integral multiple of τ is contained in $\text{span}_{\mathbb{Z}}(S^- \cup S^+)$, or equivalently $\tau \in \text{span}_{\mathbb{Q}}(S^- \cup S^+)$. \Box

Proof of Proposition 2.34. We first show that the set of admissible twists is an open subset of \mathbb{R} . Consider $\tau \in \mathbb{R}$ an admissible twist, and denote by \mathbb{G}^{τ} the associated combinatorial structure. For each edge *e* of \mathbb{G}^{τ} , let $\mathfrak{n}_{\gamma}(e)$ be the number of times which γ travels through the edge *e*. If we take τ' such that $|\tau' - \tau| < \epsilon$, then we see that the distance between the two singularities of the foliation changes by at most $\epsilon \mathfrak{n}_{\gamma}(e)$ (*cf.* Figure 18). This also holds at the boundary of the cells, when we have vertices of higher valency whose original distance would be zero. Therefore, if we choose $\epsilon > 0$ smaller than

$$\min_{e} \frac{\ell_{\mathbb{G}}(e)}{n_{\gamma}(e)}$$

where *e* runs over the edges of \mathbb{G}^{τ} visited by γ , then all singularities stay at the same or at a positive distance from each other. These lengths are realised by curves homotopic to the original length realising curve. As a consequence, $\mathcal{F}^{\tau'}$ cannot admit a cycle of singular leaves connecting singularities, and thus τ' is an admissible twist. The countable complement property follows from Lemma 2.35.

We remark that the set of non admissible twists can have accumulation points, and its set of accumulation points can be non isolated. However what is crucial for the next section is that the non-admissible twists form a measure-zero set in \mathbb{R} .

2.5 Combinatorial Fenchel–Nielsen coordinates

With the notions of cutting and gluing in the combinatorial spaces defined, we have available the key tools to attempt a definition of Fenchel–Nielsen coordinates analogously to the hyperbolic case. In fact Fenchel–Nielsen coordinates can be defined, the main difference with the hyperbolic case being that the image of such a coordinate system does not cover the whole codomain. This is due to the fact that combinatorially not all twists are allowed.

2.5.1 Seams and pants decompositions

We can now define global coordinates on the combinatorial Teichmüller space, analogous to the hyperbolic Fenchel–Nielsen coordinates. Firstly, we need a technical ingredient, the pants seams, that allows us to define a canonical way of gluing pairs of pants.

Definition 2.36. Consider a combinatorial marking (**G**, f) on a pair of pants P, with associated foliation \mathcal{F} . Define the *combinatorial seam* connecting two distinct boundary components γ and γ' of P to be the quasitransverse arc connecting γ and γ' , as indicated in Figure 19. In the cases 19c–19g, the seams are smooth leaves, located at exactly the same distance from the adjacent singular leaves.

A notion of pants seams connecting a boundary component of P to itself can be found in [22, Section 6.3]. We remark that the combinatorial seam realises the minimum among lengths of all essential arcs connecting one boundary component to another (the length can be zero). For a point $(L_1, L_2, L_3) \in \mathbb{R}^3_+ \cong \mathbb{T}_p^{\text{comb}}$, the length of the seam connecting $\partial_1 P$ and $\partial_2 P$ is given by the formula

$$\ell_{\rm comb}(L_1, L_2, L_3) = \left[\frac{L_3 - L_1 - L_2}{2}\right]_+,$$
(2.16)

while the length of a seam connecting $\partial_1 P$ to itself is given by

$$\ell_{\text{comb}}(L_1, L_2, L_3) = \left[\frac{L_2 + L_3 - L_1}{2}\right]_+ + \left[\frac{L_2 - L_1 - L_3}{2}\right]_+ + \left[\frac{L_3 - L_1 - L_2}{2}\right]_+$$

$$= \max\left\{\frac{L_2 + L_3 - L_1}{2}, L_2 - L_1, L_3 - L_1, 0\right\}$$
(2.17)

Notice that for a hyperbolic marking (X, φ) on P, there exist a notion of hyperbolic seam connecting γ and γ' , that is the shortest geodesic arc connecting the boundary components γ and γ' . On the other hand, we can consider the combinatorial marking (**G**, f) on P associated to (X, φ) defined through the spine map of Definition 2.13. The next elementary lemma, of which we omit the proof, shows that the hyperbolic and combinatorial seams are the same arcs.

Lemma 2.37. Consider a hyperbolic marking (X, φ) on P, and the associated combinatorial marking (G, f). Through their aforementioned identification, the hyperbolic and combinatorial seams connecting two boundary components of P coincide.

Remark 2.38. In the hyperbolic case, for a point $(L_1, L_2, L_3) \in \mathbb{R}^3_+ \cong \mathcal{T}_P$, the hyperbolic length of the seam connecting $\partial_1 P$ and $\partial_2 P$ is given by the formula

$$\cosh\left(\ell_{\text{hyp}}(L_1, L_2, L_3)\right) = \frac{\cosh\left(\frac{L_3}{2}\right)}{\sinh\left(\frac{L_1}{2}\right)\sinh\left(\frac{L_2}{2}\right)} + \operatorname{cotanh}\left(\frac{L_1}{2}\right)\operatorname{cotanh}\left(\frac{L_2}{2}\right), \tag{2.18}$$

while the hyperbolic length of a seam connecting $\partial_1 P$ to itself is given by

$$\cosh^{2}\left(\frac{\ell_{\text{hyp}}(L_{1},L_{2},L_{3})}{2}\right) = \frac{\cosh^{2}\left(\frac{L_{1}}{2}\right) + \cosh^{2}\left(\frac{L_{2}}{2}\right) + \cosh^{2}\left(\frac{L_{3}}{2}\right) + 2\cosh\left(\frac{L_{1}}{2}\right)\cosh\left(\frac{L_{2}}{2}\right)\cosh\left(\frac{L_{3}}{2}\right) - 1}{\sinh^{2}\left(\frac{L_{1}}{2}\right)}.$$

$$(2.19)$$

Formulae (2.16)–(2.17) and can be recovered from equations (2.18)–(2.19) by taking the following limit:

$$\lim_{\beta \to \infty} \frac{\ell_{\text{hyp}}(\beta L_1, \beta L_2, \beta L_3)}{\beta} = \ell_{\text{comb}}(L_1, L_2, L_3).$$
(2.20)

This fact is revisited and generalised in Section 5 where such a limit is shown to hold for many other expressions.

We can now define the seamed pants decomposition associated to a bordered surface Σ , consisting of a

pants decomposition of Σ together with a collection of curves and arcs.

Definition 2.39. Given a bordered surface Σ of type (g, n), a seamed pant decomposition consists of

- a pants decomposition 𝒫 = (γ₁,..., γ_{3g-3+n}), that is a maximal collection of pairwise non-homotopic, essential, simple closed curves, labelled by 1,..., 3g 3 + n,
- a collection *S* of non-homotopic, essential simple closed curves or simple arcs connecting boundary components of Σ, pairwise non-homotopic relative to the boundary, such that the intersection of *S* with any of the pair of pants P in the decomposition specified by *P* is a union of three disjoint arcs connecting the boundary components of P pairwise.

Notice that, given \mathcal{P} , we can construct an \mathcal{S} by first choosing three disjoint arcs on each pair of pants and then matching up endpoints in any fashion.

2.5.2 Existence of combinatorial Fenchel–Nielsen coordinates

Fix once and for all a seamed pants decomposition $(\mathcal{P}, \mathcal{S})$ on Σ . We define the *length parameters* of a point $\mathbb{G} \in \mathfrak{T}_{\Sigma}^{\text{comb}}$ to be the tuple of positive real numbers

$$\ell(\mathbb{G}) = \left(\ell_1(\mathbb{G}), \dots, \ell_{3g-3+n}(\mathbb{G})\right),\tag{2.21}$$

where $\ell_i(\mathbb{G}) = \ell_{\mathbb{G}}(\gamma_i)$.

As a first step towards the definition of twist parameters, consider a combinatorial marking (**G**, f) of a pair of pants P and an arc α connecting two distinct boundary components γ and γ' of P. Let δ be the combinatorial seam connecting γ and γ' – which depends only on (**G**, f). Let \tilde{P} be a universal cover of P. It contains lifts $\tilde{\gamma}$ and $\tilde{\gamma}'$ of γ and γ' respectively. Notice that they acquire orientation from \tilde{P} . Let $d = \delta \cap \gamma$ and $a = \alpha \cap \gamma$. We choose a lift \tilde{d} of d and call \tilde{a} the first lift of a met by travelling from \tilde{d} along $\tilde{\gamma}$ following its orientation. This determines lifts $\tilde{\delta}$ (resp. $\tilde{\alpha}$) of δ (resp. α) starting from \tilde{d} (resp. \tilde{a}). Now let $\tilde{d}' = \tilde{\delta} \cap \tilde{\gamma}'$ and $\tilde{a}' = \tilde{\alpha} \cap \tilde{\gamma}'$. Consider the path $c_{\tilde{d}'\tilde{a}'}$ along $\tilde{\gamma}'$ starting at \tilde{d}' and ending at \tilde{a}' . The measured foliation associated to (**G**, f) lifts to a measured foliation $\tilde{\mathcal{F}}$ on the universal cover, and we can measure the length of $c_{\tilde{d}'\tilde{a}'}$. We then set $\text{sgn}(c_{\tilde{d}'\tilde{a}'}) = \pm 1$ depending on whether the orientation of $c_{\tilde{d}'\tilde{a}'}$ agrees with the one of $\tilde{\gamma}'$. We define the twisting number of α along γ' in P to be

$$\mathbf{t}_{\alpha,\gamma'}^{\mathsf{P}}(\mathbf{G},\mathsf{f}) = \operatorname{sgn}(\mathbf{c}_{\tilde{\mathfrak{d}}'\tilde{\mathfrak{a}}'})\,\ell_{\tilde{\mathcal{F}}}(\mathbf{c}_{\tilde{\mathfrak{d}}'\tilde{\mathfrak{a}}'}). \tag{2.22}$$

The definition does not depend on the choice of d, since all different choices are related by deck transformations which leave $t^{P}_{\alpha,\nu'}(\mathbf{G}, f)$ fixed.

Given $\mathbb{G} \in \mathbb{T}_{\Sigma}^{comb}$, we define the i-th twist parameter $\tau_i(\mathbb{G})$ as follows. Fix a marking (\mathbf{G}, f) such that γ_i is quasitransverse to the measured foliation induced by the marking. Let α_i be one of the two arcs in \mathscr{S} crossing γ_i . There are two pairs of pants Q'_i and Q''_i (possibly the same) on each side of γ_i , and α_i determines two arcs $\alpha'_i = \alpha_i \cap Q'_i$ and $\alpha''_i = \alpha_i \cap Q''_i$. The i-th twist parameter of \mathbb{G} is defined to be

$$\tau_{i}(\mathbb{G}) = t_{\alpha'_{i},\gamma_{i}}^{Q'_{i}}(\mathbf{G}|_{Q'_{i}}, f|_{Q'_{i}}) + t_{\alpha''_{i},\gamma_{i}}^{Q''_{i}}(\mathbf{G}|_{Q''_{i}}, f|_{Q''_{i}}).$$
(2.23)

This twist parameter is invariant under isotopies, *i.e.* does not depend on the representative of G. Besides, it does not depend on the choice of the arc in \mathscr{S} crossing γ_i . This can be seen by passing to the universal cover of a neighbourhood of $\gamma_i - cf$. [21, Section 10.6.1] for the analogue in the hyperbolic case. Finally, it only depends on the homotopy class of α_i , since a different choice of representative would modify both t' and t" by the same quantity, but with different signs. Thus, we have well-defined *twist parameters*

$$\tau(\mathbb{G}) = (\tau_1(\mathbb{G}), \dots, \tau_{3g-3+n}(\mathbb{G})). \tag{2.24}$$

Notice that we may homotope the representative of α_i such that it is intersecting γ_i at a vertex of the combinatorial structure, showing that t' and t" in (2.23) can be expressed as a sum of edge lengths in G with half-integer coefficients. The half is coming from the definition of combinatorial seams, which were required to be equidistant from the adjacent singular leaves in the cells depicted in Figure 19c–19g.

Definition 2.40. Let Σ be a bordered surface of type (g, n) equipped with a seamed pants decomposition $(\mathcal{P}, \mathcal{S})$. *Combinatorial Fenchel–Nielsen coordinates* relative to $(\mathcal{P}, \mathcal{S})$ is by definition the map $\Phi_L \colon \mathcal{T}_{\Sigma}^{\text{comb}}(L) \to \mathbb{R}^{3g-3+n}_+ \times \mathbb{R}^{3g-3+n}$ defined by

$$\Phi_{\mathsf{L}}(\mathbb{G}) = \big(\ell(\mathbb{G}), \tau(\mathbb{G})\big). \tag{2.25}$$

Using the gluing we can establish the following result.

Theorem 2.41. *For any* $L \in \mathbb{R}^{n}_{+}$ *, the map*

$$\Phi_{\mathsf{L}}:\mathfrak{T}_{5}^{\mathrm{comb}}(\mathsf{L})\to(\mathbb{R}_{+}\times\mathbb{R})^{3g-3+\mathfrak{n}}$$
(2.26)

is a homeomorphism onto its image, which is an open dense subset whose complement has zero measure. Moreover, if $\Phi_{L}(\mathbb{G}) = (\ell, \tau)$, then the image of the map τ_{i} restricted to

$$\left\{ \mathbb{G} \in \mathfrak{T}_{\Sigma}^{\mathrm{comb}}(\mathsf{L}) \mid \ell_{j}(\mathbb{G}) = \ell_{j} \; \forall j, \; \tau_{\mathsf{k}}(\mathbb{G}) = \tau_{\mathsf{k}} \text{ for } \mathsf{k} \neq \mathfrak{i} \right\}$$
(2.27)

has a complement of zero measure in \mathbb{R} .

Proof. To prove the theorem, we use the gluing to construct a partial inverse map. More precisely, define the partial map Ψ_L : $(\mathbb{R}_+ \times \mathbb{R})^{3g-3+n} \to \mathcal{T}_{\Sigma}^{\text{comb}}(L)$ by setting

$$\Psi_{\mathsf{L}}(\ell, \tau) = \mathbb{G},$$

where G is defined as follows.

- For each pair of pants bounded by curves in \mathcal{P} , we assign boundary lengths defined by the Ls and ℓ s. This determines a unique combinatorial structure on each pair of pants.
- We glue the pairs of pants along each γ_i after twisting by τ_i. The twist zero corresponds to gluing the combinatorial seams of the pairs of pants together.

By partial map we mean that Ψ_L is not defined on the whole of $(\mathbb{R}_+ \times \mathbb{R})^{3g-3+n}$, as the gluing does not always define an embedded metric ribbon graph. Notice also that Ψ_L does not depend on the order on which we glue the pairs of pants together.

We proceed now with the proof. Firstly notice that the definition of the twist parameters implies that gluing with twist zero amounts to gluing all pairs of pants with matching seams. Also, since gluing back a cut combinatorial structure gives back the original one, we can see that Ψ_L is defined on the image of Φ_L and that $\Psi_L \circ \Phi_L$ is the identity on $\mathfrak{T}_{\Sigma}^{comb}(L)$. Hence, Φ_L is a bijection onto its image.

On the closure of each open cell, the length and twists are linear functions of the edge lengths. Therefore, we have bijective linear functions that agree on boundaries of the open cells, and therefore, the inverse has the same properties and is therefore continuous which shows that Φ_L is a homeomorphism onto it's image.

Now to prove openess we note that given any point $\mathbb{G} \in \mathcal{T}_{\Sigma}^{comb}(L)$ there exists a neighbourhood intersecting finitely many cells as there are finitely many ways to expand a singularity using Whitehead equivalence. We can therefore construct a finite simplicial complex containing \mathbb{G} as a vertex such that the intersection of a k-cell of $\mathcal{T}_{\Sigma}^{comb}(L)$ is a union of k-dimensional simplices. Then, as Φ_L is linear on each cell and a homeomorphism onto it's image, Φ_L maps the simplicial complex to a simplicial complex in $(\mathbb{R}_+ \times \mathbb{R})^{3g-3+n}$.

A point on a finite simplicial complex in an Euclidean space of the same dimension is on the boundary if and only if it is contained in a codimension one simplex that is on the boundary of only one top-dimensional simplex. Every codimension one simplex containing \mathbb{G} is contained in two top-dimensional simplices and therefore $\Phi_{L}(\mathbb{G})$ is in the interior of the image of the simplex. Thus, Φ_{L} is open.

Finally, notice that by gluing one curve at a time and using Proposition 2.31 and Lemma 2.35 for each L and ℓ , we can see that the image is dense and its complement has zero measure.

2.6 A combinatorial (9g - 9 + 3n)-theorem

In this paragraph we establish a combinatorial analogue of the hyperbolic (9g-9+3n)-theorem (see *e.g.* [21, Theorem 10.7]), that is, any combinatorial structure can be reconstructed from the data of the combinatorial lengths of (9g - 9 + 3n) simple closed curves. Partial computations in this direction were done in [22, Exposé 6]). These results are used in Section 5.1.2 to compare the hyperbolic and combinatorial twists, and in Section 5.2 to prove that combinatorial Fenchel–Nielsen coordinates transform in a piecewise linear way under a change of pants decompositions.

Let Σ be of type (g, n) and fix a seamed pants decomposition $(\mathcal{P}, \mathcal{S})$, with $\mathcal{P} = (\gamma_1, \dots, \gamma_{3g-3+n})$. The union of the pair of pants in the decomposition that are adjacent to γ_i is a surface of type (0, 4) or (1, 1), and we choose $\alpha_i \in \mathcal{S}$ crossing γ_i in Σ_i . We choose some order on the boundaries such that α_i connects $\partial_1 X$ to $\partial_2 X$ and $\partial_4 X$ is in the same pair of pants as $\partial_1 X$. We now define two other homotopy classes of curves in Σ_i (see Figure 20).

- If Σ_i has type (0,4), we let δ_i be the curve determined by a tubular neighbourhood of α_i union the boundary component it connects. If Σ_i has type (1, 1), we let δ_i be the curve α_i.
- Let η_i be the image of δ_i after a positive Dehn twist along γ_i .

In the (0,4) case there are two possible choices of α_i as above but both choices give the same (δ_i, η_i) .

Theorem 2.42. Let Σ be a bordered surface of type (g, n) and $(\mathcal{P}, \mathcal{S})$ a seamed pants decomposition. The following map is continuous and injective:

$$\begin{aligned} \mathcal{T}_{\Sigma}^{\text{comb}}(\mathsf{L}) &\longrightarrow \mathbb{R}_{+}^{9g-9+3n} \\ & \mathbb{G} &\longmapsto (\ell_{\mathbb{G}}(\gamma), \ell_{\mathbb{G}}(\delta), \ell_{\mathbb{G}}(\eta)). \end{aligned}$$
(2.28)

As a preparation to the proof, we present in Lemmata 2.43 and 2.45 closed formulae for $\ell_{\mathbb{G}}(\delta_i)$ and $\ell_{\mathbb{G}}(\eta_i)$ in the (0,4) and (1,1) cases respectively. For this purpose we can work locally on $\mathbb{G}|_{\Sigma_i}$ with a fixed seamed pants decomposition, which we denote by $\ell_i = \ell_{\mathbb{G}}(\gamma_i)$, $\ell'_i = \ell_{\mathbb{G}}(\delta_i)$ and $\ell''_i = \ell_{\mathbb{G}}(\eta_i)$.

2.6.1 Four-holed sphere

Let $\Sigma_i = X$ be a four-holed sphere. We remove the index i from the notation of γ_i , α_i , δ_i and η_i , as well as ℓ_i , ℓ'_i , ℓ''_i and τ_i . Label by $\partial_1 X, \ldots, \partial_4 X$ the boundary components of X, so that γ is separating the components $\partial_1 X$ and $\partial_4 X$ from $\partial_2 X$ and $\partial_3 X$, and α is connecting the components $\partial_1 X$ and $\partial_2 X$. Finally, denote $L_i = \ell_G(\partial_i X)$.

Lemma 2.43. In the above setting, we have

$$\ell'(\ell,\tau) = \max\left\{L_1 + L_3 - \ell, L_2 + L_4 - \ell, 2|\tau| + M_{1,4}(\ell) + M_{2,3}(\ell)\right\},$$
(2.29)

where $M_{i,j}(\ell) = \max\left\{0, L_i - \ell, L_j - \ell, \frac{L_i + L_j - \ell}{2}\right\}$. Further, $\ell''(\ell, \tau) = \ell'(\ell, \tau + \ell)$.

Proof. Let us assume that the ribbon graph underlying \mathbb{G} is trivalent, and fix a marking of it. Let us cut \mathbb{G} along the curve γ . There are sixteen possibilities for the cut combinatorial structure: the marked ribbon graph on each pair of pants can belong to each of the four top-dimensional cells of the Teichmüller space of

a pair of pants. Therefore, in order to check that Equation (2.29) holds for any \mathbb{G} , it is sufficient to check that it is satisfied in each of the sixteen cases. By symmetry considerations, the number of cases can actually be reduced to seven. We show the detailed argument for three particularly representative cases out of those seven, and argue that the other cases can be proven following the same strategy.

Firstly, suppose that $L_4 > L_1 + \ell$ and $L_3 > L_2 + \ell$ (see Figure 21a). Then

$$\max\left\{0, L_1 - \ell, L_4 - \ell, \frac{L_1 + L_4 - \ell}{2}\right\} = L_4 - \ell, \qquad \max\left\{0, L_2 - \ell, L_3 - \ell, \frac{L_2 + L_3 - \ell}{2}\right\} = L_3 - \ell,$$

so the right-hand side of Equation (2.29) reduces to

 $max\left\{L_1+L_3-\ell,L_2+L_4-\ell,2|\tau|+L_3+L_4-2\ell\right\}=2|\tau|+L_3+L_4-2\ell.$

In Figure 21b, a quasitransverse representative of δ is shown. Its orange part has length $L_4 - \ell$, its blue part has length $L_3 - \ell$, and its green part has length $2|\tau|$. In the end, we have

$$\ell' = 2|\tau| + L_3 + L_4 - 2\ell,$$

which is consistent with Equation (2.29).

Secondly, suppose that $L_4 > L_1 + \ell$ and $|L_2 - L_3| < \ell < L_2 + L_3$ (see Figure 21c). Then

$$\max\left\{0, L_1 - \ell, L_4 - \ell, \frac{L_1 + L_4 - \ell}{2}\right\} = L_4 - \ell, \qquad \max\left\{0, L_2 - \ell, L_3 - \ell, \frac{L_2 + L_3 - \ell}{2}\right\} = \frac{L_2 + L_3 - \ell}{2}.$$

In this case, we also have $L_1 + L_3 < L_2 + L_4$, so the right-hand side of Equation (2.29) reduces to

$$\max\left\{L_1+L_3-\ell, L_2+L_4-\ell, 2|\tau|+L_4-\ell+\frac{L_2+L_3-\ell}{2}\right\}=L_2+L_4-\ell+\left[2|\tau|-\frac{L_2+\ell-L_3}{2}\right]_+.$$

Suppose first that $2|\tau| < \frac{L_2 + \ell - L_3}{2}$, which is depicted in Figure 21d together with a quasitransverse representative of δ . The orange part of δ has length $L_4 - \ell$, while the blue part of δ has length L_2 . Therefore

$$\ell' = L_2 + L_4 - \ell = L_2 + L_4 - \ell + \left[2|\tau| - \frac{L_2 + \ell - L_3}{2}\right]_+$$

Suppose now that $2|\tau| > \frac{L_2 + \ell - L_3}{2}$, see Figure 21e. The orange part of δ has length $L_4 - \ell$, the blue part of δ has length $\frac{L_2 + L_3 - \ell}{2}$, and the green part of δ has length $2|\tau|$. Thus:

$$\ell' = 2|\tau| + L_4 - \ell + \frac{L_2 + L_3 - \ell}{2} = L_2 + L_4 - \ell + \left[2|\tau| - \frac{L_2 + \ell - L_3}{2}\right]_+$$

Again, in both cases Equation (2.29) is satisfied.

Figure 7: Possible models for points in a foliation. The singular leaves are depicted in blue, while smooth leaves in grey. Only singular points of low valency are depicted, but any higher valency is allowed.

Figure 9: The geometric realisation of a metric ribbon graph (left) and the associated measured foliation. The edges of the embedded graph are depicted in red, the singular leaves of the associated foliation in blue.

Figure 10: The length of a simple closed curve γ with respect to a combinatorial structure $\mathbb G$ on a pair of pants. We have $\ell_{\mathbb G}(\gamma)=a+c+b+c=a+b+2c.$

Figure 11: Arcs and pairs of pants: $Q_{\mathfrak{m}_0}(\alpha) = Q_{\mathfrak{m}_0}(\alpha') = Q_{\mathfrak{m}_0}(-\alpha') = [P].$

Figure 12: Three different cells in $\mathcal{T}_{P}^{comb}.$

Figure 13: The three cases examined in Lemma 2.30.

Figure 14: Cutting/gluing algorithm: the combinatorial structure \mathbb{G} (in red) pictured with the singular leaves (in blue), and the curve γ (in green).

Figure 15: Cutting/gluing algorithm for vertices of higher valency. Two Whitehead moves are performed.


Figure 16: A glued measured foliation that is not dual to a combinatorial structure. Notice in grey $\hat{\Sigma}^{\tau}$, that is properly contained in Σ^{τ} , and the presence of saddle connections on the boundary of $\hat{\Sigma}^{\tau}$ that cannot be removed by Whitehead moves.



Figure 17: Examples of leaf dynamics, induced on γ by the foliation \mathcal{F} . The singular leaves before gluing are depicted in blue, the leaf connecting p and p' in grey (here it is depicted as a smooth leaf, *i.e.* a > 0). In between the two fully depicted singular leaves there is a strip of smooth leaves all homotopic to each other.



Figure 18: A combinatorial structure \mathbb{G} glued to \mathbb{G}^{τ} and $\mathbb{G}^{\tau'}$ for $|\tau - \tau'|$ small. The singular leaves of \mathbb{G} are shown in blue. After gluing, they are prolonged with the purple leaves in \mathbb{G}^{τ} , and with the light blue leaves in $\mathbb{G}^{\tau'}$. The dotted lines indicate the identification of γ_{-} and γ_{+} for τ and τ' .



Figure 19: Combinatorial seams (in orange) on each cell of \mathcal{T}_{P}^{comb} . The singular leaves are depicted in blue.



Figure 20: The curves δ and η (we omit the subscript).





Thirdly, suppose that $\ell > L_1 + L_4$ and $|L_2 - L_3| < \ell < L_2 + L_3$ (see Figure 21f). Then

$$\max\left\{0, L_1 - \ell, L_4 - \ell, \frac{L_1 + L_4 - \ell}{2}\right\} = 0, \qquad \max\left\{0, L_2 - \ell, L_3 - \ell, \frac{L_2 + L_3 - \ell}{2}\right\} = \frac{L_2 + L_3 - \ell}{2}.$$

Without loss of generality, we can assume that $L_1 + L_3 > L_2 + L_4$. Then, the right-hand side of Equation (2.29) reduces to

$$\max\left\{L_1 + L_3 - \ell, L_2 + L_4 - \ell, 2|\tau| + \frac{L_2 + L_3 - \ell}{2}\right\} = \max\left\{L_1 + L_3 - \ell, 2|\tau| + \frac{L_2 + L_3 - \ell}{2}\right\}$$

The case where $2|\tau| < L_1 + \frac{L_3 - L_2 - \ell}{2}$ is depicted in Figure 21g. The length of δ is then

$$\ell' = L_1 + L_3 - \ell = \max\left\{L_1 + L_3 - \ell, 2|\tau| + \frac{L_2 + L_3 - \ell}{2}\right\}.$$

In the case where $2|\tau| > L_1 + \frac{L_3 - L_2 - \ell}{2}$, depicted in Figure 21h, the orange part of δ has length $\frac{L_2 + L_3 - \ell}{2}$, and the green part of δ has length $2|\tau|$, therefore

$$\ell' = 2|\tau| + \frac{L_2 + L_3 - \ell}{2} = \max\left\{L_1 + L_3 - \ell, 2|\tau| + \frac{L_2 + L_3 - \ell}{2}\right\}$$

In both cases, Equation (2.29) is satisfied.

The case of \mathbb{G} with higher valencies can be obtained from the trivalent case by continuity of the combinatorial lengths and twists. Lastly, since η is obtained from δ by performing a positive Dehn twist along γ , its length is given by $\ell'' = \ell'(\ell, \tau + \ell)$.

The above lemma expresses the lengths ℓ' and ℓ'' as functions of the Fenchel–Nielsen coordinates (ℓ, τ) . We can invert the perspective, expressing τ as a function of ℓ , ℓ' and ℓ'' .

Corollary 2.44. In the previous situation, we have

$$\tau = \begin{cases} \frac{1}{2}(\ell'' - M_{1,4} - M_{2,3}) - \ell & \text{if } \ell' = \max\{L_1 + L_3 - \ell, L_2 + L_4 - \ell\}, \\ -\frac{1}{2}(\ell' - M_{1,4} - M_{2,3}) & \text{if } \ell'' = \max\{L_1 + L_3 - \ell, L_2 + L_4 - \ell\}, \\ \frac{1}{2\ell} \left(\frac{\ell'' - M_{1,4} - M_{2,3}}{2}\right)^2 - \frac{1}{2\ell} \left(\frac{\ell' - M_{1,4} - M_{2,3}}{2}\right)^2 - \frac{\ell}{2} & \text{otherwise.} \end{cases}$$

$$(2.30)$$

Proof. Let us denote $p = 2|\tau| + M_{1,4} + M_{2,3}$ and $q = \max\{L_1 + L_3 - \ell, L_2 + L_4 - \ell\}$, so that $\ell' = \max\{p, q\}$. We claim that $\ell' = q$ implies $2|\tau| \leq \ell$. If $L_2 + L_4 \geq L_1 + L_3$, this comes from the observation that $q = L_2 + L_4 - \ell$ and $\ell' = q + [\lambda]_+$ with

$$\begin{split} \lambda &= p - L_2 - L_4 + \ell \\ &= 2|\tau| - \ell + max \left\{ \ell - L_4, L_1 - L_4, 0, \frac{L_1 - L_4 + \ell}{2} \right\} + max \left\{ \ell - L_2, 0, L_3 - L_2, \frac{L_3 - L_2 + \ell}{2} \right\} \geqslant 2|\tau| - \ell. \end{split}$$

If $L_1 + L_3 \ge L_2 + L_4$, we rather have $q = L_1 + L_3 - \ell$ and the claim follows by writing $\ell' = q + [\mu]_+$ with

$$\begin{split} \mu &= p - L_1 - L_3 + \ell \\ &= 2|\tau| - \ell + max \left\{ \ell - L_1, 0, L_4 - L_1, \frac{L_4 - L_1 + \ell}{2} \right\} + max \left\{ \ell - L_3, L_2 - L_3, 0, \frac{L_2 - L_3 + \ell}{2} \right\} \geqslant 2|\tau| - \ell. \end{split}$$



Figure 22: The case examined in the proof of Lemma 2.45.

Therefore, if $\ell' = \max\{L_1 + L_3 - \ell, L_2 + L_4 - \ell\}$, we must have $|\tau| \leq \ell/2$, hence $|\tau + \ell| = \tau + \ell$. From Equation (2.29) we then find

$$\ell'' = 2|\tau + \ell| + M_{1,4} + M_{2,3},$$

and solving for τ we get the first case of Equation (2.30). The case $\ell'' = \max\{L_1 + L_4 - \ell, L_2 + L_3 - \ell\}$ is similar. Finally, if none of those conditions are satisfied, then

$$\ell' = 2|\tau| + M_{1,4} + M_{2,3}, \qquad \ell'' = 2|\tau + \ell| + M_{1,4} + M_{2,3}.$$

This covers the last case in Equation (2.30).

2.6.2 One-holed torus

Let $\Sigma_i = T$ be a one-holed torus. We remove the index i from the notation of γ_i , α_i , δ_i and η_i , as well as ℓ_i , ℓ'_i , ℓ''_i and τ_i , and denote $L = \ell_{\mathbb{G}}(\partial T)$.

Lemma 2.45. In the above setting, we have

$$\ell'(\ell,\tau) = |\tau| + \left[\frac{L-2\ell}{2}\right]_+.$$
(2.31)

Further, $\ell''(\ell, \tau) = \ell'(\ell, \tau + \ell)$.

Proof. As before, we assume \mathbb{G} to be trivalent and we fix a marking. There are 4 cases, corresponding to the 4 open cells of the Teichmüller space of the pair of pants we obtain after cutting along γ .

We detail the case of Figure 22a, where $L > 2\ell$. In Figure 22b, a quasitransverse representative of δ is shown. Its orange part has length $L/2 - \ell$, its green part has length $|\tau|$. Thus, we find

$$\ell' = |\tau| + \frac{L}{2} - \ell,$$

which is consistent with Equation (2.31) under the assumption $L > 2\ell$.

The other cases are analogous. Again, the formula extends to higher valency by continuity, and since the curve η is obtained from the curve δ by performing a positive Dehn twist along γ , its length is given by $\ell'' = \ell'(\ell, \tau + \ell)$.

Again, we can recover from the above Lemma an expression for the twist parameter τ as a function of the lengths ℓ , ℓ' and ℓ'' . The proof is similar to the (1,1) case.

Corollary 2.46. In a one-holed torus T with the above setting, the twist parameter is given as a function of (ℓ, ℓ', ℓ'') by

$$\tau = \frac{1}{2\ell} \left(\ell'' - \left[\frac{L - 2\ell}{2} \right]_+ \right)^2 - \frac{1}{2\ell} \left(\ell' - \left[\frac{L - 2\ell}{2} \right]_+ \right)^2 - \frac{\ell}{2}.$$
(2.32)

2.6.3 Proof of the combinatorial (9g - 9 + 3n)-theorem

Proof of Theorem 2.42. The map is clearly continuous. Further, if \mathbb{G} , $\mathbb{G}' \in \mathbb{T}_{\Sigma}^{\text{comb}}(L)$ are mapped to the same vector of lengths, then Corollaries 2.44 and 2.46 would give the same length and twist parameters. As the combinatorial Fenchel–Nielsen map is a homeomorphism into the image, we deduce that $\mathbb{G} = \mathbb{G}'$. This justifies the injectivity.

3 The symplectic structure

The Kontsevich symplectic form was originally introduced by Kontsevich in [28] as an ingredient in the proof of Witten's conjecture. Its main feature is the connection with intersection theory of $\overline{\mathfrak{M}}_{g,n}$. In this section, we prove a combinatorial analogue of Wolpert's formula (Theorem 3.9) for the Kontsevich symplectic form, saying that Fenchel–Nielsen coordinates are Darboux. As a direct consequence, we achieve an explicit integration formula for a certain class of measurable functions on the combinatorial moduli space with respect to Kontsevich's volume form (Proposition 3.13). This integration is the key operation that connects topological and geometric recursion, and is the combinatorial analogue of Mirzakhani's integration lemma on the moduli space of bordered Riemann surfaces [32, Theorem 7.1].

3.1 Kontsevich's form

Consider a ribbon graph G of type (g, n). For each index $i \in \{1, ..., n\}$, we make the choice of a first edge $e_1^{[i]}$ on the i-th face of G. We label the edges around the i-th face by $e_1^{[i]}, ..., e_{N_i}^{[i]}$ following the orientation of the face, which is opposite to the orientation of the boundary. Notice that every edge has a double label, as it bounds two faces or it appears twice in the cycle of a single face.

Let now Σ be a connected bordered surface of type (g, n) and fix a combinatorial marking (G, f). On each cell of $\mathfrak{T}_{\Sigma}^{comb}$, we have functions $\ell_i^{[i]} \colon \mathfrak{Z}_{\Sigma,G}(L) \to \mathbb{R}_+$, which associates to each $\mathbb{G} \in \mathfrak{Z}_{\Sigma,G}(L)$ the length $\ell_{\mathbb{G}}(e_i^{[i]})$.

Definition 3.1. For each $i \in \{1, ..., n\}$, consider the differential 2-form Ψ_i on the cell complex $\mathcal{T}_{\Sigma}^{comb}(L)$, defined on each cell $\mathfrak{Z}_{\Sigma,G}(L)$ by

$$\Psi_{i} = \sum_{1 \leq k < m \leq N_{i}} \frac{d\ell_{k}^{[i]}}{L_{i}} \wedge \frac{d\ell_{m}^{[i]}}{L_{i}}.$$
(3.1)

The form is Mod_{Σ}^{∂} -invariant (it depends only on the ribbon graph underlying the marking) and we denote the induced form on the quotient $\mathcal{M}_{q,n}^{comb}(L)$ with the same symbol³.

It can be shown that the definition of Ψ_i does not depend on the choice of the first edge: the difference between two possible choices is of the form $L_i^{-2}dL_i \wedge \vartheta_i$ for some differential 1-forms ϑ_i , and thus is zero along the fibres $\mathcal{M}_{q,n}^{comb}(L)$ of the perimeter map.

In his original work [28], Kontsevich related the above differential forms to the geometry of a certain circle bundle over $\mathcal{M}_{g,n}^{comb}(L)$.

Definition 3.2. For each $i \in \{1, ..., n\}$, define S_i^{comb} as the space of ordered pairs (\mathbf{G}, q) where $\mathbf{G} \in \mathcal{M}_{g,n}^{comb}(L)$ and q is a point belonging to an edge that borders the i-th face of $|\mathbf{G}|$. Its topology is the one induced by the natural cell structure. This defines a topological circle bundle $S_i^{comb} \to \mathcal{M}_{q,n}^{comb}(L)$.

Theorem 3.3. [28, 46] The class $[\Psi_i] \in H^2(\mathcal{M}_{g,n}^{comb}(L))$ equals $-c_1(\mathbb{S}_i^{comb})$. Further, under the identification $\mathfrak{M}_{g,n} \cong \mathcal{M}_{g,n}^{comb}(L)$, the pullback of Ψ_i extends continuously to the Deligne–Mumford compactification $\overline{\mathfrak{M}}_{g,n}$ and the associated cohomology class equals $\psi_i = c_1(\mathcal{L}_i) \in H^2(\overline{\mathfrak{M}}_{g,n})$, where \mathcal{L}_i is the relative cotangent bundle at the *i*-th marked point.

Definition 3.4. Define the *Kontsevich* 2-*form* on $\mathcal{T}_{\Sigma}^{\text{comb}}(L)$ as

$$\omega_{\rm K} = \frac{1}{2} \sum_{i=1}^{n} L_i^2 \Psi_i.$$
(3.2)

³See [46, Section 5.2] for a discussion on the differential geometry of cell complexes. What we need here is that the combinatorial Teichmüller spaces and the combinatorial moduli spaces have a well-defined notion of polytopal differential forms, and that the associated polytopal de Rham cohomology groups coincide with the usual cohomology groups over \mathbb{R} . In particular, we can consider the cohomology class $[\Psi_i] \in H^2(\mathfrak{M}_{a,n}^{comb}(L))$.

Theorem 3.5. [28] The differential form ω_K is non-degenerate when restricted to strata corresponding to graphs with no vertices of even valency.

A *fortiori*, ω_{K} descends to a symplectic form on the top-dimensional stratum of $\mathcal{M}_{g,n}^{comb}(L)$, that is denoted with the same symbol.

Definition 3.6. Define the Kontsevich measure

$$d\mu_{K} = \frac{\omega_{K}^{d_{g,n}}}{d_{g,n}!}, \qquad d_{g,n} = 3g - 3 + n.$$
(3.3)

Strictly speaking, $d\mu_K$ is not a volume form on the whole $\mathcal{M}_{g,n}^{comb}(L)$, although it is a volume form on the top-dimensional stratum. In any case, we have a notion of volume

$$V_{g,n}^{K}(L) = \int_{\mathcal{M}_{g,n}^{comb}(L)} d\mu_{K}$$
(3.4)

where the integral is taken over the top-dimensional stratum. Equivalently, if G is a trivalent ribbon graph of type (g, n) and $P_G(L) \subseteq \mathbb{R}_+^{E_G}$ is the polytope corresponding to those metrics on G with fixed perimeter $L \in \mathbb{R}_+^n$, then we set

$$V_{g,n}^{K}(L) = \sum_{\substack{G \in \mathcal{R}_{g,n} \\ \text{trivalent}}} \frac{1}{\# \text{Aut}(G)} \int_{P_{G}(L)} d\mu_{K}.$$
(3.5)

We remark that, by abuse of notation, we are using the same symbol to denote the measure on $P_G(L)$ and its quotient $P_G(L)/Aut(G)$.

Notice that the volumes are finite, because the pullback of ω_K extends continuously to $\overline{\mathfrak{M}}_{g,n}$. We moreover remark that, from Definition 3.4 and Theorem 3.3, the volumes are polynomials in the variables L_i , with coefficients given by ψ -classes intersections.

Corollary 3.7. [28] The volume $V_{g,n}^K(L)$ is a polynomial in L_1^2, \ldots, L_n^2 and satisfies

$$V_{g,n}^{K}(L) = \sum_{\substack{d_{1},\dots,d_{n} \ge 0\\d_{1}+\dots+d_{n}=d_{g,n}}} \left(\int_{\overline{\mathfrak{M}}_{g,n}} \prod_{i=1}^{n} \psi_{i}^{d_{i}} \right) \prod_{i=1}^{n} \frac{L_{i}^{2d_{i}}}{2^{d_{i}}d_{i}!}.$$
(3.6)

It is useful to record the expression of the Kontsevich measure in terms of edge lengths.

Lemma 3.8. If G is a trivalent ribbon graph of type (g, n), let $(l_e)_{e \in E_G}$ be the edge lengths. We have the equality of measures in $\mathbb{R}^{E_G}_+$

$$d\mu_{K} \cdot \prod_{i=1}^{n} dL_{i} = 2^{2g-2+n} \prod_{e \in E_{G}} dl_{e}.$$
(3.7)

Proof. From [28, Appendix C] or [11], we get

$$\frac{1}{d_{g,n}!} \left(\sum_{i=1}^n L_i^2 \Psi_i\right)^{d_{g,n}} \prod_{i=1}^n dL_i = 2^{5g-5+2n} \prod_{e \in E_G} dl_e.$$

Dividing on both sides by $2^{d_{g,n}}$ yields the result.

3.2 A combinatorial analogue of Wolpert's formula

The purpose of this section is to show that combinatorial Fenchel–Nielsen coordinates are Darboux for ω_K .

Theorem 3.9. Let Σ be a bordered surface of type (g, n) and fix any combinatorial Fenchel–Nielsen coordinates (ℓ_i, τ_i) for $\mathfrak{T}_{\Sigma}^{comb}(L)$. Denote by $\iota_L \colon \mathfrak{Z}_{\Sigma,G}(L) \hookrightarrow \mathfrak{T}_{\Sigma}^{comb}(L)$ the inclusion of a cell $\mathfrak{Z}_{\Sigma,G}(L)$. We have

$$\omega_{\rm K} = \iota_{\rm L}^* \left(\sum_{i=1}^{3g-3+n} d\ell_i \wedge d\tau_i \right). \tag{3.8}$$

The advantage of this formula is that, while the left-hand side is clearly pure mapping class group invariant and it does not depend on the pants decomposition, the right-hand side has a simple expression in terms of the global coordinates and does not rely on the cells decomposition.

3.2.1 Symplectic properties of the twist

The main technical ingredient for the proof is Proposition 3.11: the vector field associated to the twist along a simple closed curve is the Hamiltonian vector field of the length function of the curve. To prove such a result, we need to understand how small changes in the twist parameter affect the metric on the embedded ribbon graph.

More precisely, fix $\mathbb{G} \in \mathfrak{T}_{\Sigma}^{comb}$ in an open cell, and γ an essential, simple closed curve. Notice that, from Proposition 2.34, we can cut \mathbb{G} along γ and glue it back after twisting by a small amount $\tau \in \mathbb{R}$. In particular, since we are in an open cell, it makes sense to talk about the vector field ϑ_{τ} generated by infinitesimal changes in the twist.

To get an expression for ∂_{τ} , we observe that each time γ passes along an edge, a twist by small τ has the effect of either adding τ to the edge length, subtracting τ to the edge length, or leaving the edge length invariant (see the proof of Proposition 2.34 and Figure 18). This depends on the direction taken by γ at two consecutive vertices. Then one simply sums the changes in the length of the edges visited by γ . In the notation of Figure 23 (edges may appear twice along γ), the vector field describing the twisting along γ is given by

$$\partial_{\tau} = \sum_{i=1}^{F} \left(\frac{\partial}{\partial \ell_{p_{i}}^{[b_{i}]}} - \frac{\partial}{\partial \ell_{q_{i}}^{[b_{i}]}} \right) = \sum_{i=1}^{F} \left(\frac{\partial}{\partial \ell_{r_{i}}^{[c_{i}]}} - \frac{\partial}{\partial \ell_{s_{i}}^{[c_{i}]}} \right), \tag{3.9}$$

or in a more symmetric expression,

$$\partial_{\tau} = \frac{1}{2} \sum_{i=1}^{F} \left(\frac{\partial}{\partial \ell_{p_{i}}^{[b_{i}]}} - \frac{\partial}{\partial \ell_{q_{i}}^{[b_{i}]}} + \frac{\partial}{\partial \ell_{r_{i}}^{[c_{i}]}} - \frac{\partial}{\partial \ell_{s_{i}}^{[c_{i}]}} \right).$$
(3.10)

To compute the contraction $\iota_{\partial_{\tau}}\omega_{K}$, we need the following technical lemma. Note that an edge $e_{p}^{[i]}$ is either adjacent to two different faces *i.e.* $e_{p}^{[i]} = e_{\tau}^{[j]}$ for $i \neq j$, or adjacent to the same face on both sides *i.e.* $e_{p}^{[i]} = e_{\tau}^{[i]}$ for $p \neq r$.

Lemma 3.10. In the interior of a top-dimensional cell, we have with the above notations

$$\iota_{\partial_{p}^{[i]}+\partial_{r}^{[j]}}\omega_{K} = \sum_{k=p+1}^{N_{i}} d\ell_{k}^{[i]} - \sum_{k=1}^{p-1} d\ell_{k}^{[i]} + \sum_{u=r+1}^{N_{j}} d\ell_{u}^{[j]} - \sum_{u=1}^{r-1} d\ell_{u}^{[j]}.$$
(3.11)



Figure 23: A schematic picture of γ used to calculate the vector field associated to the twist. The labels can be redundant if γ visits an edge multiple times.

Proof. We recall that $\omega_{K} = \frac{1}{2} \sum_{i=1}^{n} L_{i}^{2} \Psi_{i}$ and Ψ_{i} defined in (3.1) only involves edges around the i-th face. Consider first the case $e_{p}^{[i]} = e_{r}^{[j]}$ for $i \neq j$. The interior product only receives contributions from Ψ_{i} and Ψ_{j} . The interior product with Ψ_{i} gives

$$\frac{2}{L_i^2} \left(\sum_{k=p+1}^{N_i} d\ell_k^{[i]} - \sum_{k=1}^{p-1} d\ell_k^{[i]} \right),$$

whereas the insertion into Ψ_j gives

$$\frac{2}{L_j^2} \left(\sum_{u=r+1}^{N_j} d\ell_u^{[i]} - \sum_{u=1}^{r-1} d\ell_u^{[j]} \right).$$

Therefore we obtain Equation (3.11). In the second case, we have $e_p^{[i]} = e_r^{[i]}$ for $p \neq r$, and the interior product only receives a contribution from Ψ_i . Assuming p < r, the interior product with Ψ_i is

$$\begin{split} \frac{2}{L_{i}^{2}} \left(\sum_{k=p+1}^{r-1} d\ell_{k}^{[i]} + 2 \sum_{k=r+1}^{N_{i}} d\ell_{k}^{[i]} - 2 \sum_{k=1}^{p-1} d\ell_{k}^{[i]} - \sum_{k=p+1}^{r-1} d\ell_{k}^{[i]} \right) \\ &= \frac{2}{L_{i}^{2}} \left(\sum_{k=p+1}^{N_{i}} d\ell_{k}^{[i]} - \sum_{k=1}^{p-1} d\ell_{k}^{[i]} + \sum_{u=r+1}^{N_{i}} d\ell_{u}^{[i]} - \sum_{u=1}^{r-1} d\ell_{u}^{[i]} \right). \end{split}$$

Therefore, once again, we obtain Equation (3.11). The case p > r is similar.

We are ready now to state the main property of the twist vector field: it is the Hamiltonian vector field associated to the combinatorial length function $\ell: \mathbb{G} \mapsto \ell_{\mathbb{G}}(\gamma)$. This is analogous to the situation in the hyperbolic case explored by Wolpert [44, Theorem 1.3] and generalises a result previously proved locally in [5, Lemma 3.2] and only for very special curves cutting out small pairs of pants.

Proposition 3.11. On the top-dimensional cells of $T_{\Sigma}^{\text{comb}}(L)$, we have

$$d\ell = \iota_{\partial_{\tau}} \omega_{K}. \tag{3.12}$$

Proof. Fix a top-dimensional cell and suppose that γ is given by the schematic of Figure 23. Then we have

$$d\ell = \sum_{i=1}^{r} \left(\frac{1}{2} d\ell_{p_{i}}^{[b_{i}]} + \frac{1}{2} d\ell_{q_{i}}^{[b_{i}]} + \sum_{p_{i} \prec k \prec q_{i}} d\ell_{k}^{[b_{i}]} + \frac{1}{2} d\ell_{r_{i}}^{[c_{i}]} + \frac{1}{2} d\ell_{s_{i}}^{[c_{i}]} + \sum_{r_{i} \prec u \prec s_{i}} d\ell_{u}^{[c_{i}]} \right),$$

where the symbol $\sum_{\mu \prec \lambda \prec \nu}$ indicates the sum over all edges of a certain face indexed by λ , that are between the edges indexed by μ and ν , following the orientation of the face and excluding the extremes μ and ν . Notice that the orientation of the face is opposite to the orientation of the corresponding boundary. On the other hand, we can reduce the computation of $\iota_{\partial_{\tau}}\omega_{K}$ to the insertion of the coordinate vector fields appearing in Equation (3.10).

Let us explain why this calculation leads to a well defined answer. Notice first that Definition 3.2 of Ψ_i makes perfect sense on the whole $\mathcal{T}_{\Sigma}^{comb}$, where the perimeter is not fixed, but the definition depends on the choice of a first edge in the i-th face. However, its pullback to $\mathcal{T}_{\Sigma}^{comb}(L)$ is independent of such a choice. Secondly, observe that the vector fields ∂_e are defined on $\mathcal{T}_{\Sigma}^{comb}$ but do not have a meaning on $\mathcal{T}_{\Sigma}^{comb}(L)$, as they do not preserve the boundary lengths L. However, particular linear combinations of them, such as ∂_{τ} , do. Therefore, it is legitimate to compute each contribution $\iota_{\partial_e} \Psi$ separately on $\mathcal{T}_{\Sigma}^{comb}(L)$.

This said, a repeated use of Lemma 3.10 results in the following computation.

$$\begin{split} \iota_{\vartheta_{\tau}} \omega_{K} &= \frac{1}{2} \sum_{i=1}^{F} \left[\left(\iota_{\vartheta_{p_{i}^{[b_{i}]} + \vartheta_{r_{i}^{[c_{i}]}}}^{[c_{i}]}} \right) \omega_{K} - \left(\iota_{\vartheta_{q_{i}^{[b_{i}]} + \vartheta_{s_{i}^{[c_{i}]}}}^{[c_{i}]}} \right) \omega_{K} \right] \\ &= \frac{1}{2} \sum_{i=1}^{F} \left[\left(\sum_{k=p_{i}+1}^{N_{b_{i}}} d\ell_{k}^{[b_{i}]} - \sum_{k=1}^{p_{i}-1} d\ell_{k}^{[b_{i}]} + \sum_{u=r_{i}+1}^{N_{c_{i}}} d\ell_{u}^{[c_{i}]} - \sum_{u=1}^{r_{i}-1} d\ell_{u}^{[c_{i}]} \right) \right. \\ &- \left(\sum_{m=q_{i}+1}^{N_{b_{i}}} d\ell_{m}^{[b_{i}]} - \sum_{m=1}^{q_{i}-1} d\ell_{m}^{[b_{i}]} + \sum_{\nu=s_{i}+1}^{N_{c_{i}}} d\ell_{\nu}^{[c_{i}]} - \sum_{\nu=1}^{s_{i}-1} d\ell_{\nu}^{[c_{i}]} \right) \right] \\ &= \sum_{i=1}^{F} \left(\frac{1}{2} d\ell_{p_{i}}^{[b_{i}]} + \frac{1}{2} d\ell_{q_{i}}^{[b_{i}]} + \sum_{p_{i}\prec k\prec q_{i}} d\ell_{k}^{[b_{i}]} + \frac{1}{2} d\ell_{r_{i}}^{[c_{i}]} + \frac{1}{2} d\ell_{s_{i}}^{[c_{i}]} + \sum_{r_{i}\prec u\prec s_{i}} d\ell_{u}^{[c_{i}]} \right). \end{split}$$

This indeed coincides with $d\ell$.

3.2.2 Proof of the combinatorial Wolpert formula

For a fixed oriented surface Σ , denote by $\overline{\Sigma}$ the surface with opposite orientation. The following lemma is based on the work of Wolpert [45, Lemma 1.1].

Lemma 3.12. Let Σ be a bordered surface, $\rho: \Sigma \to \overline{\Sigma}$ be an isotopy class of orientation-reversing diffeomorphism that restricts to the identity on the boundary. Fix $\gamma \in S_{\Sigma}^{\circ}$. Then ρ induces a homeomorphism $\mathbb{T}_{\Sigma}^{\text{comb}} \to \mathbb{T}_{\overline{\Sigma}}^{\text{comb}}$ and

• $\rho^* d\ell(\gamma) = d\ell(\rho \circ \gamma)$,

- $\rho^* \omega_K = -\omega_{K'}$
- *if* ρ *fixes* γ *, then* $\rho^* d\tau(\gamma) = -d\tau(\gamma) + \frac{n}{2}d\ell(\gamma)$ *for* $n \in \mathbb{Z}$.

Here $d\tau$ is the differential form dual with respect to $\omega_{\rm K}$ to the vector field ∂_{τ} of Equation(3.10).

Proof. The aforementioned induced map $\mathfrak{T}_{\Sigma}^{comb} \to \mathfrak{T}_{\Sigma}^{comb}$ is the composition of ρ with the marking. It inverts the orientations of all curves, but it fixes the length functions, hence the first point. Further, the Ψ -classes are going to be calculated using the opposite orientation, which yields the sign for the second point. The last point follows from the fact that the elements of $\operatorname{Stab}(\gamma)$ are generated by (half-) Dehn twists along curves that do not intersect γ . Then, as ρ reverses the orientation of the surface, $d\tau(\gamma)$ acquires a sign from the orientation reversal and an ambiguity of $\frac{1}{2}\mathbb{Z}d\ell(\gamma)$ from a potential (half-) Dehn twists along γ .

We are now ready to give a proof of the combinatorial Wolpert formula.

Proof of Theorem 3.9. Fix a seamed pants decomposition. We know that (ℓ_i, τ_i) give global coordinates on $\mathcal{T}_{\Sigma}^{comb}(L)$. Therefore, on the top-dimensional cells

$$\omega_K = \sum_{i < j} a_{ij} \, d\ell_i \wedge d\ell_j + \sum_{i < j} b_{ij} \, d\tau_i \wedge d\tau_j + \sum_{i,j} c_{ij} \, d\ell_i \wedge d\tau_j$$

for some functions a_{ij} , b_{ij} and c_{ij} . Notice that from Proposition 3.11 we have

$$\iota_{\mathfrak{d}_{\tau_i}}\omega_K = \sum_{i < j} b_{ij} \, d\tau_j - \sum_{j < i} b_{ji} \, d\tau_j + \sum_j c_{ji} \, d\ell_j = d\ell_i,$$

and hence $b_{ij} = 0$, $c_{ij} = \delta_{ij}$. Finally, if ρ is the isotopy class of an orientation-reversing diffeomorphism fixing γ_i , we have

$$\rho_*\Big(\frac{\partial}{\partial \ell_i}\Big) = \frac{\partial}{\partial \ell_i} + \frac{n_i}{2}\frac{\partial}{\partial \tau_i}.$$

Therefore from Lemma 3.12, for i < j, we find

$$\begin{split} \mathfrak{a}_{ij} = & \omega_{K} \left(\frac{\partial}{\partial \ell_{i}}, \frac{\partial}{\partial \ell_{j}} \right) = \omega_{K} \left(\frac{\partial}{\partial \ell_{i}} + \frac{n_{i}}{2} \frac{\partial}{\partial \tau_{i}}, \frac{\partial}{\partial \ell_{j}} + \frac{n_{j}}{2} \frac{\partial}{\partial \tau_{j}} \right) = \omega_{K} \left(\rho_{*} \frac{\partial}{\partial \ell_{i}}, \rho_{*} \frac{\partial}{\partial \ell_{j}} \right) = \rho^{*} \omega_{K} \left(\frac{\partial}{\partial \ell_{i}}, \frac{\partial}{\partial \ell_{j}} \right) \\ = & - \omega_{K} \left(\frac{\partial}{\partial \ell_{i}}, \frac{\partial}{\partial \ell_{j}} \right). \end{split}$$

and thus $a_{ij} = 0$. This proves the result on the top-dimensional cells.

To extend it to cells $\iota_L : \mathfrak{Z}_{\Sigma,G}(L) \hookrightarrow \mathfrak{T}_{\Sigma}^{comb}(L)$ of positive codimension, we can consider it at the boundary of a top-dimensional cell. Then ι_L^* simply sets $d\ell_e = 0$ for each edge *e* of zero length on the boundary of the top-dimensional cell. This has exactly the same affect as excluding such edges from the sum in Equation (3.1) of Ψ_i , which coincides with the definition of Ψ_i on cells of positive codimension.

3.3 Integration over the combinatorial moduli space

In this section we establish an integration result, analogous to [32, Theorem 7.1] for $(\mathcal{M}_{g,n}(L), \omega_{WP})$, exploiting the symplectic structure of $(\mathcal{M}_{g,n}^{comb}(L), \omega_K)$. What allows us to establish such a result is the combinatorial Wolpert formula (3.8). This improves the results of [5, Theorem 1.1], which in fact can be extended from their original use to the integration of functions with support restricted⁴ to "small pairs of pants".

Let us introduce some notation. Consider a bordered surface Σ of type (g, n) and let γ be a primitive multicurve with ordered components $(\gamma_j)_{j=1}^k$. We denote by Γ the orbit $Mod_{\Sigma}^{\partial} \cdot \gamma$ (although it is not important for

⁴This restriction on the support is related to the one appearing, *e.g.*, in (4.14).

what follows, such an object can be seen as a stable graph with ordered edges, see Section 4.5). Furthermore, consider an assignment

$$\Sigma' \longmapsto \Xi_{\Sigma'} \in \operatorname{Mes}(\mathfrak{T}_{\Sigma'}^{\operatorname{comb}}, \mu_K)$$

of a measurable function on the combinatorial Teichmüller space to each bordered surface Σ' diffeomorphic to the cut surface Σ_{γ} . We assume that for any diffeomorphism $\phi \colon \Sigma' \to \Sigma''$ which preserves the labelling of the boundary components, we have $\phi_* \Xi_{\Sigma'} = \Xi_{\Sigma''}$ where ϕ_* is the map induced between the combinatorial Teichmüller spaces. In particular, $\Xi_{\Sigma'}$ is invariant under the action of $\operatorname{Mod}_{\Sigma'}^{\partial}$ and descends to a function Ξ_{Γ} on the moduli space

$$\mathfrak{M}_{\Gamma}^{comb} = \prod_{\nu \in \pi_0(\Sigma_{\gamma})} \mathfrak{M}_{g(\nu), \mathfrak{n}(\nu)}^{comb},$$

which depends on Γ only. We have $\pi_0(\partial \Sigma') = \pi_0(\gamma) \sqcup \pi_0(\gamma) \sqcup \pi_0(\partial \Sigma)$, so it makes sense to consider⁵ $\mathcal{M}_{\Gamma}^{\text{comb}}(\ell, \ell, L)$. We further assume that Ξ_{Γ} is integrable with respect to the Kontsevich measure on $\mathcal{M}_{\Gamma}^{\text{comb}}(\ell, \ell, L)$ for almost every $(L, \ell) \in \mathbb{R}^n_+ \times \mathbb{R}^k_+$, and we denote

$$V\Xi_{\Gamma}(\ell,\ell,L) = \int_{\mathcal{M}_{\Gamma}^{comb}(\ell,\ell,L)} \Xi_{\Gamma} \, d\mu_{K}$$

Finally, consider a measurable function $f: \mathbb{R}^n_+ \times \mathbb{R}^k_+ \to \mathbb{R}$ and define a new function $\Xi^{f,\Gamma}_{\Sigma}$ on $\mathcal{T}^{comb}_{\Sigma}$ by setting

$$\Xi_{\Sigma}^{f,\Gamma}(\mathbb{G}) = \sum_{\alpha \in \Gamma} f(\vec{\ell}_{\mathbb{G}}(\partial \Sigma), \vec{\ell}_{\mathbb{G}}(\alpha)) \Xi_{\Sigma_{\alpha}}(\mathbb{G}|_{\Sigma_{\alpha}}),$$
(3.13)

where $\vec{\ell}_{\mathbb{G}}(\partial \Sigma) = (\ell_{\mathbb{G}}(\partial_i \Sigma))_{i=1}^n$ and $\vec{\ell}_{\mathbb{G}}(\alpha) = (\ell_{\mathbb{G}}(\alpha_j))_{j=1}^k$. When the series (3.13) is absolutely convergent, it defines a Mod_{Σ}^{0} -invariant function, which descends to a function $\Xi_{g,n}^{f,\Gamma}$ on the moduli space $\mathcal{M}_{g,n}^{comb}$.

Proposition 3.13. Assume that the series (3.13) is absolutely convergent, and that for almost every $L \in \mathbb{R}^{n}_{+}$ its limit is integrable with respect to μ_{K} on $\mathcal{M}_{\Sigma}^{comb}(L)$. Assume as well that for almost every $(L, \ell) \in \mathbb{R}^{n}_{+} \times \mathbb{R}^{k}_{+}$ the function Ξ_{Γ} is integrable on $\mathcal{M}_{\Gamma}^{comb}(\ell, \ell, L)$ with respect to the Kontsevich measure. Then

$$\int_{\mathcal{M}_{g,n}^{\text{comb}}(L)} \Xi_{g,n}^{f,\Gamma} d\mu_{K} = \int_{\mathbb{R}^{k}_{+}} f(L,\ell) \, V\Xi_{\Gamma}(\ell,\ell,L) \prod_{j=1}^{k} \ell_{j} \, d\ell_{j}.$$
(3.14)

Proof. We adapt Mirzakhani's proof of [32, Theorem 7.1], which concerned the hyperbolic setting with $\Xi_{\Sigma'} = 1$, functions $f(L, \ell) = F(\ell_1 + \cdots + \ell_k)$ and γ a primitive multicurve with unordered components. Because the components are ordered, our formula does not contain automorphism factors. The main difference is instead that, in the combinatorial setting, we have to remove the zero measure set of ill-defined twists.

Consider the space

$$\mathfrak{M}^{comb,\Gamma}_{g,n}(L) = \mathfrak{T}^{comb}_{\Sigma}(L) \Big/ \bigcap_{j=1}^{k} Stab(\gamma_{j}),$$

where $\operatorname{Stab}(\gamma_j)$ is the stabiliser of γ_j in $\operatorname{Mod}_{\Sigma}^{\partial}$. We denote $\Pi^{\Gamma} \colon \mathfrak{M}_{g,n}^{\operatorname{comb},\Gamma}(L) \to \mathfrak{M}_{g,n}^{\operatorname{comb}}(L)$ the natural projection. Notice that

$$\mathfrak{M}_{g,\mathfrak{n}}^{\operatorname{comb},\Gamma}(L) \cong \left\{ \left. \left(\mathbf{G}, \alpha \right) \; \middle| \; \mathbf{G} \in \mathfrak{M}_{g,\mathfrak{n}}^{\operatorname{comb}}(L), \quad \alpha \in \Gamma \right. \right\}.$$

Since the symplectic structure on $\mathcal{T}_{\Sigma}^{comb}$ is invariant under the action of the pure mapping class group, it induces a symplectic structure on $\mathcal{M}_{g,n}^{comb,\Gamma}(L)$, which is the same as the pullback $(\Pi^{\Gamma})^* \omega_K$. We denote the associated measure by μ_K^{Γ} .

⁵As explained in § 2.3.1, the boundary components of the cut surface are labeled in a specific way. Thus, with the symbol $\mathcal{M}_{\Gamma}^{\text{comb}}(\ell,\ell,L)$, we mean the product of moduli spaces with fixed boundary lengths from $\ell_1, \ldots, \ell_k, L_1, \ldots, L_n$ ordered in such a way that they match the labeled boundary components of the cut surface.

Consider now the map $\mathcal{T}_{\Sigma}^{comb}(L) \to \mathbb{R}_{+}^{k}$ given by the tuple of combinatorial lengths of the components of γ . It descends to a map $\mathcal{L}^{\Gamma} \colon \mathcal{M}_{g,n}^{comb,\Gamma}(L) \to \mathbb{R}_{+}^{k}$. We denote by $\mathcal{M}_{g,n}^{comb,\Gamma}(L)[\ell] = (\mathcal{L}^{\Gamma})^{-1}(\ell)$ the level sets for $\ell \in \mathbb{R}_{+}^{k}$. We have a map

$$\Pi: \mathcal{M}_{q,n}^{\operatorname{comb},\Gamma}(\mathsf{L})[\ell] \longrightarrow \mathcal{M}_{\Gamma}^{\operatorname{comb}}(\ell,\ell,\mathsf{L})$$
(3.15)

defined in the natural way: given an element $(\mathbf{G}, \alpha) \in \mathcal{M}_{g,n}^{comb,\Gamma}(L)[\ell]$, we take a lift $\mathbb{G} \in \mathfrak{T}_{\Sigma}^{comb}(L)$ of \mathbf{G} , we restrict \mathbb{G} to the cut surface Σ_{α} as explained in Section 2.4 and we project the restriction to the moduli space $\mathcal{M}_{\Sigma_{\alpha}}^{comb}(\ell, \ell, L) \cong \mathcal{M}_{\Gamma}^{comb}(\ell, \ell, L)$. The result does not depend on the choice of the lift \mathbb{G} since we are projecting to the combinatorial moduli space of Σ_{α} after restriction.

Notice that the spaces on both sides of (3.15) have a natural measure: $\mathfrak{M}_{g,n}^{comb,\Gamma}(L)[\ell]$ is equipped with the disintegration of μ_{K}^{Γ} along \mathcal{L}^{Γ} , and $\mathfrak{M}_{\Gamma}^{comb}(\ell, \ell, L)$ with its Kontsevich measure. By construction of (3.13) and the property of disintegration,

$$\begin{split} \int_{\mathcal{M}_{g,n}^{\text{comb}}(L)} \Xi_{g,n}^{f,\Gamma} d\mu_{K} &= \int_{\mathcal{M}_{g,n}^{\text{comb},\Gamma}(L)} (f \circ \mathcal{L}^{\Gamma}) \cdot (\Xi_{\Gamma} \circ \Pi) \, d\mu_{K}^{\Gamma} \\ &= \int_{\mathbb{R}_{+}^{k}} f(L,\ell) \left(\int_{\mathcal{M}_{g,n}^{\text{comb},\Gamma}(L)[\ell]} (\Xi_{\Gamma} \circ \Pi) \, d\mu_{K}^{\Gamma} \right) \prod_{j=1}^{k} d\ell_{j}. \end{split}$$
(3.16)

We can complete γ into a seamed pants decomposition and use the Fenchel–Nielsen coordinates to describe the space $\mathcal{M}_{g,n}^{comb,\Gamma}(L)[\ell]$. The combinatorial Wolpert formula (3.8) implies that the measure μ_K^{Γ} has a product structure with respect to the fibration Π . Besides, the fibre of $\mathbf{G}' \in \mathcal{M}_{\Gamma}^{comb}(\ell, \ell, L)$ is identified with an open subset of full measure in $\prod_{i=1}^{k} \mathbb{R}/2^{-t_i} \ell_j \mathbb{Z}$, where

$$t_j = \begin{cases} 1 & \text{if } \gamma_j \text{ separates off a torus with one boundary,} \\ 0 & \text{otherwise.} \end{cases}$$

This follows from the description of the image of the Fenchel–Nielsen coordinates in Theorem 2.41. The factor of 2^{-t_j} in the case when γ_j separates off a torus with one boundary is due to the fact that any element in $\mathcal{M}_{1,1}^{\text{comb}}(L)$ comes with an elliptic involution, so $\text{Stab}(\gamma_i)$ contains the half-twist along γ_j and the fundamental region of the twist coordinate in the combinatorial moduli space becomes $[0, \ell_j/2]$ minus a measure-zero set. So, for any open set $U \subseteq \mathcal{M}_{\Gamma}^{\text{comb}}(\ell, \ell, L)$, we have

$$\prod_{j=1}^{k} 2^{-t_j} \int_{\Pi^{-1}(U)} (\Xi_{\Gamma} \circ \Pi) \, d\mu_{K}^{\Gamma} = \prod_{j=1}^{k} 2^{-t_j} \ell_j \int_{U} \Xi_{\Gamma} \, d\mu_{K}$$

whenever the functions we wish to integrate are integrable. We noticed that the integrals over $\mathcal{M}_{1,1}^{\text{comb}}$ have an extra factor of $\frac{1}{2}$ due to the presence of the elliptic involution, while such a factor is not present on $\mathcal{M}_{g,n}^{\text{comb},\Gamma}(L)[\ell]$. We must therefore include an extra factor 2^{-t_j} in the right-hand side, and this cancels the factor of 2^{-t_j} coming from the half-twist. By a partition of unity argument, we obtain

$$\int_{\mathcal{M}_{g,\pi}^{comb,\Gamma}(L)[\ell]} (\Xi_{\Gamma} \circ \Pi) \, d\mu_{K}^{\Gamma} = V \Xi_{\Gamma}(\ell,\ell,L) \prod_{j=1}^{k} \ell_{j},$$

which we insert in (3.16) to complete the proof.

Remark 3.14. In Mirzakhani's work [32] there is an unnatural convention for the integral over $\mathcal{M}_{1,1}(L)$ which does not include the factor of $\frac{1}{2}$ coming from the elliptic involution. For this reason, she finds an extra factor $\prod_{i=1}^{k} 2^{-t_i}$.

Remark 3.15. As briefly mentioned before, an equivalent way to state Proposition 3.13, which used in Section 6.3, relies on the notion of stable graphs (introduced in Section 4.5). In this language, the set of connected components of Σ_{γ} is the set V_{Γ} of vertices of the stable graph with ordered edges $\Gamma = \text{Mod}_{\Sigma}^{\partial} \cdot \gamma$, the set of components of γ descends to the set E_{Γ} of edges of Γ , and the set of boundary component of Σ is the set Λ_{Γ} of leaves of Γ . With this notation, Equation (3.14) becomes

$$\int_{\mathcal{M}_{g,n}^{comb}(L)} \Xi_{g,n}^{f,\Gamma} d\mu_{K} = \int_{\mathbb{R}_{+}^{E_{\Gamma}}} f((L_{\lambda})_{\lambda \in \Lambda_{\Gamma}}, (\ell_{e})_{e \in E_{\Gamma}}) \prod_{\nu \in V_{\Gamma}} V \Xi_{g(\nu),n(\nu)} ((\ell_{e})_{e \in E_{\nu}}, (L_{\lambda})_{\lambda \in \Lambda_{\nu}}) \prod_{e \in E_{\Gamma}} \ell_{e} d\ell_{e}.$$
(3.17)

If in the above arguments we do not suppose that the components of the primitive multicurve are ordered, then we have to include the automorphism factor $\frac{1}{\#Aut(\Gamma)}$ multiplying the right-hand side of Equation (3.17). Moreover, if the multicurve is not primitive but comes with a weight $a \in \mathbb{Z}_{+}^{E_{\Gamma}}$, we have to consider the automorphism factor $\frac{1}{\#Aut(\Gamma,a)}$ of the weighted multicurve.

Remark 3.16. We can get rid of the integrability assumption for $\Xi_{g,n}^{f,\Gamma}$, if we suppose that each term in the series (3.13) is non-negative. In this case, the disintegration property still holds in (3.16), and the result of Proposition 3.13 becomes an equality between (possibly infinite) integrals.

4 Functions from geometric recursion

In this section, we review the construction of mapping class group invariant functions on the combinatorial and hyperbolic Teichmüller spaces via the general framework of geometric recursion (GR) proposed in [2]. We moreover show that the integral on the moduli space of these functions against the Kontsevich measure satisfy topological recursion (TR). We shall establish the combinatorial version of the Mirzakhani–McShane identity: the constant function 1 on the combinatorial Teichmüller space is an output of the geometric recursion for some particular input data. As a corollary, we can present purely geometric proofs of Witten conjecture/Kontsevich theorem [43, 28] and of Norbury's lattice point count by topological recursion [36], which are completely parallel to Mirzakhani's proof of the topological recursion for the Weil–Petersson volumes in [32].

In Section 5 we study a particular flow related to the boundary lengths in the classical sense, which interpolates between the hyperbolic and the combinatorial geometries, allowing us to transfer several of the results discussed from T_{Σ} to T_{Σ}^{comb} directly.

4.1 Definition of the geometric recursion

We begin by reviewing in a simplified form the framework of geometric recursion introduced in [2].

Let us introduce the category \mathcal{B}_1 whose objects are bordered surfaces (see Definition 2.8) and morphisms are isotopy classes of diffeomorphisms relatively to the boundary which preserve $\partial_1 \Sigma$ but are allowed to permute the labelling of the other boundary components.

Let us discuss the excision of pairs of pants. We have introduced in Definition 2.22 various sets of homotopy classes of embeddings $\varphi: P \hookrightarrow \Sigma$. From now on, we only consider the ones corresponding to $m_0 = 1$, *i.e.* $\varphi(\partial_1 P) = \partial_1 \Sigma$, and drop the index m_0 from the notations. We therefore have a set partition

$$\mathfrak{P}_{\Sigma} = \left(\bigsqcup_{\mathfrak{m}=2}^{\mathfrak{n}} \mathfrak{P}_{\Sigma}^{\mathfrak{m}}\right) \sqcup \mathfrak{P}_{\Sigma}^{\varnothing}.$$

If $[P] \in \mathcal{P}_{\Sigma}$, we denote by $\Sigma - P$ the bordered surface $\overline{\Sigma \setminus \varphi(P)}$ obtained from some representative $\varphi: P \hookrightarrow \Sigma$ of [P]. To define the labelling of the boundary components of $\Sigma - P$, we say that the (labelled) boundary components of P that appear in $\Sigma - P$ are put first, followed by the (labelled) boundary components of Σ that appear in $\Sigma - P$. Choosing another representative of [P] gives a canonically isomorphic object.

The geometric recursion (abbreviated GR) starts with a functor E from \mathcal{B}_1 to a category of topological vector spaces and aims at constructing E-valued functorial assignments $\Sigma \mapsto \Omega_{\Sigma} \in E(\Sigma)$. For this purpose, E must come with extra functorial data that satisfy a number of axioms, subsumed in the notion of *target theory*. Instead of repeating the fully general definition of target theories and associated geometric recursion [2], we shall describe in concrete terms the geometric recursion for the two examples of E used in the present work, namely the spaces of measurable functions on the Teichmüller space of Σ , either seen as a space of hyperbolic or combinatorial structures on Σ . The first one was developed in [2, Sections 7–10] while the second one is new. In general, if \mathcal{X} is a topological space, we denote by $Mes(\mathcal{X})$ the space of realvalued measurable functions on \mathcal{X} . Most results of this section still hold true after replacing everywhere "measurable" by "continuous".

4.1.1 The hyperbolic case

For any bordered surface, we let $E(\Sigma) = Mes(T_{\Sigma})$. GR initial data consists of a quadruple (A, B, C, D) where A, B, C are measurable functions on $T_P \cong \mathbb{R}^3_+$ such that $X(L_1, L_2, L_3) = X(L_1, L_3, L_2)$ for $X \in \{A, C\}$ and D_T is

a measurable function on \mathfrak{T}_T . We require $\mathsf{T} \mapsto \mathsf{D}_T$ to be functorial. We let

$$\Omega_{\varnothing} = 1, \qquad \Omega_{\mathsf{P}}(\sigma) = \mathsf{A}(\vec{\ell}_{\sigma}(\partial \mathsf{P})), \qquad \Omega_{\mathsf{T}} = \mathsf{D}_{\mathsf{T}},$$

where $\vec{\ell}_{\sigma}(\partial P)$ is the ordered triple of hyperbolic lengths of the boundary components of P. For disconnected surfaces we set

$$\Omega_{\Sigma_1 \sqcup \cdots \sqcup \Sigma_k}(\sigma_1, \ldots, \sigma_k) = \prod_{i=1}^{\kappa} \Omega_{\Sigma_i}(\sigma_i)$$

It remains to define Ω_{Σ} for connected surfaces with Euler characteristic $\chi_{\Sigma} \leq -2$. This is done inductively χ_{Σ} by geometric recursion:

$$\Omega_{\Sigma}(\sigma) = \sum_{m=2}^{n} \sum_{[P] \in \mathcal{P}_{\Sigma}^{m}} B(\vec{\ell}_{\sigma}(\partial P)) \Omega_{\Sigma-P}(\sigma|_{\Sigma-P}) + \frac{1}{2} \sum_{[P] \in \mathcal{P}_{\Sigma}^{\varnothing}} C(\vec{\ell}_{\sigma}(\partial P)) \Omega_{\Sigma-P}(\sigma|_{\Sigma-P}).$$
(4.1)

Here, we choose as representative of [P] the embedding as a pair of pants with geodesic boundaries. In this case, the restriction of σ to $\Sigma - P$ makes it a hyperbolic surface with geodesic boundaries. The sum (4.1) has countably many terms and therefore its convergence should be discussed.

Definition 4.1. Let $\mathfrak{T}_{\Sigma}^{(\varepsilon)} \subset \mathfrak{T}_{\Sigma}$ be the ε -thick part of the Teichmüller space, *i.e.* the set of $\sigma \in \mathfrak{T}_{\Sigma}$ such that $\ell_{\sigma}(\gamma) \ge \varepsilon$ for any $\gamma \in S_{\Sigma}$. We say that initial data (A, B, C, D) are *admissible* if for any $\varepsilon > 0$ there exists $t \ge 0$ such that, for all $s \ge 0$, there exists $M_{\varepsilon,s} > 0$ for which

$$\sup_{L_{1},L_{2},L_{3} \geq \epsilon} \frac{|A(L_{1},L_{2},L_{3})|}{\left((1+L_{1})(1+L_{2})(1+L_{3})\right)^{t}} \leq M_{\epsilon,0} \qquad \sup_{\sigma \in \mathcal{T}_{T}^{(\epsilon)}} \frac{|D_{T}(\sigma)|}{\left(1+\ell_{\sigma}(\partial T)\right)^{t}} \leq M_{\epsilon,0} \\
\sup_{L_{1},L_{2},\ell \geq \epsilon} \frac{|B(L_{1},L_{2},\ell)|\left(1+[\ell-L_{1}-L_{2}]_{+}\right)^{s}}{\left((1+L_{1})(1+L_{2})\right)^{t}} \leq M_{\epsilon,s} \qquad \sup_{L_{1},\ell,\ell' \geq \epsilon} \frac{|C(L_{1},\ell,\ell')|\left(1+[\ell+\ell'-L_{1}]_{+}\right)^{s}}{(1+L_{1})^{t}} \leq M_{\epsilon,s}$$
(4.2)

Theorem 4.2. [2, Corollary 8.3] If (A, B, C, D) are admissible initial data, then for any bordered surface Σ

- the series (4.1) converges absolutely and uniformly on any compact of T_{Σ} ,
- $\Sigma \mapsto \Omega_{\Sigma} \in \text{Mes}(\mathfrak{T}_{\Sigma})$ is a well-defined functorial assignment (in particular, Ω_{Σ} is $\text{Mod}_{\Sigma}^{\mathfrak{d}}$ -invariant),
- there exists $u \ge 0$ depending only on the topological type of Σ , such that for any $\varepsilon > 0$ we have

$$\sup_{\sigma \in \mathcal{T}_{\Sigma}^{(\epsilon)}} |\Omega_{\Sigma}(\sigma)| \leqslant \mathsf{K}_{\epsilon} \prod_{\mathbf{b} \in \pi_{0}(\mathfrak{d}\Sigma)} (1 + \ell_{\sigma}(\mathbf{b}))^{\mathsf{u}}$$
(4.3)

for some constant K_{ε} depending only on ε and the topological type of Σ .

4.1.2 The combinatorial case

For any bordered surface, we let $E(\Sigma) = Mes(\mathfrak{T}_{\Sigma}^{comb})$. GR initial data consist of a quadruple (A, B, C, D) such that A, B, C are measurable functions on $\mathfrak{T}_{P}^{comb} \cong \mathbb{R}^{3}_{+}$ such that $X(L_{1}, L_{2}, L_{3}) = X(L_{1}, L_{3}, L_{2})$ for $X \in \{A, C\}$ and D is a measurable function on \mathfrak{T}_{T}^{comb} . We require $T \mapsto D_{T}$ to be functorial. We let

$$\Xi_{\varnothing} = 1, \qquad \Xi_{\mathsf{P}}(\mathbb{G}) = \mathsf{A}(\overline{\ell}_{\mathbb{G}}(\partial \mathsf{P})), \qquad \Xi_{\mathsf{T}} = \mathsf{D}_{\mathsf{T}},$$

For disconnected surfaces we set

$$\Xi_{\Sigma_1\sqcup\cdots\sqcup\Sigma_k}(\mathbb{G}_1,\ldots,\mathbb{G}_k)=\prod_{\mathfrak{i}=1}^k\Xi_{\Sigma_\mathfrak{i}}(\mathbb{G}_\mathfrak{i}).$$

For connected surfaces Σ with $\chi_{\Sigma} \leq -2$, we define inductively

$$\Xi_{\Sigma}(\mathbb{G}) = \sum_{m=2}^{n} \sum_{[P] \in \mathcal{P}_{\Sigma}^{m}} B(\vec{\ell}_{\mathbb{G}}(\partial P)) \Xi_{\Sigma-P}(\mathbb{G}|_{\Sigma-P}) + \frac{1}{2} \sum_{[P] \in \mathcal{P}_{\Sigma}^{\varnothing}} C(\vec{\ell}_{\mathbb{G}}(\partial P)) \Xi_{\Sigma-P}(\mathbb{G}|_{\Sigma-P}),$$
(4.4)

where $\mathbb{G}|_{\Sigma-P}$ have been defined by the cutting procedure in Section 2.4. We remark that, given a homotopy class of embedded pair of pants, we can restrict the combinatorial structure \mathbb{G} to a representative of P, so that the triple of lengths $\tilde{\ell}_{\mathbb{G}}(\partial P)$ makes sense.

Definition 4.3. Let $\mathfrak{T}_{\Sigma}^{\operatorname{comb},(\varepsilon)} \subset \mathfrak{T}_{\Sigma}^{\operatorname{comb}}$ be the ε -thick part of the combinatorial Teichmüller space, *i.e.* the set of $\mathbb{G} \in \mathfrak{T}_{\Sigma}^{\operatorname{comb}}$ such that $\ell_{\mathbb{G}}(\gamma) \ge \varepsilon$ for any $\gamma \in S_{\Sigma}$. We say that initial data (A, B, C, D) are *admissible* if they satisfy the same conditions appearing in Definition 4.1, except we use $\mathfrak{T}_{T}^{\operatorname{comb},(\varepsilon)}$ in the condition for D_{T} .

Theorem 4.4. If (A, B, C, D) are admissible initial data, then for any bordered surface Σ

- the series (4.4) converges absolutely and uniformly on any compact of T_{Σ}^{comb} ,
- $\Sigma \mapsto \Xi_{\Sigma} \in \text{Mes}(\mathcal{T}_{\Sigma}^{\text{comb}})$ is a well-defined functorial assignment (in particular, Ξ_{Σ} is $\text{Mod}_{\Sigma}^{\partial}$ -invariant),
- there exists $u \ge 0$ depending only on the topological type of Σ , such that for any $\varepsilon > 0$ we have

$$\sup_{\mathbb{G}\in\mathcal{T}_{\Sigma}^{\mathrm{comb},(\varepsilon)}} |\Xi_{\Sigma}(\mathbb{G})| \leqslant \mathsf{K}_{\varepsilon} \prod_{\mathfrak{b}\in\pi_{0}(\mathfrak{d}\Sigma)} (1+\ell_{\mathbb{G}}(\mathfrak{b}))^{\mathfrak{u}}$$
(4.5)

for some constant K_{ε} depending only on ε and the topological type of Σ .

Although the spaces $Mes(\mathcal{T}_{\Sigma})$ and $Mes(\mathcal{T}_{\Sigma}^{comb})$ can be identified via the spine homeomorphism of Theorem 2.14, the way we measure lengths and we restrict to P and $\Sigma - P$ is different and as a result the hyperbolic/combinatorial structure in (4.1) and (4.4) are completely different. So, for identical initial data, the hyperbolic and the combinatorial GR do not produce the same functions (even after identification of their domains). The relation between the hyperbolic and combinatorial GR is elucidated in Section 5.

Proof. The result follows from the general theory of [2] after proving that $Mes(\mathfrak{T}_{\Sigma}^{comb})$ is a target theory. We present a self-contained proof which does not rely on these general notions, by specialising the strategy of [2] to this simpler setting.

It is enough to prove the result for connected surfaces. By definition of the initial data, the result holds for connected surfaces of Euler characteristic -1. Let us assume it holds for all surfaces of Euler characteristic strictly greater than χ . Let Σ be a bordered surface of type (g, n) with $2 - 2g - n = \chi$, take $\varepsilon > 0$ and fix $\mathbb{G} \in \mathcal{T}_{\Sigma}^{comb,(\varepsilon)}$. For any $[P] \in \mathcal{P}_{\Sigma}$, we have as well $\mathbb{G}|_{\Sigma-P} \in \mathcal{T}_{\Sigma-P}^{comb,(\varepsilon)}$. Therefore, by induction hypothesis, there exist $u \ge 0$ and $K_{\varepsilon} > 0$ which we can choose to depend only on ε and the topological type of Σ , such that

$$|\Xi_{\Sigma-P}(\mathbb{G}_{\Sigma-P})| \leqslant K_{\varepsilon} \prod_{b \in \pi_0(\partial(\Sigma-P))} (1 + \ell_{\mathbb{G}}(b))^{\mathfrak{u}}.$$

We now study the absolute convergence of the GR series (4.4). We use the notation X_P for the function B when $[P] \in \mathcal{P}_{\Sigma}^m$ and for the function $\frac{1}{2}C$ when $[P] \in \mathcal{P}_{\Sigma}^{\emptyset}$. We first isolate the sum over G-small pairs of pants. Using the fact that there are at most 2(6g - 6 + 3n) G-small pairs of pants (Remark 2.27), together with the

admissibility conditions on X_P (Definition 4.3) and the inequality $(1 + L_1 + L_2)^t \leq (1 + L_1)^t (1 + L_2)^t$ for any $L_1, L_2 > 0$, we get

$$\sum_{\substack{[P]\in\mathcal{P}_{\Sigma}\\G-small}} \left| X_{P}(\vec{\ell}_{G}(\partial P)) \Xi_{\Sigma-P}(G|_{\Sigma-P}) \right| \leq 2(6g-6+3n) M_{\varepsilon,0} K_{\varepsilon} \prod_{i=1}^{n} \left(1 + \ell_{G}(\partial_{i}\Sigma) \right)^{\max\{u,t\}}.$$
(4.6)

We now turn to the contributions of the G-big pairs of pants in \mathcal{P}_{Σ}^{m} . We have for any s > 0

$$\begin{split} &\sum_{\substack{[\mathsf{P}]\in\mathfrak{P}_{\Sigma}^{\mathfrak{m}}\\ \mathbb{G}\text{-big}}} \left| \mathsf{B}(\vec{\ell}_{\mathbb{G}}(\partial\mathsf{P}))\Xi_{\Sigma-\mathsf{P}}(\mathbb{G}|_{\Sigma-\mathsf{P}}) \right| \\ &\leqslant \mathsf{M}_{\varepsilon,s}\mathsf{K}_{\varepsilon} \left(\prod_{i\neq 1,\mathfrak{m}} \left(1+\ell_{\mathbb{G}}(\partial_{i}\Sigma)\right)^{\mathfrak{u}} \right) \left(\sum_{L\in\ell_{\mathbb{G}}(\partial_{1}\Sigma)+\ell_{\mathbb{G}}(\partial_{\mathfrak{m}}\Sigma)+\mathbb{N}} \frac{(2+\mathsf{L})^{\mathsf{t}}\,\#\{\gamma\in\mathsf{S}_{\Sigma}^{\circ}\mid\mathsf{L}\leqslant\ell_{\mathbb{G}}(\gamma)<\mathsf{L}+1\}}{\left(1+\mathsf{L}-\ell_{\mathbb{G}}(\partial_{1}\Sigma)-\ell_{\mathbb{G}}(\partial_{\mathfrak{m}}\Sigma)\right)^{s}} \right) \\ &\leqslant \mathsf{M}_{\varepsilon,s}\mathsf{K}_{\varepsilon}\,\mathfrak{m}_{\varepsilon} \left(\prod_{i\neq 1,\mathfrak{m}} (1+\ell_{\mathbb{G}}(\partial_{i}\Sigma))^{\mathfrak{u}} \right) \left(\sum_{L\geqslant 1} \left(1+\mathsf{L}+\ell_{\mathbb{G}}(\partial_{1}\Sigma)+\ell_{\mathbb{G}}(\partial_{\mathfrak{m}}\Sigma)\right)^{\mathsf{t}+6g-6+2n}\mathsf{L}^{-s} \right). \end{split}$$

In the last line, we invoked the polynomial growth of the number of multicurves with respect to combinatorial length, justified later in Proposition 6.3. Specialising to s=(6g-6+2n+t)+2 makes the sum in brackets converging to a polynomial of degree t'=t+6g-6+2n in the variable $\ell_{\mathbb{G}}(\mathfrak{d}_{1}\Sigma)+\ell_{\mathbb{G}}(\mathfrak{d}_{m}\Sigma)$ and, together with (4.6), it implies the existence of a constant $K'_{\varepsilon}>0$ such that

$$\sum_{[P]\in \mathfrak{P}_{\Sigma}^{\mathfrak{m}}} \left|B(\vec{\ell}_{\mathbb{G}}(\partial P)) \Xi_{\Sigma-P}(\mathbb{G}|_{\Sigma-P})\right| \leqslant K_{\varepsilon}' \prod_{i=1}^{n} \bigl(1+\ell_{\mathbb{G}}(\partial_{i}\Sigma)\bigr)^{max\{u,t'\}}$$

A similar argument shows that

$$\sum_{[P]\in\mathfrak{P}_{\Sigma}^{\varnothing}} \left| C(\vec{\ell}_{\mathbb{G}}(\partial P)) \Xi_{\Sigma-P}(\mathbb{G}|_{\Sigma-P}) \right| \leqslant K_{\varepsilon}' \prod_{i=1}^{n} (1 + \ell_{\mathbb{G}}(\partial_{i}\Sigma))^{max\{u,t'\}}$$

for a perhaps larger constant K'_{e} . Consequently, the series

$$\sum_{[P]\in\mathfrak{P}_{\Sigma}}X_{P}(\vec{\ell}_{\mathbb{G}}(\partial P))\Xi_{\Sigma-P}(\mathbb{G}|_{\Sigma-P})$$

converges absolutely and uniformly on any compact of $\mathfrak{T}_{\Sigma}^{comb,(\varepsilon)}$, to a limit that we denote Ξ_{Σ} . Further, the bounds that we just proved imply that this limit satisfies

$$\forall \mathbb{G} \in \mathfrak{T}_{\Sigma}^{comb,(\,\varepsilon\,)}, \qquad |\Xi_{\Sigma}(\mathbb{G})| \leqslant K_{\varepsilon}'' \prod_{i=1}^n \bigl(1+\ell_{\mathbb{G}}(\mathfrak{d}_i\Sigma)\bigr)^{\mathfrak{u}'}$$

for some constant $K_e' > 0$ and $u' = \max\{u, t'\}$. The proof is then completed by induction.

4.2 (Discrete) integration and topological recursion

Since the functions produced by the hyperbolic or combinatorial GR (generically called GR amplitudes) are pure mapping class group invariant, they descend to functions on the corresponding moduli spaces. For a connected surface Σ of type (g, n) and hyperbolic (resp. combinatorial) GR amplitudes Ω_{Σ} (resp. Ξ_{Σ}), we denote by $\Omega_{g,n}$ (resp. $\Xi_{g,n}$) the functions induced on the associated moduli spaces. For the initial datum $T \mapsto D_T$, we denote by D the induced function on $\mathcal{M}_{1,1}$ or $\mathcal{M}_{1,1}^{comb}$. In the first part of this section we discuss how to integrate hyperbolic and combinatorial GR amplitudes respectively against the Weil–Petersson measure and the Kontsevich measure. In the second part, which belongs exclusively to the combinatorial setting, we discuss how to define discrete integration on the combinatorial moduli space via sums over integral metric ribbon graphs.

4.2.1 Integration and topological recursion

For $L \in \mathbb{R}^{n}_{+}$, $\mathcal{M}_{g,n}(L)$ is naturally endowed with the Weil–Petersson measure μ_{WP} , while $\mathcal{M}_{g,n}^{comb}(L)$ is equipped with the Kontsevich measure μ_{K} . It is important to note that these measures are compatible with cutting, as expressed in the hyperbolic setting by [32, Theorem 7.1] and in the combinatorial setting by Proposition 3.13. Thus, integration of functions obtained by GR against these measures over the moduli space with fixed boundary lengths produce functions on \mathbb{R}^{n}_{+} that also satisfy a recursion on the Euler characteristic, called topological recursion (TR). In order to guarantee integrability we are going to introduce stronger assumptions on the initial data.

Definition 4.5. In the hyperbolic setting, the initial data (A, B, C, D) are called *strongly admissible* if there exists $\eta \in [0, 2)$ and $t \ge 0$ such that, for any $s \ge 0$, there exists $M_s > 0$ such that for any $L_1, L_2, L_3, \ell, \ell' > 0$

$$\begin{aligned} |A(L_{1}, L_{2}, L_{3})| &\leq M_{0} \left((1 + L_{1})(1 + L_{2})(1 + L_{3}) \right)^{t}, \\ |B(L_{1}, L_{2}, \ell)| &\leq \frac{M_{s} \left((1 + L_{1})(1 + L_{2}) \right)^{t}}{\ell^{\eta} \left(1 + [\ell - (L_{1} + L_{2})]_{+} \right)^{s}}, \\ |C(L_{1}, \ell, \ell')| &\leq \frac{M_{s} \left(1 + L_{1} \right)^{t}}{(\ell \ell')^{\eta} \left(1 + [\ell + \ell' - L_{1}]_{+} \right)^{s}}, \end{aligned}$$

$$(4.7)$$

and D is integrable on $\mathcal{M}_{1,1}(L_1)$ and satisfies

$$\left| \int_{\mathcal{M}_{1,1}(L_1)} \mathsf{D} \, d\mu_{\mathrm{WP}} \right| \leqslant \mathsf{M}_0 \, (1+L_1)^{\mathsf{t}}. \tag{4.8}$$

We define strong admissibility of the initial data (A, B, C, D) in the combinatorial setting by the same conditions, except for (4.8) which gets substituted by

$$\left| \int_{\mathcal{M}_{1,1}^{\text{comb}}(L_1)} D \, d\mu_K \right| \leqslant M_0 \, (1+L_1)^t. \tag{4.9}$$

Whenever these integrals are defined, we use the generic notations

$$V\Omega_{g,n}(L) = \int_{\mathcal{M}_{g,n}(L)} \Omega_{g,n} \, d\mu_{WP}, \qquad \quad V\Xi_{g,n}(L) = \int_{\mathcal{M}_{g,n}^{comb}(L)} \Xi_{g,n} \, d\mu_{K}.$$
(4.10)

Theorem 4.6. [2, Theorem 8.8] Let (A, B, C, D) be strongly admissible initial data for the hyperbolic GR, and Ω_{Σ} be the resulting functions. Then, Ω_{Σ} is integrable against μ_{WP} on $\mathcal{M}_{\Sigma}(L)$ for any $L \in \mathbb{R}^{n}_{+}$, and the integrals satisfy the following recursion on 2g - 2 + n > 1.

$$\begin{split} &V\Omega_{g,n}(L_{1},...,L_{n}) \\ &= \sum_{m=2}^{n} \int_{\mathbb{R}_{+}} B(L_{1},L_{m},\ell) \, V\Omega_{g,n-1}(\ell,L_{2},...,\widehat{L_{m}},...,L_{n}) \, \ell d\ell \\ &+ \frac{1}{2} \int_{\mathbb{R}_{+}^{2}} C(L_{1},\ell,\ell') \bigg(V\Omega_{g-1,n+1}(\ell,\ell',L_{2},...,L_{n}) + \sum_{\substack{h+h'=g\\J \sqcup J' = \{L_{2},...,L_{n}\}} V\Omega_{h,1+\#J}(\ell,J) \, V\Omega_{h',1+\#J'}(\ell',J') \bigg) \ell \ell' d\ell \, d\ell' \end{split}$$

$$(4.11)$$

with the conventions $V\Omega_{0,1} = 0$ and $V\Omega_{0,2} = 0$, and the base cases

$$V\Omega_{0,3}(L_1, L_2, L_3) = A(L_1, L_2, L_3)$$
 and $V\Omega_{1,1}(L_1) = \int_{\mathcal{M}_{1,1}(L_1)}^{\infty} D \, d\mu_{WP}.$

Theorem 4.7. Let (A, B, C, D) be strongly admissible initial data for the combinatorial GR, and Ξ_{Σ} be the resulting functions. Then, Ξ_{Σ} is integrable against μ_{K} on $\mathcal{M}_{\Sigma}^{comb}(L)$ for any $L \in \mathbb{R}^{n}_{+}$, and the integrals $V\Xi_{g,n}(L)$ satisfy the same recursion as in (4.11), i.e.

$$\begin{aligned} & \nabla \Xi_{g,n}(L_{1},...,L_{n}) \\ &= \sum_{m=2}^{n} \int_{\mathbb{R}_{+}} B(L_{1},L_{m},\ell) \, \nabla \Xi_{g,n-1}(\ell,L_{2},...,\widehat{L_{m}},...,L_{n}) \ell d\ell \\ &\quad + \frac{1}{2} \int_{\mathbb{R}_{+}^{2}} C(L_{1},\ell,\ell') \bigg(\nabla \Xi_{g-1,n+1}(\ell,\ell',L_{2},...,L_{n}) + \sum_{\substack{h+h'=g\\ J \sqcup J' = \{L_{2},...,L_{n}\}}} \nabla \Xi_{h,1+\#J}(\ell,J) \, \nabla \Xi_{h',1+\#J'}(\ell',J') \bigg) \ell \ell' d\ell \, d\ell' \end{aligned}$$
(4.12)

with the conventions $V\Xi_{0,1} = 0$ and $V\Xi_{0,2} = 0$, and the base cases

$$V\Xi_{0,3}(L_1, L_2, L_3) = A(L_1, L_2, L_3) \quad \text{and} \quad V\Xi_{1,1}(L_1) = \int_{\mathcal{M}_{1,1}^{comb}(L_1)} D \, d\mu_K$$

Proof. We first note that the initial data $V\Xi_{0,3}$ is well-defined as A is, and that $V\Xi_{1,1}$ is well-defined by strong admissibility.

Now, for a connected surface Σ of type (g, n) with 2g - 2 + n > 1, apply the integration over $\mathcal{M}_{g,n}^{comb}(L)$ with respect to μ_{K} to both sides of the combinatorial GR (4.4). We analyse the integration of the sum over $[P] \in \mathcal{P}_{\Sigma}^{m}$. Let Γ be the Mod_{Σ}^{∂} -orbit of a simple closed curve bounding a pair of pants together with $\partial_{1}\Sigma$ and $\partial_{m}\Sigma$. We have

$$\sum_{P]\in\mathcal{P}_{\Sigma}^{\mathfrak{m}}} B(\vec{\ell}_{\mathbb{G}}(\partial P)) \Xi_{\Sigma-P}(\mathbb{G}|_{\Sigma-P}) = \sum_{\alpha\in\Gamma} B(L_{1}, L_{\mathfrak{m}}, \ell_{\mathbb{G}}(\alpha)) \Xi_{\Sigma-P_{\alpha}}(\mathbb{G}|_{\Sigma-P_{\alpha}}),$$

where P_{α} is the pair of pants bounded by $\partial_1 \Sigma$, $\partial_m \Sigma$ and α . Now applying Proposition 3.13 to $f(L, \ell) = B(L_1, L_m, x)$ and the assignment $\Sigma_{\alpha} \mapsto \Xi_{\Sigma - P_{\alpha}}$, we find

$$\int_{\mathcal{M}_{g,n}^{\text{comb}}(L)} \sum_{\alpha \in \Gamma} B(L_1, L_m, \ell_*(\alpha)) \Xi_{g,n-1} d\mu_K$$

=
$$\int_{\mathbb{R}_+} B(L_1, L_m, \ell) \left(\int_{\mathcal{M}_{g,n-1}^{\text{comb}}(\ell, L_2, \dots, \widehat{L_m}, \dots, L_n)} \Xi_{g,n-1} d\mu_K \right) \ell d\ell$$

=
$$\int_{\mathbb{R}_+} B(L_1, L_m, \ell) V \Xi_{g,n-1}(\ell, L_2, \dots, \widehat{L_m}, \dots, L_n) \ell d\ell.$$

The treatment of the C summands is similar, the main difference being that the excised pair of pants has two simple closed curves in Σ° , whose lengths are part of the combinatorial Fenchel–Nielsen coordinates over which we need to integrate.

4.2.2 Discrete integration and topological recursion

As the combinatorial moduli spaces have an integral structure, we can also study discrete integration of GR amplitudes. Again, the geometric recursion here is the key property that guarantees a topological recursion for the discrete integrals, *i.e.* the sum over lattice points.

Definition 4.8. For 2g - 2 + n > 0, let $\mathcal{M}_{g,n}^{comb,\mathbb{Z}} \subset \mathcal{M}_{g,n}^{comb}$ be the set of classes of metric ribbon graphs whose edge lengths are positive integers, and $\mathcal{M}_{g,n}^{comb,\mathbb{Z}}(L)$ the one with fixed perimeter $L \in \mathbb{Z}_{+}^{n}$. We denote likewise $\mathcal{T}_{\Sigma}^{comb,\mathbb{Z}}$ and $\mathcal{T}_{\Sigma}^{comb}(L)$ their preimages in the combinatorial Teichmüller space of a bordered surface Σ of type (g, n).

Since for any $\mathbf{G} \in \mathcal{M}_{g,n}^{\text{comb},\mathbb{Z}}(L)$, we have $\sum_{i=1}^{n} L_i = \sum_{e \in E_G} 2\ell_e$, the set $\mathcal{M}_{g,n}^{\text{comb},\mathbb{Z}}(L)$ is finite for any fixed L, and it is empty if $\sum_{i=1}^{n} L_i$ is odd. For instance,

$$\mathcal{T}_{P}^{comb,\mathbb{Z}} \cong \mathcal{M}_{0,3}^{comb,\mathbb{Z}} \cong \left\{ \left(L_{1},L_{2},L_{3} \right) \in \mathbb{Z}_{+}^{3} \mid L_{1} + L_{2} + L_{3} \text{ is even } \right\}.$$

To handle this situation, for X: $\mathbb{R}^n_+ \to \mathbb{R}$ we introduce the notation

$$X_{\mathbb{Z}}(L) = \begin{cases} X(L) & \text{if } L \in \mathbb{Z}^n_+ \text{ and } \sum_{i=1}^n L_i \text{ is even,} \\ 0 & \text{otherwise.} \end{cases}$$

Further, for any function $\Xi_{g,n}$ on $\mathcal{M}_{g,n}^{comb}$, set

$$\mathsf{N}\Xi_{g,\mathfrak{n}}(\mathsf{L}) = \sum_{\mathbf{G}\in\mathcal{M}_{g,\mathfrak{n}}^{\mathrm{omb},\mathbb{Z}}(\mathsf{L})} \frac{\Xi_{g,\mathfrak{n}}(\mathbf{G})}{\#\mathrm{Aut}(\mathbf{G})} = \sum_{\mathsf{G}\in\mathcal{R}_{g,\mathfrak{n}}} \frac{1}{\#\mathrm{Aut}(\mathsf{G})} \sum_{\mathsf{x}\in\mathsf{P}_{\mathsf{G}}(\mathsf{L})\cap\mathbb{Z}_{+}^{\mathfrak{n}}} \Xi_{g,\mathfrak{n}}(\mathsf{x}),\tag{4.13}$$

where we recall that $P_G(L)$ is the set of metrics on G with perimeter L.

Theorem 4.9. Let A, B, C be three functions on \mathbb{R}^3_+ such that A and C are symmetric under exchange of their last two variables, and satisfying

$$\ell > L_1 + L_2 \implies B(L_1, L_2, \ell) = 0, \quad and \quad \ell + \ell' > L_1 \implies C(L_1, \ell, \ell') = 0.$$
(4.14)

Let D be Mod_T -invariant function on \mathfrak{T}_T^{comb} . Denote by Ξ_{Σ} the corresponding combinatorial GR amplitudes. We have the following recursion on 2g - 2 + n > 1.

$$\begin{split} &\mathsf{N}\Xi_{g,n}(\mathsf{L}_{1},\ldots,\mathsf{L}_{n}) \\ &= \sum_{m=2}^{n} \sum_{\ell \geqslant 1} \ell \, \mathsf{B}_{\mathbb{Z}}(\mathsf{L}_{1},\mathsf{L}_{m},\ell) \, \mathsf{N}\Xi_{g,n-1}(\ell,\mathsf{L}_{2},\ldots,\widehat{\mathsf{L}_{m}},\ldots,\mathsf{L}_{n}) \\ &+ \frac{1}{2} \sum_{\ell,\ell' \geqslant 1} \ell \ell' \, \mathsf{C}_{\mathbb{Z}}(\mathsf{L}_{1},\ell,\ell') \bigg(\mathsf{N}\Xi_{g-1,n+1}(\ell,\ell',\mathsf{L}_{2},\ldots,\mathsf{L}_{n}) + \sum_{\substack{\mathsf{h}+\mathsf{h}'=g\\ J \sqcup J' = \{\mathsf{L}_{2},\ldots,\mathsf{L}_{n}\}} \mathsf{N}\Xi_{\mathsf{h},1+\#J}(\ell,J) \, \mathsf{N}\Xi_{\mathsf{h}',1+\#J'}(\ell',J') \bigg) \end{split}$$
(4.15)

with conventions $N\Xi_{0,1} = 0$ and $N\Xi_{0,2} = 0$, and base cases

$$N\Xi_{0,3}(L_1, L_2, L_3) = A_{\mathbb{Z}}(L_1, L_2, L_3) \quad and \quad N\Xi_{1,1}(L_1) = \sum_{\mathbf{G} \in \mathcal{M}_{1,1}^{comb,\mathbb{Z}}(L_1)} \frac{D(\mathbf{G})}{\#Aut(\mathbf{G})}.$$

Since cutting combinatorial structures preserve their integrability, in order to obtain the GR amplitudes in Theorem 4.9, it is sufficient to have (A, B, C) defined on $\mathcal{T}_{P}^{\text{comb},\mathbb{Z}}$ and D defined on $\mathcal{T}_{T}^{\text{comb},\mathbb{Z}}$. When the vanishing condition (4.14) hold, the GR sums have only finitely many non-zero terms: they are always well-defined, without the need of admissibility conditions for (A, B, C, D).

We can recover the continuous integration of Theorem 4.7 as a limit where we rescale the mesh of the lattice down to 0. If k > 0, we let $\mathcal{M}_{g,n}^{\text{comb},\mathbb{Z}/k}$ be the set of metric ribbon graphs in $\mathcal{M}_{g,n}^{\text{comb}}$ whose edge lengths all become integral after dilation by k.

Proposition 4.10. Assume that (A, B, C, D) are continuous functions on their respective combinatorial Teichmüller spaces, satisfying the vanishing conditions (4.14), and such that for any fixed $L_1, L_2 > 0$, the functions $\ell \mapsto B(L_1, L_2, \ell)$ and $(\ell, \ell') \mapsto C(L_1, L_2, \ell')$ are bounded, and the function D_T is bounded on $\mathcal{T}_T^{comb}(L_1)$. Then (A, B, C, D) is strongly admissible. If d be a positive integer and $L \in (\mathbb{Z}_+/d)^n$, we have for 2g - 2 + n > 0

$$\lim_{\substack{k \to \infty \\ k \in d\mathbb{Z}_+}} \left(\frac{1}{k^{6g-6+2n}} \sum_{\mathbf{G} \in \mathcal{M}_{g,n}^{\text{comb},\mathbb{Z}/k}(\mathbf{L})} \frac{\Xi_{g,n}(\mathbf{G})}{\# \text{Aut}(\mathbf{G})} \right) = \begin{cases} 2^{-2g+3-n} \, V\Xi_{g,n}(\mathbf{L}) & \text{if } \sum_{i=1}^n d \cdot L_i \text{ is even,} \\ 0 & \text{otherwise.} \end{cases}$$
(4.16)

The continuous integration also appears in the asymptotics of the lattice count for large boundary lengths.

Corollary 4.11. Assume that (A, B, C, D) are continuous functions on their respective combinatorial Teichmüller spaces and satisfy the vanishing conditions (4.14). We further assume the existence of b, $c \in \mathbb{R}$ for which, for any M > 0 and $L_1, L_2, \ell, \ell' \in (0, M]$

- $k^{-2b-c} A(kL_1, kL_2, kL_3)$ converges uniformly to $\hat{A}(L_1, L_2, L_3)$ as $k \to \infty$,
- $k^{-b} B(kL_1, kL_2, k\ell)$ converges uniformly to $\hat{B}(L_1, L_2, \ell)$ as $k \to \infty$,
- $k^{-c} C(kL_1, k\ell, k\ell')$ converges uniformly to $\hat{C}(L_1, \ell, \ell')$ as $k \to \infty$,
- $k^{-b} D_T(k\mathbb{G})$ converges uniformly to $\hat{D}_T(\mathbb{G})$ as $k \to \infty$, for $\mathbb{G} \in \bigcup_{\ell \in (0,M]} T_T^{comb}(\ell)$, where $k\mathbb{G}$ is obtained from \mathbb{G} by dilation of the metric by a factor k,
- $(\hat{A}, \hat{B}, \hat{C}, \hat{D})$ satisfy the assumptions of Proposition 4.10.

Denoting by $\hat{\Xi}_{\Sigma}$ the GR amplitudes associated to the initial data $(\hat{A}, \hat{B}, \hat{C}, \hat{D})$, we have for any $d \in \mathbb{Z}_+$, $L \in (\mathbb{Z}_+/d)^n$ and 2g - 2 + n > 0

$$\lim_{\substack{k \to \infty \\ k \in d\mathbb{Z}_+}} \frac{\mathsf{N}\Xi_{g,n}(k\mathsf{L})}{\mathsf{k}^{(g-1)(b+c)+nb} \cdot \mathsf{k}^{6g-6+2n}} = \begin{cases} 2^{-2g+3-n} \, \mathsf{V}\hat{\Xi}_{g,n}(\mathsf{L}) & \text{if } \sum_{i=1}^n \mathsf{d} \cdot \mathsf{L}_i \text{ even,} \\ 0 & \text{otherwise.} \end{cases}$$
(4.17)

Proof of Theorem 4.9. When $\sum_{i=1}^{n} L_i$ is odd, both sides vanish, so we only need to prove the result when $\sum_{i=1}^{n} L_i$ is even, which we now assume. The vanishing conditions for B and C imply that for each $L \in \mathbb{Z}_+^n$, the sums in the right-hand side have finitely many terms.

Let us substitute the GR sum for Ξ_{Σ} in (4.13). For $m \in \{2, ..., n\}$, we examine in detail how to handle the term

$$\Xi_{\Sigma}^{B,m}(\mathbb{G}) = \sum_{[P] \in \mathcal{P}_{\Sigma}^{m}} B(\vec{\ell}_{\mathbb{G}}(\partial P)) \Xi_{g,n-1}(\mathbb{G}|_{\Sigma-P}).$$

Let Γ be the Mod_{Σ}^{∂} -orbit of a simple closed curve that bounds some $[P] \in \mathcal{P}_{\Sigma}^{m}$. Adapting the notation of Section 3.3, we denote by $\mathcal{M}_{g,n}^{comb,\Gamma,\mathbb{Z}}(L)$ the integral points in $\mathcal{M}_{g,n}^{comb,\Gamma}(L)$. Then

$$\begin{split} \mathsf{N}\Xi^{\mathsf{B},\mathsf{m}}_{g,\mathsf{n}}(\mathsf{L}) &= \sum_{\mathbf{G}\in\mathcal{M}_{g,\mathsf{n}}^{\mathrm{comb},\mathbb{Z}}(\mathsf{L})} \frac{\Xi^{\mathsf{B},\mathsf{m}}_{g,\mathsf{n}}(\mathbf{G})}{\#\mathrm{Aut}(\mathbf{G})} \\ &= \sum_{(\mathbf{G},\alpha)\in\mathcal{M}_{g,\mathsf{n}}^{\mathrm{comb},\Gamma,\mathbb{Z}}(\mathsf{L})} \frac{1}{\#\mathrm{Aut}(\mathbf{G})} \, \mathsf{B}(\mathsf{L}_{1},\mathsf{L}_{\mathsf{m}},\mathcal{L}^{\Gamma}(\mathbf{G},\alpha)) \, \Xi_{g,\mathsf{n}-1}(\Pi(\mathbf{G})) \\ &= \sum_{\ell \geqslant 1} \sum_{(\mathbf{G},\alpha)\in\mathcal{M}_{g,\mathsf{n}}^{\mathrm{comb},\Gamma,\mathbb{Z}}(\mathsf{L})[\ell]} \frac{1}{\#\mathrm{Aut}(\mathbf{G})} \, \mathsf{B}(\mathsf{L}_{1},\mathsf{L}_{\mathsf{m}},\ell) \, \Xi_{g,\mathsf{n}-1}(\Pi(\mathbf{G})), \end{split}$$

where we recall from Section 3.3 the map

$$\mathcal{L}^{\Gamma} \colon \mathcal{M}^{\operatorname{comb},\Gamma,\mathbb{Z}}_{g,n}(\mathsf{L}) \longrightarrow \mathbb{Z}_{+}$$

assigning to (\mathbf{G}, α) the combinatorial length with respect to \mathbf{G} of α . It has fibers $(\mathcal{L}^{\Gamma})^{-1}(\ell) = \mathcal{M}_{g,n}^{\text{comb}, \Gamma, \mathbb{Z}}(L)[\ell]$. Moreover, we have the projection map

$$\Pi: \mathcal{M}_{g,n}^{\operatorname{comb}, \Gamma, \mathbb{Z}}(L)[\ell] \longrightarrow \mathcal{M}_{g,n-1}^{\operatorname{comb}, \mathbb{Z}}(\ell, L_2, \dots, \widehat{L_m}, \dots, L_n).$$

We want to cluster this sum according to the fibres of the map Π . We first notice that, as $\mathcal{M}_{g,n}^{comb,\Gamma,\mathbb{Z}}(L)[\ell]$ is empty when ℓ and $L_1 + L_m$ have different parity, we can replace B by $B_{\mathbb{Z}}$, and now only consider ℓ that has same parity as $L_1 + L_m$. In this case, for any $\mathbf{G}' \in \mathcal{M}_{g,n-1}^{comb,\mathbb{Z}}(\ell, L_2, \dots, \widehat{L_m}, \dots, L_n)$ and any $\mathbf{G} \in \Pi^{-1}(\mathbf{G}')$, we remark that $\#Aut(\mathbf{G}') = \#Aut(\mathbf{G})$. Thus

$$\frac{1}{\#\operatorname{Aut}(\mathbf{G})} B_{\mathbb{Z}}(L_1, L_m, \ell) \Xi_{g,n-1}(\Pi(\mathbf{G}))$$

is constant on the fibres of Π . Due to the vanishing conditions (4.14), the points (\mathbf{G}, α) with non-trivial contribution have associated pair of pants $[\mathsf{P}_{\alpha}]$ that is small. Therefore from Corollary 2.33, $\Pi^{-1}(\mathbf{G}')$ is in bijection with the set of $[\tau] \in \mathbb{R}/\ell\mathbb{Z}$ such that the gluing of \mathbb{G}' to the combinatorial structure of the pair of pants with boundary lengths ($\mathsf{L}_1, \mathsf{L}_m, \ell$) after a twist τ produces a combinatorial structure with integral edge lengths. In order to satisfy the latter condition, the twists must belong to the set $(\tau_{\gamma}(\mathbb{G}) + \mathbb{Z})/\ell\mathbb{Z} \simeq \mathbb{Z}/\ell\mathbb{Z} \simeq \Pi^{-1}(\mathbf{G}')$, whose cardinality is ℓ . Therefore, our sum becomes

$$\Xi_{g,n}^{B}(L) = \sum_{\ell \ge 1} \sum_{\mathbf{G}' \in \mathcal{M}_{g,n-1}(\ell, L_{2},...,\widehat{L_{m}},...,L_{n})} \frac{1}{\# \operatorname{Aut}(\mathbf{G}')} B_{\mathbb{Z}}(L_{1}, L_{m}, \ell) \Xi_{g,n-1}(\mathbf{G}')$$

$$= \sum_{\ell \ge 1} \ell B_{\mathbb{Z}}(L_{1}, L_{m}, \ell) \operatorname{N}\Xi_{g,n-1}(\ell, L_{2}, ..., \widehat{L_{m}}, ..., L_{n}).$$
(4.18)

The treatment of the C summand is similar, except that we should be cautious about automorphism factors. For each of the finitely many Mod_{Σ}^{∂} -orbit Γ of a simple closed curve that bounds some $[P] \in \mathcal{P}_{\Sigma}^{\varnothing}$, and we observe that

$$#Aut(\Gamma) = \begin{cases} 2 & \text{if } \Sigma - P \text{ is connected,} \\ 1 & \text{otherwise.} \end{cases}$$

Since in the analogue of (4.18) we have for any $\mathbf{G} \in \Pi^{-1}(\mathbf{G}')$

$$#\operatorname{Aut}(\Gamma) \cdot #\operatorname{Aut}(\mathbf{G}) = #\operatorname{Aut}(\mathbf{G}'),$$

the automorphism factors are again naturally included in N Ξ_{Γ} , and we get the C-terms in (4.15) without extra automorphism factors (as the $\frac{1}{2}$ is already present in C).

Remark 4.12. The vanishing assumptions (4.14) are essential to allow the use of Corollary 2.33. If they did not hold, the fibres $\Pi^{-1}(\mathbf{G}')$ of the gluing fibration could, and do, meet integral non-admissible twists, hence their cardinality could be smaller than ℓ or $\ell\ell'$ and depend on \mathbf{G}' . It would then not be possible to derive a recursion for the weighted sum over lattice points. This problem did not arise for the integration against $\mu_{\rm K}$ as the set of non-admissible twists has zero measure with respect to $\mu_{\rm K}$.

Before turning to the proof of Proposition 4.10, we need two preliminary results.

Proof of Proposition 4.10. The case (g, n) = (0, 3) is obvious, as there is equality before taking the limit. This initial case is special with respect to the other topologies since the moduli space is reduced to a point: in the rest of the proof, we suppose $(g, n) \neq (0, 3)$. In general, when $\sum_{i=1}^{n} d \cdot L_i$ is not even, the set $\mathcal{M}_{g,n}^{\operatorname{comb},\mathbb{Z}/k}(L)$ with $k \in d\mathbb{Z}_+$ is empty, so the left-hand side of Equation (4.16) vanishes, which proves half of the result. Hereafter we assume that $k \in d\mathbb{Z}_+$, and fix $L \in (\mathbb{Z}_+/d)^n$ such that $\sum_{i=1}^{n} d \cdot L_i$ is even.

The thesis follows now from a general discussion. Consider a bounded function f defined on $\mathcal{M}_{g,n}^{\text{comb}}(L)$. The sum over rescaled lattice points is, by definition,

$$\sum_{\mathbf{G}\in \mathcal{M}_{g,n}^{comb,\mathbb{Z}/k}(L)}\frac{f(\mathbf{G})}{\#Aut(\mathbf{G})} = \sum_{G\in \mathcal{R}_{g,n}}\frac{1}{\#Aut(G)}\sum_{x\in \mathsf{P}_G(L)\cap \mathbb{Z}_+^n/k}f(x),$$

where we recall that $P_G(L) \subset \mathbb{R}^{E_G}_+$ is the set of metrics on G with perimeters L. We first estimate the sum over non-trivalent graphs by

$$\sum_{\substack{G \in \mathcal{R}_{g,n} \\ \text{non-trivalent}}} \frac{1}{\# Aut(G)} \sum_{x \in P_G(L) \cap \mathbb{Z}_+^n/k} f(x) \Bigg| \leqslant \Big(\sup_{\mathcal{M}_{g,n}^{comb}(L)} |f| \Big) \sum_{\substack{G \in \mathcal{R}_{g,n} \\ \text{non-trivalent}}} \frac{\# \Big(P_G(L) \cap \mathbb{Z}_+^n/k \Big)}{\# Aut(G)}.$$

By dimensional reasons, the right-hand side is $O(k^{6g-7+2n})$ as $k \to +\infty$. Hence

$$\lim_{\substack{k\to\infty\\k\in d\mathbb{Z}_+}} \left(\frac{1}{k^{6g-6+2n}}\sum_{\mathbf{G}\in\mathcal{M}_{g,n}^{comb,\mathbb{Z}/k}(L)}\frac{f(\mathbf{G})}{\#Aut(\mathbf{G})}\right) = \lim_{\substack{k\to\infty\\k\in d\mathbb{Z}_+}} \left(\frac{1}{k^{6g-6+2n}}\sum_{\substack{\mathbf{G}\in\mathcal{R}_{g,n}\\trivalent}}\frac{1}{\#Aut(\mathbf{G})}\sum_{x\in\mathsf{P}_{\mathsf{G}}(L)\cap\mathbb{Z}_{+}^{n}/k}f(x)\right).$$

By Lemma C.2 and by definition of the Riemann integral, we have

$$\lim_{\substack{k \to \infty \\ k \in d\mathbb{Z}_+}} \left(\frac{1}{k^{6g-6+2n}} \sum_{x \in \mathsf{P}_{\mathsf{G}}(\mathsf{L}) \cap \mathbb{Z}_+^n / k} f(x) \right) = \int_{\mathsf{P}_{\mathsf{G}}(\mathsf{L})} f \, d\ell_{n+1} \dots d\ell_{6g-6+3n}$$

provided f is continuous. Let G be a trivalent ribbon graph, and let $\iota_L : P_G(L) \hookrightarrow \mathbb{R}^{E_G}_+$ be the inclusion map. The cell $P_G(L)$ is naturally equipped with the measure μ_{Leb} defined as $\iota_L^*(\prod_{e \in E_G} d\ell_e)$. Using Lemma C.1, we see that $\prod_{j=n+1}^{6g-6+3n} d\ell_j = 2\mu_{Leb}$. Besides, we know from Lemma 3.8 that $\mu_{Leb} = 2^{2-2g-n}\mu_K$. As a consequence, we find

$$2^{3-2g-n} \int_{\mathfrak{M}_{g,n}^{comb}(L)} f \, d\mu_{K} = \lim_{\substack{k \to \infty \\ k \in d\mathbb{Z}_{+}}} \bigg(\frac{1}{k^{6g-6+2n}} \sum_{\substack{G \in \mathcal{R}_{g,n} \\ trivalent}} \frac{1}{\#Aut(G)} \sum_{x \in \mathsf{P}_{G}(L) \cap \mathbb{Z}_{+}^{n}/k} f(x) \bigg).$$

Proof of Corollary **4.11**. Let $\Xi_{\Sigma}^{(k)}$ be the GR amplitudes for the initial data

$$\begin{split} A^{(k)}(L_1, L_2, L_3) &= k^{-2b-c} A(kL_1, kL_2, kL_3), \\ B^{(k)}(L_1, L_2, L_3) &= k^{-b} B(kL_1, kL_2, k\ell), \\ C^{(k)}(L_1, L_2, L_3) &= k^{-c} C(kL_1, k\ell, k\ell'), \\ D^{(k)}_T(\mathbb{G}) &= k^{-b} D_T(k\mathbb{G}). \end{split}$$

Tracking the powers of k^{-1} in the GR sum (4.4), one can show by induction that

$$\Xi_{\Sigma}(k\mathbb{G}) = k^{(g-1)(b+c)+bn} \Xi_{\Sigma}^{(k)}(\mathbb{G})$$

and thus

$$\frac{N\Xi_{g,n}(kL_{1},\ldots,kL_{n})}{k^{6g-6+2n}\cdot k^{(g-1)(b+c)+bn}} = \frac{1}{k^{6g-6+2n}} \left(\sum_{\mathbf{G}\in\mathcal{M}_{g,n}^{\text{comb},\mathbb{Z}/k}(\mathbf{L})} \frac{\Xi_{\Sigma}^{(k)}(\mathbf{G})}{\#\text{Aut}(\mathbf{G})}\right).$$
(4.19)

The vanishing conditions (4.14) have two consequences for us.

- (i) The amplitude $\Xi_{\Sigma}^{(k)}(\mathbb{G})$ is given by a finite sum of products with 2g 2 + n factors that can be either $A^{(k)}, B^{(k)}, C^{(k)}, D^{(k)}$.
- (ii) If we fix $L \in \mathbb{R}^{n}_{+}$, for any $\mathbb{G} \in \mathbb{T}^{comb}_{\Sigma}(L)$, the factors of $A^{(k)}, B^{(k)}, C^{(k)}$ are evaluated on triples of lengths that are smaller than $M = \sum_{i=1}^{n} L_{i}$, and D is evaluated on elements of $\bigcup_{\ell \in (0,M]} \mathbb{T}^{comb}_{T}(\ell)$.

Hence $\Xi_{\Sigma}^{(k)}$ converges uniformly to $\hat{\Xi}_{\Sigma}$ on $\mathcal{T}_{\Sigma}^{comb}(L)$. We can then replace $\Xi_{\Sigma}^{(k)}$ with $\hat{\Xi}_{\Sigma}$ in (4.19) up to an error that tends to 0 when $k \to \infty$, and we conclude by using Proposition 4.10 for $\hat{\Xi}_{\Sigma}$.

4.3 Remark: inducing D from C

There is a natural way to complete (A, B, C) into an initial data (A, B, C, D), satisfying all the assumptions that we may desire to impose.

Lemma 4.13. If we are only given (A, B, C) satisfying the conditions in Definition 4.1, the series

$$D_{\mathsf{T}}(\sigma) = \sum_{\gamma \in S_{\mathsf{T}}^{\circ}} C(\ell_{\sigma}(\partial \mathsf{T}), \ell_{\sigma}(\gamma), \ell_{\sigma}(\gamma))$$
(4.20)

converges absolutely on any compact of T_T to a Mod_T-invariant function, and (A, B, C, D) are admissible initial data. Furthermore, if (A, B, C) satisfy the conditions in Definition 4.5, where in the bound for C one assumes $0 \le \eta < 1$, then (A, B, C, D) are strongly admissible and

$$\int_{\mathcal{M}_{1,1}(\mathsf{L})} \mathsf{D} \, \mathrm{d}\mu_{\mathsf{WP}} = \int_{\mathbb{R}_+} \mathsf{C}(\mathsf{L},\ell,\ell)\ell \mathrm{d}\ell. \tag{4.21}$$

The same lemma holds in the combinatorial setting – replacing hyperbolic with combinatorial lengths and μ_{WP} with μ_{K} . In the last statement, the stronger condition $\eta < 1$ for C (instead of $\eta < 2$) guarantees that $\ell \mapsto \ell C(L, \ell, \ell)$ is integrable near 0. We omit the proof of Lemma 4.13 and its combinatorial analog, as it is very similar to arguments already used in Theorems 4.2, 4.4, 4.6 and 4.7.

When we say that (A, B, C) are admissible initial data, we implicitly assume that they should be completed by the choice (4.20) of D_T. With this convention, notice that (strong) admissibility of (A, B, C) is the same condition in the hyperbolic and in the combinatorial setting.

Remark 4.14. When $C(L_1, \ell, \ell')$ vanishes for $L_1 < \ell + \ell'$, the sum (4.20) is finite. It is thus well-defined without admissibility conditions and satisfies

$$\sum_{\mathbf{G}\in\mathcal{M}_{1,1}^{\text{comb},\mathbb{Z}}(L)} \frac{\mathsf{D}(\mathbf{G})}{\#\text{Aut}(\mathbf{G})} = \sum_{\ell=1}^{\lfloor L/2 \rfloor} \ell \operatorname{C}(L,\ell,\ell)$$

This is in particular relevant for the discrete setting, *e.g.* for Theorem 4.9.

4.4 Mirzakhani–McShane identities

4.4.1 In the hyperbolic case

Mirzakhani proved in [32, Theorem 1.3] a generalisation for all bordered surfaces of the identity discovered by McShane [31] for the punctured torus, and used it to prove a topological recursion in the form of (4.11) for the Weil–Petersson volumes. It can be reformulated by saying that the constant function 1 can be obtained from geometric recursion. We state it below and we are going to use it later on.

Theorem 4.15. [32] Let $F(x) = 2 \ln(1 + e^{x/2})$. The following initial data

$$\begin{split} A^{M}(L_{1},L_{2},L_{3}) &= 1, \\ B^{M}(L_{1},L_{2},\ell) &= \frac{1}{2L_{1}} \big(F(L_{1}+L_{2}-\ell) + F(L_{1}-L_{2}-\ell) - F(-L_{1}+L_{2}-\ell) - F(-L_{1}-L_{2}-\ell) \big), \\ C^{M}(L_{1},\ell,\ell') &= \frac{1}{L_{1}} \big(F(L_{1}-\ell-\ell') - F(-L_{1}-\ell-\ell') \big), \end{split}$$
(4.22)

are admissible, and lead by geometric recursion to $\Omega_{\Sigma}^{M}(\sigma) = 1$ for any Σ and $\sigma \in \mathfrak{T}_{\Sigma}$. In other words, for any connected bordered surface Σ with $\chi_{\Sigma} < -1$ and any $\sigma \in \mathfrak{T}_{\Sigma}$, we have

$$1 = \sum_{m=2}^{n} \sum_{[\mathsf{P}] \in \mathcal{P}_{\Sigma}^{m}} \mathsf{B}^{\mathsf{M}}(\vec{\ell}_{\sigma}(\partial\mathsf{P})) + \frac{1}{2} \sum_{[\mathsf{P}] \in \mathcal{P}_{\Sigma}^{\varnothing}} \mathsf{C}^{\mathsf{M}}(\vec{\ell}_{\sigma}(\partial\mathsf{P})),$$
(4.23)

and for a torus T with one boundary component and any $\sigma \in T_T$, we have

$$1 = \sum_{\gamma \in S_{\mathsf{T}}^{\circ}} C^{\mathsf{M}} \big(\ell_{\sigma}(\partial \mathsf{T}), \ell_{\sigma}(\gamma), \ell_{\sigma}(\gamma) \big).$$
(4.24)

4.4.2 In the combinatorial setting

We are now going to prove a recursion for the constant function 1 on the combinatorial Teichmüller space. The initial data are the functions (A^K, B^K, C^K)

$$\begin{aligned} A^{K}(L_{1}, L_{2}, \ell) &= 1, \\ B^{K}(L_{1}, L_{2}, \ell) &= \frac{1}{2L_{1}} \Big([L_{1} - L_{2} - \ell]_{+} - [-L_{1} + L_{2} - \ell]_{+} + [L_{1} + L_{2} - \ell]_{+} \Big), \\ C^{K}(L_{1}, \ell, \ell') &= \frac{1}{L_{1}} [L_{1} - \ell - \ell']_{+}. \end{aligned}$$

$$(4.25)$$

i.e. the same as in the Theorem 4.15 above, but for $F(x) = [x]_+ = \max\{x, 0\}$. Notice that B^K and C^K were already introduced in (2.15) while discussing homotopy classes of embedded pairs of pants. The proof adapts to the combinatorial setting the strategy used by Mirzakhani to obtain Theorem 4.15.

Theorem 4.16. For any connected bordered surface Σ such that $\chi_{\Sigma} < -1$ and any $\mathbb{G} \in \mathbb{T}_{\Sigma}^{\text{comb}}$, we have

$$1 = \sum_{m=2}^{n} \sum_{[\mathsf{P}] \in \mathfrak{P}_{\Sigma}^{m}} \mathsf{B}^{\mathsf{K}}(\vec{\ell}_{\mathbb{G}}(\mathfrak{d}\mathsf{P})) + \frac{1}{2} \sum_{[\mathsf{P}] \in \mathfrak{P}_{\Sigma}^{\varnothing}} \mathsf{C}^{\mathsf{K}}(\vec{\ell}_{\mathbb{G}}(\mathfrak{d}\mathsf{P})),$$
(4.26)

and for a torus T with one boundary component and any $\mathbb{G} \in \mathbb{T}_T^{\text{comb}}$, we have

$$1 = \sum_{\gamma \in S^{\circ}_{T}} C^{K} (\ell_{\mathbb{G}}(\partial T), \ell_{\mathbb{G}}(\gamma), \ell_{\mathbb{G}}(\gamma)).$$
(4.27)

Proof. Consider the case of connected Σ with $\chi_{\Sigma} < -1$. For simplicity, set X_{P}^{K} equal to B^{K} or $\frac{1}{2}C^{K}$ depending on the type of $[P] \in \mathcal{P}_{\Sigma}$. The basic idea to prove such an identity is to write $\ell_{\mathbb{G}}(\partial_{1}\Sigma)$ as a sum of lengths of the edges around $\partial_{1}\Sigma$. Recall that in the proof of Theorem 2.20, we introduced a map⁶ $\mathfrak{T}_{\Sigma}^{comb} \to \mathbb{R}_{\geq 0}^{\mathcal{A}_{\Sigma}}$ that

⁶In the proof of Theorem 2.20, the map was assigning to a point in $\mathcal{T}_{\Sigma}^{\text{comb}}$ a functional on $\mathcal{A}_{\Sigma}^{\text{all}}$. Here we are restricting the functionals to $\mathcal{A}_{\Sigma} \subset \mathcal{A}_{\Sigma}^{\text{all}}$, *i.e.* to arcs with initial point in $\partial_1 \Sigma$, where we recall the convention $\mathcal{A}_{\Sigma} = \mathcal{A}_{\Sigma,1}$ adopted at the beginning of this section.

assigns to a combinatorial structure \mathbb{G} the functional on \mathcal{A}_{Σ} given by

$$\alpha \longmapsto \begin{cases} \ell_{\mathbb{G}}(\partial_{1}\Sigma) \Big(B^{K}(\vec{\ell}_{\mathbb{G}}(\partial P_{\alpha})) - C^{K}(\vec{\ell}_{\mathbb{G}}(\partial P_{\alpha})) \Big) & \text{ if } \alpha \in \mathcal{A}_{\Sigma}^{\mathfrak{m}} \\ \frac{1}{2} \ell_{\mathbb{G}}(\partial_{1}\Sigma) C^{K}(\vec{\ell}_{\mathbb{G}}(\partial P_{\alpha})) & \text{ if } \alpha \in \mathcal{A}_{\Sigma}^{\varnothing} \end{cases}$$

where $[P_{\alpha}] = Q(\alpha)$ is the homotopy class of pair of pants determined by the arc α . Fix once and for all $\mathbb{G} \in \mathcal{T}_{\Sigma}^{comb}$, and denote by $l_{\mathbb{G}} : \mathcal{A}_{\Sigma} \to \mathbb{R}_{\geq 0}$ the value of above map at \mathbb{G} . This function has finite support, and the arcs with non-zero contribution are in bijection with the edges of \mathbb{G} around $\partial_1 \Sigma$. Furthermore, for α dual to an edge *e* around $\partial_1 \Sigma$, the value $l_{\mathbb{G}}(\alpha)$ is the combinatorial length of *e* (*cf.* Lemma 2.30). As a consequence,

$$\ell_{\mathbb{G}}(\mathfrak{d}_{1}\Sigma)=\sum_{\alpha\in\mathcal{A}_{\Sigma}}\iota_{\mathbb{G}}(\alpha).$$

Now from Remark 2.24, we know that the map $Q: \mathcal{A}_{\Sigma} \to \mathcal{P}_{\Sigma}$ is not injective, but has finite fibers. More precisely, we have the following situation.

• If $[P] \in \mathcal{P}_{\Sigma}^{\emptyset}$, then $Q^{-1}([P])$ consists of a single arc $\alpha_0 \in \mathcal{A}_{\Sigma}^{\emptyset}$. Then

0

$$\sum_{\alpha \in Q^{-1}([P])} l_{\mathbb{G}}(\alpha) = l_{\mathbb{G}}(\alpha_0) = \frac{1}{2} \ell_{\mathbb{G}}(\vartheta_1 \Sigma) C^{K}(\vec{\ell}_{\mathbb{G}}(\vartheta P))$$

• If $[P] \in \mathcal{P}_{\Sigma}^{\mathfrak{m}}$, then $Q^{-1}([P])$ consists of three arcs: $\alpha_0 \in \mathcal{A}_{\Sigma}^{\mathfrak{m}}$, α' and its inverse $-\alpha'$ in $\mathcal{A}_{\Sigma}^{\varnothing}$. Then

$$\begin{split} \sum_{\alpha \in Q^{-1}([P])} l_{\mathbb{G}}(\alpha) &= l_{\mathbb{G}}(\alpha_0) + l_{\mathbb{G}}(\alpha') + l_{\mathbb{G}}(-\alpha') \\ &= \ell_{\mathbb{G}}(\partial_1 \Sigma) \Big(B^K(\vec{\ell}_{\mathbb{G}}(\partial P)) - C^K(\vec{\ell}_{\mathbb{G}}(\partial P)) + \frac{1}{2}C^K(\vec{\ell}_{\mathbb{G}}(\partial P)) + \frac{1}{2}C^K(\vec{\ell}_{\mathbb{G}}(\partial P)) \Big) \\ &= \ell_{\mathbb{G}}(\partial_1 \Sigma) B^K(\vec{\ell}_{\mathbb{G}}(\partial P)). \end{split}$$

Finally, as Q is surjective

$$\ell_{\mathbb{G}}(\mathfrak{d}_{1}\Sigma) = \sum_{[P]\in \mathfrak{P}_{\Sigma}} \sum_{\alpha \in Q^{-1}([P])} l_{\mathbb{G}}(\alpha) = \ell_{\mathbb{G}}(\mathfrak{d}_{1}\Sigma) \sum_{[P]\in \mathfrak{P}_{\Sigma}} X_{P}^{K}(\vec{\ell}_{\mathbb{G}}(P)).$$

Dividing by $\ell_{\mathbb{G}}(\partial_1 \Sigma)$ concludes the proof of the identity (4.26). The case of the torus with one boundary component can be handled in a similar way and leads to (4.27).

A notable difference with the original Mirzakhani–McShane identity is that in (4.26) there is a finite number of non-zero terms; this reflects the much simpler dynamics of geodesics in combinatorial surfaces compared to the hyperbolic ones. As in the hyperbolic case, the identity can be reformulated in terms of geometric recursion.

Corollary 4.17. *The initial data* (4.25) *are admissible, and lead by geometric recursion to* $\Xi_{\Sigma}^{K}(\mathbb{G}) = 1$ *for any* Σ *and* $\mathbb{G} \in \mathbb{T}_{\Sigma}^{\text{comb}}$.

Proof. The only thing left to check is the admissibility condition for (A^K, B^K, C^K, D^K) , which follows from the fact that the functions B^K and C^K are supported on small pairs of pants.

We can now apply the integration results of Section 4.2 to the combinatorial Mirzakhani–McShane identity. The continuous integration computes the Kontsevich volumes and Theorem 4.6 establishes again Witten's conjecture [43] first proved by the combination of [28] and [18]: the generating function of ψ -class intersection numbers satisfies Virasoro constraints/topological recursion. The discrete integration calculates the number of integral points in $\mathcal{M}_{g,n}^{\text{comb}}$ and Theorem 4.9 together with Corollary 4.11 with b = c = 0, establishes again Norbury's result [36], *i.e.* the discrete topological recursion for these counts and its connection with Kontsevich volumes.

Corollary 4.18 (Witten's conjecture/Kontsevich theorem). The Kontsevich volumes $V_{g,n}^{K}(L)$ equal $V\Xi_{g,n}^{K}(L)$ and satisfy the topological recursion of Theorem (4.7) with initial data (4.25).

Corollary 4.19 (Norbury's theorem). The lattice point counting functions

$$N_{g,n}(L) = \sum_{\mathbf{G} \in \mathcal{M}_{g,n}^{\operatorname{comb},\mathbb{Z}}(L)} \frac{1}{\#\operatorname{Aut}(\mathbf{G})}$$
(4.28)

equal $N\Xi_{a,n}^{K}(L)$ and satisfy the discrete topological recursion (4.15) with initial data (4.25). Further,

$$\lim_{\substack{k \to \infty \\ k \in d\mathbb{Z}_+}} \frac{N_{g,n}(kL)}{k^{6g-6+2n}} = \begin{cases} 2^{3-2g-n} V_{g,n}^K(L) & \text{if } \sum_{i=1}^n d \cdot L_i \text{ is even,} \\ 0 & \text{otherwise.} \end{cases}$$
(4.29)

The recursive formula for Kontsevich volumes in this integral form firstly appeared in [5], where it is derived by constructing a partition of unity similar to the combinatorial Mirzakhani–McShane identity and exploiting the local torus symmetries on the combinatorial moduli space, whose symplectic quotients are also combinatorial moduli space of higher Euler characteristics. Compared to [5], the new element of the proof that we propose is to make it a *complete* analogue to Mirzakhani's proof of the recursion for Weil– Petersson volumes, by means of Theorem 4.16. Our perspective stresses that, at a general level, recursions between volumes (or more generally, between integrals over the moduli spaces) arise from finer recursions that hold at the geometric level (here between functions on Teichmüller spaces).

4.5 Combinatorial length statistics of multicurves

Following [2, Theorem 10.1] in the hyperbolic world, we can generalise the combinatorial Mirzakhani– McShane identity (4.26) to obtain statistics of combinatorial lengths of primitive multicurves via GR.

Theorem 4.20. Let (A, B, C, D) be admissible initial data and denote by Ξ_{Σ} the associated GR amplitudes. Let $f: \mathbb{R}_+ \to \mathbb{R}$ be a measurable function such that for any $\varepsilon > 0$ and $s \ge 0$, there exists $M_{s,\varepsilon}$ such that

$$\sup_{\ell \geqslant \varepsilon} |f(\ell)| \, \ell^s \leqslant M_{s,\varepsilon}$$

Then, the following initial data are admissible:

$$\begin{split} A[f](L_{1}, L_{2}, L_{3}) &= A(L_{1}, L_{2}, L_{3}), \\ B[f](L_{1}, L_{2}, \ell) &= B(L_{1}, L_{2}, \ell) + A(L_{1}, L_{2}, \ell) f(\ell), \\ C[f](L_{1}, \ell, \ell') &= C(L_{1}, \ell, \ell') + B(L_{1}, \ell, \ell') f(\ell) + B(L_{1}, \ell', \ell) f(\ell') + A(L_{1}, \ell, \ell') f(\ell) f(\ell'), \\ D_{T}[f](\mathbb{G}) &= D_{T}(\mathbb{G}) + \sum_{\gamma \in S_{T}^{\circ}} A(\ell_{\mathbb{G}}(\partial T), \ell_{\mathbb{G}}(\gamma), \ell_{\mathbb{G}}(\gamma)) f(\ell_{\mathbb{G}}(\gamma)). \end{split}$$
(4.30)

Denote by $\Xi_{\Sigma}[f]$ the corresponding GR amplitudes. If for all Σ , Ξ_{Σ} is invariant under all braidings of boundary components of Σ , we have

$$\Xi_{\Sigma}[f](\mathbb{G}) = \sum_{c \in \mathcal{M}'_{\Sigma}} \Xi_{\Sigma_{c}}(\mathbb{G}|_{\Sigma_{c}}) \prod_{\gamma \in \pi_{0}(c)} f(\ell_{\mathbb{G}}(\gamma)),$$
(4.31)

where Σ_c is the bordered surface obtained by cutting Σ along c (the choice of the first boundary component is irrelevant due to the assumed invariance).

Remark 4.21. A useful version of the above result can be stated for combinatorial length statistics of multicurves (not only primitive ones). Consider F: $\mathbb{R}_{>0} \rightarrow (-1, 1)$ a measurable function such that

$$f(x) = \sum_{k \geqslant 1} F(x)^k = \frac{F(x)}{1 - F(x)}$$

satisfies the conditions of Theorem 4.20. Then the GR amplitudes $\Xi_{\Sigma}[f]$ associated to the initial data (4.30) are given by the combinatorial length statistic of multicurves weighted by F:

$$\Xi_{\Sigma}[f](\mathbb{G}) = \sum_{c \in \mathcal{M}_{\Sigma}} \Xi_{\Sigma_{c}}(\mathbb{G}|_{\Sigma_{c}}) \prod_{\gamma \in \pi_{0}(c)} F(\ell_{\mathbb{G}}(\gamma)).$$

Sketch of the proof of Theorem 4.20. The proof repeats the one of [2, Theorem 10.1], with combinatorial lengths instead of hyperbolic ones. We only sketch the induction step here. Notice that, as all series are absolutely convergent, we can apply Fubini's theorem and interchange the summations:

$$\begin{split} \sum_{c \in M'_{\Sigma}} \Xi_{\Sigma_{c}}(\mathbb{G}|_{\Sigma_{c}}) \prod_{\gamma \in \pi_{0}(c)} f(\ell_{\mathbb{G}}(\gamma)) &= \sum_{c \in M'_{\Sigma}} \sum_{[P] \in \mathcal{P}_{\Sigma_{c}}} X_{P}(\mathbb{G}) \Xi_{\Sigma_{c}-P}(\mathbb{G}|_{\Sigma_{c}-P}) \prod_{\gamma \in \pi_{0}(c)} f(\ell_{\mathbb{G}}(\gamma)) \\ &= \sum_{[P] \in \mathcal{P}_{\Sigma}} X_{P}[f](\mathbb{G}) \left(\sum_{\gamma \in M'_{\Sigma-P}} \Xi_{(\Sigma-P)_{\gamma}}(\mathbb{G}|_{(\Sigma-P)_{\gamma}}) \prod_{c \in \pi_{0}(\gamma)} f(\ell_{\mathbb{G}}(\gamma_{c})) \right) \\ &= \sum_{[P] \in \mathcal{P}_{\Sigma}} X_{P}[f](\mathbb{G}) \Xi_{\Sigma-P}[f](\mathbb{G}|_{\Sigma-P}), \end{split}$$

where X_P is either B or $\frac{1}{2}C$ depending on the type of [P]. The second to last equality follows from a case discussion: to interchange the summands we must also sum over the possible ways in which ∂P and c can have homotopic components, weighted by their contributions. For example, for $[P] \in \mathcal{P}_{\Sigma}^{\emptyset}$ with ∂P and c sharing one component, say $\partial_2 P = \gamma_0$, then $[P] \in \mathcal{P}_{\Sigma_c}^{\gamma_0}$. This means that we get a contribution of

$$B(L_1, \ell_{\mathbb{G}}(\gamma_0), \ell_{\mathbb{G}}(\partial_3 P)) f(\ell_{\mathbb{G}}(\gamma_0)) \Xi_{(\Sigma - P)_c}(\mathbb{G}|_{(\Sigma - P)_c}) \prod_{\gamma \in \pi_0(c - \gamma_0)} f(\ell_{\mathbb{G}}(\gamma)),$$

which matches one of the terms in the expression for C[f]. We refer to [2] for the complete proof.

We can then integrate this identity over the combinatorial moduli space with respect to μ_K , and the result can be calculated in two ways: by the topological recursion, and by direct integration. To express the later, we introduce the set of stable graphs

$$\mathcal{G}_{q,n} = M_{\Sigma}^{\prime} / \operatorname{Mod}_{\Sigma}^{\delta}, \tag{4.32}$$

where Σ is a bordered surface of type (g, n). We refer to [1] for a full definition and for the notation, but let us say that a stable graph Γ encodes the class of a primitive multicurve *c* in the following way: vertices $\nu \in V_{\Gamma}$ correspond to connected components of Σ_c and the genera of the components is recorded in a function h: $V_{\Gamma} \to \mathbb{N}$ which is part of the data of Γ ; edges $e \in E_{\Gamma}$ correspond to the components of *c*, and the set of edges incident to ν is denoted by $E(\nu)$; leaves $\lambda \in \Lambda(\nu)$ correspond to boundary components of the surface attached to $\nu \in V_{\Gamma}$; the valency of the vertex ν is denoted by $k(\nu)$. The automorphism group of Γ is identified with that of the multicurve *c*.

Corollary 4.22. Assume that (A, B, C, D) are strongly admissible and f is a measurable function for which there exists $\eta' \in [0, 2)$ such that

$$\sup_{\ell>0} |f(\ell)| \ell^{\eta'} < +\infty.$$

Then (A[f], B[f], C[f], D[f]) are strongly admissible and $V\Xi_{g,n}[f](L) = \int_{\mathcal{M}_{g,n}^{comb}(L)} \Xi_{g,n}[f] d\mu_K$ satisfy the topological recursion (4.12) with this initial data. Besides, we have

$$V\Xi_{g,n}[f](L_1,\ldots,L_n) = \sum_{\Gamma \in \mathfrak{G}_{g,n}} \frac{1}{\#\operatorname{Aut}(\Gamma)} \int_{\mathbb{R}^{+}_+} \prod_{\nu \in V_{\Gamma}} V\Xi_{h(\nu),k(\nu)} \left((\ell_e)_{e \in E(\nu)}, (L_{\lambda})_{\lambda \in \Lambda(\nu)} \right) \prod_{e \in E_{\Gamma}} \ell_e f(\ell_e) \, d\ell_e.$$
(4.33)

Proof. The topological recursion follows from Theorem 4.7 and from the second part from the integration formula of Proposition 3.13.

We can also obtain an analogous result for the discrete integration, substituting $V\Xi_{h(\nu),k(\nu)}$ with $N\Xi_{h(\nu),k(\nu)}$ and the integral over $\mathbb{R}_{+}^{E_{\Gamma}}$ with the sum over edge decorations of the form $\ell: E_{\Gamma} \to \mathbb{Z}_{+}$. However, we cannot directly apply Theorem 4.9, as the twisted initial data are in general not supported on small pairs of pants. Instead, we introduce an (*a priori*) different lattice count, for $L \in \mathbb{Z}_{+}^{n}$

$$\mathsf{N}\Xi_{g,n}[f](\mathsf{L}) := \sum_{\Gamma \in \mathfrak{G}_{g,n}} \frac{1}{\#\mathsf{Aut}(\Gamma)} \sum_{\ell:\mathsf{E}_{\Gamma} \to \mathbb{Z}_{+}} \prod_{\nu \in V_{\Gamma}} \mathsf{N}\Xi_{\mathsf{h}(\nu),\mathsf{k}(\nu)} \big((\ell_{e})_{e \in \mathsf{E}(\nu)}, (\mathsf{L}_{\lambda})_{\lambda \in \Lambda(\nu)} \big) \prod_{e \in \mathsf{E}_{\Gamma}} \ell_{e} \, \mathsf{f}(\ell_{e}). \tag{4.34}$$

We omit the proof of the next proposition: it follows the same scheme as the proof of Theorem 4.20, in the simpler situation where multicurves are replaced by their mapping class group orbits (stable graphs), and with integrals replaced by discrete sums. The key principle is that topological recursion is preserved under the twisting operation.

Proposition 4.23. Let (A, B, C, D) and $N\Xi_{g,n}$ as in Theorem 4.9. Let $f: \mathbb{Z}_+ \to \mathbb{R}$ be a continuous function such that for any s > 0,

$$\sup_{\ell\in\mathbb{Z}_+}|\mathsf{f}(\ell)|\,\ell^{\mathrm{s}}<+\infty.$$

Then $N\Xi_{q,n}[f](L)$ *is finite and, for* $2g - 2 + n \ge 2$ *, it is calculated by the discrete topological recursion formula*

$$\begin{split} \mathsf{N}\Xi_{g,n}[f](\mathsf{L}_{1},\ldots,\mathsf{L}_{n}) &= \sum_{m=2}^{n} \sum_{\ell \geqslant 1} \ell \, \mathsf{B}_{\mathbb{Z}}[f](\mathsf{L}_{1},\mathsf{L}_{m},\ell) \, \mathsf{N}\Xi_{g,n-1}[f](\ell,\mathsf{L}_{2},\ldots,\widehat{\mathsf{L}_{m}},\ldots,\mathsf{L}_{n}) \\ &+ \frac{1}{2} \sum_{\ell,\ell' \geqslant 1} \ell \ell' \, \mathsf{C}_{\mathbb{Z}}[f](\mathsf{L}_{1},\ell,\ell') \Bigg(\mathsf{N}\Xi_{g-1,n+1}[f](\ell,\ell',\mathsf{L}_{2},\ldots,\mathsf{L}_{n}) \\ &+ \sum_{\substack{\mathsf{h}+\mathsf{h}'=\mathsf{g}\\\mathsf{J}\sqcup J'=\{\mathsf{L}_{2}\ldots,\mathsf{L}_{n}\}} \mathsf{N}\Xi_{\mathsf{h},1+\#J}[f](\ell,\mathsf{J}) \, \mathsf{N}\Xi_{\mathsf{h}',1+\#J'}[f](\ell',J') \Bigg), \end{split}$$
(4.35)

with conventions $N\Xi_{0,1}[f] = 0$ and $N\Xi_{0,2}[f] = 0$ and base cases

$$N\Xi_{0,3}[f](L_1, L_2, L_3) = N\Xi_{0,3}(L_1, L_2, L_3) = A_{\mathbb{Z}}(L_1, L_2, L_3),$$

$$N\Xi_{1,1}[f](L_1) = N\Xi_{1,1}(L_1) + \frac{1}{2} \sum_{\ell \ge 1} \ell A_{\mathbb{Z}}(L_1, \ell, \ell) f(\ell).$$
(4.36)

4.6 Kontsevich amplitudes for hyperbolic GR

On the one hand, Corollary 4.17 says that the combinatorial GR amplitudes Ξ_{Σ}^{K} for the initial data (A^{K}, B^{K}, C^{K}) coincide with the constant function 1 on $\mathcal{T}_{\Sigma}^{comb}$. On the other hand, we can consider the hyperbolic GR amplitudes Ω_{Σ}^{K} associated to the same initial data. They are rather non-trivial functions of $\sigma \in \mathcal{T}_{\Sigma}$, but since

the topological recursion formulae are the same in the combinatorial and hyperbolic setting, we have

$$V_{g,n}^{K}(L) = \int_{\mathcal{M}_{g,n}^{comb}(L)} \Xi_{g,n}^{K} d\mu_{K} = \int_{\mathcal{M}_{g,n}(L)} \Omega_{g,n}^{K} d\mu_{WP} = \int_{\overline{\mathfrak{M}}_{g,n}} \exp\left(\sum_{i=1}^{n} \frac{L_{i}^{2}}{2} \psi_{i}\right)$$
(4.37)

for any $L = (L_1, ..., L_n)$. Here we propose a geometric interpretation of Ω_{Σ}^K and give some of its basic properties.

4.6.1 A geometric interpretation

In the following we discuss the combinatorial analogue of the geometric reasoning at the core of Mirzakhani's proof of the Mirzakhani–McShane identities. Consider a connected bordered surface Σ . Recall from § 2.2.2 the construction of the spine as a subset $sp_{\sigma}(\Sigma) \subset \Sigma$ that depends on a hyperbolic marking (X, f) representing $\sigma \in \mathfrak{T}_{\Sigma}$. We also denote by $sp'_{\sigma}(\Sigma) \subset sp_{\sigma}(\Sigma)$ the complement of the set of vertices of the spine.

For a given $\sigma \in \mathfrak{T}_{\Sigma}$, we equip $\partial_1 \Sigma$ with the curvilinear measure induced by σ .

Lemma 4.24. For all but finitely many $x \in \partial_1 \Sigma$, the geodesic shot from x orthogonally to $\partial_1 \Sigma$ intersects the spine for the first time at a point $s_{\Sigma}(x) \in sp'_{\sigma}(\Sigma)$.

Proof. Cutting out the spine, we have a cylinder around each boundary component of the surface. Consider the one around $\partial_1 \Sigma$ and take a geodesic that realises the distance between the two boundaries of the cylinder. Cutting along this, we obtain a hyperbolic polygon. As there are no hyperbolic triangles with two right angles, every geodesic shot orthogonally from $x \in \partial_1 \Sigma$ must reach the boundary corresponding to the spine at a certain point $s_{\Sigma}(x)$, that is not a vertex for all but finitely many x that are rib-ends on $\partial_1 \Sigma$.

Let $x \in \partial_1 \Sigma$ be such that $s_{\Sigma}(x)$ exists and is not a vertex of the spine. By definition of the spine and the vertices, there exists a unique second geodesic joining $s_{\Sigma}(x)$ to $\partial_i \Sigma$ for some $i \in \{1, ..., n\}$. The union of these two geodesics form a piecewise geodesic arc, denoted by γ_x . This arc determines a unique homotopy class of embedded pair of pants, and we denote by P_x its representative that has geodesic boundaries in Σ . This pair of pants has a spine $sp_{\sigma}(P_x)$ of its own, and we can ask whether $s_{\Sigma}(x)$ is still part of $sp_{\sigma}(P_x)$. With these notations, we define the following process.

Definition 4.25. Choose a random point $x \in \partial_1 \Sigma$ uniformly on $\partial_1 \Sigma$ – for the measure coming from the curvilinear abscissa induced by σ . If the geodesic shot from x orthogonally to $\partial_1 \Sigma$ hits for the first time $sp_{\sigma}(\Sigma)$ at a vertex, or if $s_{\Sigma}(x) \notin sp_{\sigma}(P_x)$, quit the process. Otherwise, consider the bordered hyperbolic surface $\Sigma - P_x$. If it is empty, we have finished the process successfully; if not, we repeat it with $\Sigma - P_x^7$, each connected component of $\Sigma - P_x$ being treated independently. Denote by $\Pi_{\Sigma}(\sigma)$ the probability that the process ends successfully, *i.e.* by giving a pants decomposition. It is clear that $\Pi_{\Sigma}(\sigma)$ only depends on the projection $[\sigma] \in \mathcal{M}_{\Sigma}$ and the process makes no reference to a marking.

Proposition 4.26. We have $\Pi_{\Sigma}(\sigma) = \Omega_{\Sigma}^{K}(\sigma)$. In particular, $\Omega_{\Sigma}^{K}(\sigma) \in [0, 1]$, and if we consider the above process for a random hyperbolic surface of type (g, n) with fixed boundary lengths $L \in \mathbb{R}^{n}_{+}$ (with respect to the Weil–Petersson measure on $\mathcal{M}_{\Sigma}(L)$), we have

$$\mathbb{E}\left[\Pi_{\Sigma}(\sigma) \mid \ell_{\sigma}(\partial\Sigma) = L\right] = \frac{V_{g,n}^{K}(L)}{V_{g,n}^{WP}(L)},$$
(4.38)

where $V_{q,n}^{WP}(L)$ are the Weil–Petersson volumes of $\mathcal{M}_{g,n}(L)$.

⁷The arcs give an order to the boundary components of P_x , and by a similar process used in [2, Section 2.3] we can define the first boundary on $\Sigma - P_x$.





(a) The piecewise geodesic γ_x (in blue) and $sp_{\sigma}(\Sigma)$ (in red).

(b) The piecewise geodesic γ_x (in blue) and $sp_{\sigma}(P_x)$ (in orange).

Figure 24: Example of a successful first step of the process.

Proof. We prove the result by induction on 2g - 2 + n > 0. The base case (g, n) = (0, 3) is trivial: for any pair of pants P, we have $\Pi_P \equiv 1$. Furthermore, the following argument can be adjusted to prove that for a one-holed torus T,

$$\Pi_{\mathsf{T}}(\sigma) = \sum_{\gamma \in S^\circ_{\mathsf{T}}} \, C^K \big(\ell_\sigma(\eth \mathsf{T}), \ell_\sigma(\gamma), \ell_\sigma(\gamma) \big) = \mathsf{D}_{\mathsf{T}}(\sigma).$$

Consider Σ of type (g, n) and suppose now by induction that the proposition holds for surfaces of Euler characteristic $\chi > -(2g - 2 + n)$. By definition of the process, we can write $\Pi_{\Sigma}(\sigma)$ as a sum over $\cup_{x \in \partial_1 \Sigma} P_x$ (where $P_x = \emptyset$ if x does not define a pair of pants):

$$\Pi_{\Sigma}(\sigma) = \sum_{[P] \in \cup_{x \in \mathfrak{d}_{1}\Sigma} P_{x}} Y_{P}(\sigma) \Pi_{\Sigma - P} \big(\sigma|_{\Sigma - P}\big).$$

Here, we identify [P] with its representative $P \subseteq \Sigma$ having geodesic boundaries, and we set

$$Y_{\mathsf{P}}(\sigma) = \frac{\ell_{\sigma}(\mathfrak{Y}_{\mathsf{P}}(\sigma))}{\ell_{\sigma}(\mathfrak{d}_{1}\Sigma)}, \qquad \mathfrak{Y}_{\mathsf{P}}(\sigma) = \left\{ x \in \mathfrak{d}_{1}\Sigma \mid s_{\Sigma}(x) \in \operatorname{sp}_{\sigma}(\mathsf{P}) \right\}.$$
(4.39)

By induction hypothesis, we have $\Pi_{\Sigma-P}(\sigma|_{\Sigma-P}) = \Omega_{\Sigma-P}^{K}(\sigma|_{\Sigma-P})$. So, we only need to show that $Y_{P}(\sigma) = X_{P}^{K}(\vec{\ell}_{\sigma}(\partial P))$ and that the [P]'s outside $\cup_{x \in \partial_{1}\Sigma} P_{x}$ do not contribute to the GR sum defining Ω_{Σ}^{K} .

For the latter, we observe that if $[P] \notin \bigcup_{x \in \partial_1 \Sigma} P_x$, then all points of $sp_{\sigma}(P)$ incident to $\partial_1 \Sigma$ are equidistant to an internal boundary of P. Therefore we see that the spine of $\sigma|_P$ has the whole of the first boundary adjacent to an internal boundary, and therefore from Lemma 2.28 we see that $X_P^K(\vec{\ell}_{\sigma}(\partial P)) = 0$. So, we can indeed restrict the range of the GR sum defining Ω_{Σ}^K to $[P] \in \bigcup_{x \in \partial_1 \Sigma} P_x$.

For the former, we first give a description of X_P^K similar to (4.39). Let $s_P(x)$ be the first intersection point (if it exists) between the geodesic shot from x orthogonally to $\partial_1 \Sigma$ and the spine of P – considered as a subset of Σ . We introduce the sets

$$\mathfrak{sp}_{\sigma}(\mathsf{P}) = \begin{cases} \left\{ \begin{array}{l} s \in sp'_{\sigma}(\mathsf{P}) \mid s \text{ is incident to } \partial_{\mathfrak{m}}\Sigma \text{ and } \partial_{1}\Sigma \text{ or only to } \partial_{1}\Sigma \right\} & \text{if } [\mathsf{P}] \in \mathfrak{P}_{\Sigma}^{\mathfrak{m}} \\ \left\{ \begin{array}{l} s \in sp'_{\sigma}(\mathsf{P}) \mid s \text{ is incident to } \partial_{1}\Sigma \text{ on both sides} \end{array} \right\} & \text{if } [\mathsf{P}] \in \mathfrak{P}_{\Sigma}^{\varnothing} \\ \mathfrak{X}_{\mathsf{P}}(\sigma) = \left\{ x \in \partial_{1}\Sigma \mid s_{\mathsf{P}}(x) \in \mathfrak{sp}_{\sigma}(\mathsf{P}) \right\} \end{cases}$$

where it is understood that if $s_P(x)$ does not exist, x is not in $\mathcal{X}_P(\sigma)$. The properties of B^K and C^K stressed in Lemma 2.28 show that

$$X_{\mathrm{P}}^{\mathrm{K}}(\vec{\ell}_{\sigma}(\partial \mathrm{P})) = \frac{\ell_{\sigma}(\mathcal{X}_{\mathrm{P}}(\sigma))}{\ell_{\sigma}(\partial_{1}\Sigma)}.$$

It remains to justify that $\mathfrak{X}_{P}(\sigma) = \mathfrak{Y}_{P}(\sigma)$.

(\subseteq) Consider $x \in \mathfrak{X}_{P}(\sigma)$. From the definition of $\mathfrak{sp}_{\sigma}(P)$, it is clear that $\mathfrak{sp}_{\sigma}(P) \subseteq sp'_{\sigma}(\Sigma)$ and $s_{P}(x)$ is equidistant to $\partial\Sigma$ in exactly two ways. In particular,

$$\operatorname{dist}_{\sigma}(\mathsf{s}_{\mathsf{P}}(\mathsf{x}), \mathfrak{d}_{1}\Sigma) = \operatorname{dist}_{\sigma}(\mathsf{s}_{\mathsf{P}}(\mathsf{x}), \mathfrak{d}_{\mathfrak{i}}\Sigma) < \operatorname{dist}_{\sigma}(\mathsf{s}_{\mathsf{P}}(\mathsf{x}), \mathfrak{d}\Sigma - (\mathfrak{d}_{1}\Sigma \cup \mathfrak{d}_{\mathfrak{i}}\Sigma))$$

for some $i \in \{1, ..., n\}$. Thus, $s_{\Sigma}(x) = s_{P}(x)$ and clearly $s_{P}(x) \in sp_{\sigma}(P)$.

(2) Consider $x \in \mathcal{Y}_{P}(\sigma)$. Then we have $s_{P}(x) \in sp_{\sigma}(\Sigma)$, so that $s_{P}(x) = s_{\Sigma}(x)$ and clearly $s_{P}(x) \in \mathfrak{sp}_{\sigma}(P)$.

4.6.2 Properties of the Kontsevich amplitudes

The geometric interpretation of Ω_{Σ}^{K} already shows the non-trivial fact that such functions take values in $[0,1] \subset \mathbb{R}$. In the remaining part of this section we are going to show two other properties of Ω_{Σ}^{K} .

Non-invariance under all braidings. From their definition, the GR amplitudes are *a priori* only invariant under mapping classes that preserve the first boundary; the invariance under braidings of $\partial_1 \Sigma$ with some $\partial_m \Sigma$ for $m \neq 1$ is not guaranteed. This full invariance turns out to hold for the hyperbolic GR amplitudes $\Omega_{\Sigma}^{\rm M}$ and the combinatorial GR amplitudes $\Xi_{\Sigma}^{\rm K}$, for the obvious reason that they are identically 1. One could wonder if the full invariance also holds for $\Omega_{\Sigma}^{\rm K}$, but we show this is already not the case for four-holed spheres.

Proposition 4.27. Let X be a four-holed sphere, and take $\gamma \in S_X^\circ$ separating $\partial_1 X$ and $\partial_2 X$ from $\partial_3 X$ and $\partial_4 X$. Consider $\rho = [\phi \colon X \to X] \in Mod_X$ the involution that fixes γ and such that $\phi(\partial_1 X) = \partial_3 X$ and $\phi(\partial_2 X) = \partial_4 X$. Then $\rho.\Omega_X^K \neq \Omega_X^K$.

Proof. The curve γ together with an arc from $\partial_1 X$ to $\partial_3 X$, determine a seamed pants decomposition of X, and we denote by $(L, \ell, \tau) \in \mathbb{R}^4_+ \times \mathbb{R}_+ \times \mathbb{R}$ the corresponding combinatorial Fenchel–Nielsen coordinates. If we choose $\sigma \in \mathfrak{T}_X$ such that $\ell_{\sigma}(\gamma)$ is small enough, then by the collar lemma we can make the length of any simple closed geodesic intersecting γ greater than max{ $L_1 + L_2, L_3 + L_4$ }. As $B^K(L_i, L_j, \ell)$ vanishes for $\ell \ge L_i + L_j$, we have

$$\Omega_{X}^{K}(L_{1}, L_{2}, L_{3}, L_{4}, \ell, \tau) = B^{K}(L_{1}, L_{2}, \ell)$$

and

$$\rho.\Omega_X^K(L_1, L_2, L_3, L_4, \ell, \tau) = \Omega_X^K(L_3, L_4, L_1, L_2, \ell, \tau) = B^K(L_3, L_4, \ell).$$

Choosing $(L_1, \ldots, L_4) \in \mathbb{R}^4_+$ such that $B^K(L_1, L_2, \ell) \neq B^K(L_3, L_4, \ell)$, we obtain the thesis.

1

This is the first example where the non-invariance of some GR amplitudes can be established. Nevertheless, we know that after integration over the moduli space against μ_{WP} , we obtain the symmetric polynomials (4.37). This is clear by the definition of the Kontsevich volumes, but it can also be directly proved from the theory of the topological recursion, as (A^K, B^K, C^K) obey a set of quadratic relations that guarantee the invariance of the corresponding TR amplitudes under permutations of L_1, \ldots, L_n – see *e.g.* [8].

Support with non-empty complement. We prove that the Kontsevich amplitudes are actially zero on some open subset. In order to achieve this, we use geometric recursion to construct an auxiliary function which has the same support as the Kontsevich amplitude and takes integer values.

Consider the following geometric recursion initial data

$$\begin{split} &\mathsf{A}(\mathsf{L}_1,\mathsf{L}_2,\mathsf{L}_3) = 1, \\ &\mathsf{B}(\mathsf{L}_1,\mathsf{L}_2,\ell) = \theta(\mathsf{L}_1 + \mathsf{L}_2 - \ell), \\ &\mathsf{C}(\mathsf{L}_1,\ell,\ell') = \theta(\mathsf{L}_1 - \ell - \ell'), \end{split}$$

where $\theta(x) = \mathbf{1}_{\mathbb{R}_+}(x)$ is the Heaviside theta function. Let $\Omega_{\Sigma}^{\theta} \in \text{Mes}(\mathfrak{T}_{\Sigma})$ be the geometric recursion amplitude associated to Σ computed with respect to hyperbolic lengths. The value $\Omega_{\Sigma}^{\theta}(\sigma) \in \mathbb{N}$ is counting the number of σ -small pair of pants decompositions⁸ of Σ . In particular, these are piecewise constant functions on \mathfrak{T}_{Σ} with values in non-negative integers and, for Σ connected of type (g, n), we see that Ω_{Σ}^{θ} is bounded by $\prod_{k=1}^{2g-2+n} 6k$. The following lemma easily follows by induction on 2g - 2 + n.

Lemma 4.28. The GR amplitudes Ω_{Σ}^{θ} have the same support in $\mathfrak{T}_{\Sigma}^{\text{comb}}$ as Ω_{Σ}^{K} , and the TR amplitudes $V\Omega_{g,n}^{\theta}(L) = \int_{\mathfrak{M}_{g,n}(L)} \Omega_{\Sigma}^{\theta} d\mu_{WP}$ are homogeneous polynomials of degree 6g - 6 + 2n.

Corollary 4.29. For Σ not of type (0,3), the support of Ω_{Σ}^{K} has non-empty complement.

Proof. The Weil–Petersson volumes $V_{g,n}^{WP}(L)$ are polynomials in $L = (L_1, ..., L_n)$ with non-zero constant term while $V\Omega_{g,n}^{\theta}(L)$ are homogeneous of positive degree for $(g, n) \neq (0, 3)$. Hence

$$\lim_{L \to 0} \frac{V\Omega_{g,n}^{\theta}(L)}{V_{q,n}^{WP}(L)} = 0$$

It shows that $V\Omega_{g,n}^{\theta}(L)$ is strictly less than $V_{g,n}^{WP}(L)$ for small L and we deduce there exists an open set of $\mathcal{T}_{\Sigma}^{comb}(L)$ for small L on which $\Omega_{\Sigma}^{\theta} < 1$ and therefore on which $\Omega_{\Sigma}^{\theta} = 0$, hence Ω_{Σ}^{K} vanishes.

Remark 4.30. To conclude, we remark that the Laplace transform of the TR amplitudes $V\Omega_{g,n}^{\theta}(L)$, namely

$$\omega_{g,n}(z_1,\ldots,z_n) = \left(\int_{\mathbb{R}^n_+} \prod_{i=1}^n dL_i L_i e^{-z_i L_i} V\Omega^{\theta}_{g,n}(L_1,\ldots,L_n)\right) dz_1 \otimes \cdots \otimes dz_n$$

satisfy a topological recursion à *la* Eynard–Orantin [20]:

$$\omega_{g,n}(z_1,\ldots,z_n) = \underset{z\to 0}{\text{Res}} \ \mathsf{K}(z_1,z) \bigg(\omega_{g-1,n+1}(z,-z,z_2,\ldots,z_n) + \sum_{\substack{h+h'=g\\J \sqcup J' = \{z_2,\ldots,z_n\}}}^{\operatorname{no}(0,1)} \omega_{h,1+\#J}(z,J) \otimes \omega_{h',1+\#J'}(-z,J') \bigg),$$

where

$$K(z_1,z) = \frac{1}{2z(z-z_1)^2} \frac{dz_1}{dz}$$
 and $\omega_{0,2}(z_1,z_2) = \frac{dz_1 \otimes dz_2}{(z_1-z_2)^2}.$

However, in this case the recursion kernel $K(z_1, z)$ does not have the usual structure of [20] and the multidifferentials $\omega_{g,n}(z_1, \ldots, z_n)$ are not symmetric under permutation of their n variables. It seems to be the first known example where a non-symmetric topological recursion yields a geometrically meaningful quantity.

⁸Such pants decomposition are "rooted", in the sense that they always bound the first boundary component of the surface at each step (cf. [2, Section 3.6])

5 Rescaling flow: from hyperbolic to combinatorial geometry

The hyperbolic and combinatorial Teichmüller spaces can be identified via the spine homeomorphism sp: $\mathcal{T}_{\Sigma} \rightarrow \mathcal{T}_{\Sigma}^{\text{comb}}$ defined in § 2.2.2. We introduce a flow that interpolates between their respective geometries, coming from the work of Mondello and Do [35, 19].

Definition 5.1. Let $\mathbb{G} \in \mathcal{T}_{\Sigma}^{comb}(L)$ and $\beta \in \mathbb{R}_+$. Define $\beta \mathbb{G} \in \mathcal{T}_{\Sigma}^{comb}(\beta L)$ to be the combinatorial structure represented by the same marked ribbon graph, but with all lengths multiplied by β . This define a flow, called the *rescaling flow*, on $\mathcal{T}_{\Sigma}^{comb}$ which preserves the strata and is continuous. It can be lifted to $\Phi_{\beta}: \mathcal{T}_{\Sigma}(L) \to \mathcal{T}_{\Sigma}(\beta L)$ by the spine map: for $\sigma \in \mathcal{T}_{\Sigma}(L)$, set

$$\Phi_{\beta}(\sigma) = \sigma^{\beta} = \operatorname{sp}^{-1}(\beta \operatorname{sp}(\sigma)) \in \mathfrak{T}_{\Sigma}(\beta L).$$
(5.1)

The maps Φ_{β} and β are Mod_{Σ} -equivariant, and thus descend to the moduli spaces $\mathcal{M}_{g,n}$ and $\mathcal{M}_{g,n}^{comb}$. We also define the map $R_{\beta} : \mathcal{T}_{\Sigma}^{comb}(L) \to \mathcal{T}_{\Sigma}(\beta L)$ by postcomposing β with sp⁻¹.

$$\begin{array}{c} \mathfrak{T}_{\Sigma}(L) \xrightarrow{sp} \mathfrak{T}_{\Sigma}^{comb}(L) \\ \Phi_{\beta} \downarrow & \downarrow^{\beta} \\ \mathfrak{T}_{\Sigma}(\beta L) \xleftarrow{sp^{-1}} \mathfrak{T}_{\Sigma}^{comb}(\beta L) \end{array}$$

The large β asymptotic of the rescaling flow has been previously studied pointwise on \mathcal{T}_{Σ} .

Theorem 5.2. [35, 19] For any bordered surface Σ , the metric space $(\Sigma, \beta^{-1}\sigma^{\beta})$ converges in the Gromov–Hausdorff topology to the metric ribbon graph $\operatorname{sp}(\sigma)$ when $\beta \to \infty$. Moreover, the Poisson structure $\beta^2 R^*_\beta \pi_{WP}$ converges pointwise to π_K on the top-dimensional strata of $\mathcal{T}^{comb}_{\Sigma}$.

Here we shall complete this description by giving effective bounds on the thick part of the Teichmüller space for lengths, and on compacts for twist parameters, showing convergence of the Fenchel–Nielsen coordinates. Under the flow, combinatorial geometry appears as a kind of tropicalisation of hyperbolic geometry. We apply these results to prove that Fenchel–Nielsen coordinates equip T_{Σ}^{comb} with a natural piecewise linear structure, and that the flow carries the hyperbolic GR/TR to the combinatorial GR/TR.

5.1 Convergence of lengths and twists

5.1.1 Convergence of lengths

We first obtain an effective comparison between lengths of simple closed curves along the flow, by refining the arguments of [19].

Definition 5.3. Let Σ be bordered surface. The hyperbolic (resp. combinatorial) *systole* with respect to $\sigma \in \mathcal{T}_{\Sigma}$ (resp. $\mathbb{G} \in \mathcal{T}_{\Sigma}^{comb}$) as

$$\operatorname{sys}_{\sigma} = \inf_{\gamma \in S_{\Sigma}} \ell_{\sigma}(\gamma), \qquad \operatorname{sys}_{\mathbb{G}} = \inf_{\gamma \in S_{\Sigma}} \ell_{\mathbb{G}}(\gamma). \tag{5.2}$$

Proposition 5.4. Let Σ be a connected bordered surface of topology (g, n). For any $\mathbb{G} \in \mathbb{T}_{\Sigma}^{comb}$, we denote $\sigma = sp^{-1}(\mathbb{G})$ and $\sigma^{\beta} = sp^{-1}(\beta\mathbb{G})$. Then for all $\beta \ge 1$, we have

$$\forall \gamma \in S_{\Sigma}, \qquad \ell_{\mathbb{G}}(\gamma) \leqslant \frac{\ell_{\sigma^{\beta}}(\gamma)}{\beta}.$$
(5.3)

Moreover, for any $\epsilon > 0$ there exists $\beta_{\epsilon} \ge 1$ and $\kappa_{\epsilon} > 0$ depending only on ϵ and the topological type of Σ , such that for any $\beta \ge \beta_{\epsilon}$ and $\mathbb{G} \in \mathbb{T}_{\Sigma}^{\text{comb}}$ for which the systole in \mathbb{G} is bounded from below by ϵ , we have

$$\forall \gamma \in \mathsf{S}_{\Sigma}, \qquad \frac{\ell_{\sigma^{\beta}}(\gamma)}{\beta} \leqslant \left(1 + \frac{\kappa_{\varepsilon}}{\beta}\right) \ell_{\mathbb{G}}(\gamma). \tag{5.4}$$

We can take κ_{ε} *and* β_{ε} *increasing with* 2g - 2 + n*.*
Corollary 5.5. *For any* $\gamma \in S_{\Sigma}$ *and* $\sigma \in T_{\Sigma}$ *, we have*

$$\lim_{\beta \to \infty} \frac{\ell_{\sigma^{\beta}}(\gamma)}{\beta} = \ell_{\mathrm{sp}(\sigma)}(\gamma), \tag{5.5}$$

and the limit is uniform when $sp(\sigma)$ belongs to the thick part of $\mathcal{T}_{\Sigma}^{comb}$.

Proof of Proposition 5.4. If γ is a boundary curve, there is nothing to prove since $\ell_{\sigma^{\beta}}(\gamma) = \beta \ell_{\sigma}(\gamma)$. We now assume $\gamma \in S_{\Sigma}^{\circ}$ and start from the last inequality in the proof of [19, Lemma 11]:

$$\forall \gamma \in S_{\Sigma}^{\circ} \qquad \ell_{\mathbb{G}}(\gamma) \leqslant \frac{\ell_{\sigma^{\beta}}(\gamma)}{\beta} \leqslant \ell_{\mathbb{G}}(\gamma) + \frac{2E_{\mathbb{G}}(\gamma)r_{\beta}}{\beta}, \tag{5.6}$$

where $E_{\mathbb{G}}(\gamma)$ is the number (with multiplicity) of edges along which the non-backtracking representative of γ in \mathbb{G} travels, and r_{β} is the maximal of rib lengths with respect to σ^{β} . This already gives the lower bound. Recall that \mathbb{G} contains at most 6g - 6 + 3n edges. We say that an edge is ε -big if its length in \mathbb{G} is larger or equal to $\varepsilon/(6g - 6 + 3n)$, and that it is ε -short otherwise. Denote by $E_{\mathbb{G}}^{(\varepsilon)}(\gamma)$ the number (with multiplicity) of ε -big edges along which γ travels. If we assume that $sys_{\mathbb{G}} \ge \varepsilon$, the union of the ε -short edges appearing in γ must be a forest. As γ is a closed loop, it has to exit each tree it passes through via an ε -big edge. Hence $E_{\mathbb{G}}(\gamma) \le (6g - 6 + 3n)E_{\mathbb{G}}^{(\varepsilon)}(\gamma)$. Observing that $\frac{\varepsilon}{6g - 6 + 3n}E_{\mathbb{G}}^{(\varepsilon)}(\gamma) \le \ell_{\mathbb{G}}(\gamma)$, we obtain

$$\mathsf{E}_{\mathbb{G}}(\gamma) \leqslant \frac{(6g-6+3n)^2}{\varepsilon}\, \ell_{\mathbb{G}}(\gamma).$$

We must now bound uniformly the maximum of rib lengths for σ^{β} . For this purpose, we bound the distance between any point of the surface to the boundary for the metric σ^{β} . Recall that the injectivity radius at a point $q \in \Sigma$ for the hyperbolic structure σ , here denoted $\mathfrak{r}_{\sigma}(q)$, is the supremum of all $\rho > 0$ such that there is a locally isometric embedding of an open hyperbolic disk of radius ρ . Since the area of (Σ, σ^{β}) is $2\pi(2g - 2 + n)$, which must be greater or equal to the area of such disks, we have

$$\mathfrak{r}_{\sigma}(q) \leqslant \sqrt{2(2g-2+n)}. \tag{5.7}$$

Besides, it is clear that

$$\mathfrak{r}_{\sigma}(q) = \min\left\{\frac{1}{2}\operatorname{sys}_{\sigma}(q),\operatorname{dist}_{\sigma}(q,\partial\Sigma)\right\},\tag{5.8}$$

where $sys_{\sigma}(q)$ is defined to be the infimum over the lengths of non-constant geodesic loops based at q. We apply this to the hyperbolic structure σ^{β} . Using the lower bound (5.6), we remark that

$$\beta \epsilon \leq \mathrm{sys}_{\beta \mathbb{G}} \leq \mathrm{sys}_{\sigma^{\beta}} \leq \mathrm{sys}_{\sigma^{\beta}}(q)$$

If $\beta \geqslant \beta_{\varepsilon} = \frac{2\sqrt{2(2g-2+n)}}{\varepsilon}$, we deduce from (5.7) and (5.8) that

$$\forall q \in \Sigma, \qquad \operatorname{dist}_{\sigma^{\beta}}(q, \partial \Sigma) \leqslant \sqrt{2(2g-2+n)}.$$

In particular, choosing for q the vertices of G, we find $r_{\beta} \leq \sqrt{2(2g-2+n)}$. Together with (5.6), it shows that

$$\forall \beta \geqslant \beta_{\varepsilon} \quad \forall \gamma \in S_{\Sigma}^{\circ}, \qquad \ell_{G}(\gamma) \leqslant \frac{\ell_{\sigma^{\beta}}(\gamma)}{\beta} \leqslant \left(1 + \frac{18\sqrt{2(2g - 2 + n)^{5/2}}}{\beta\varepsilon}\right) \ell_{G}(\gamma),$$

which gives the thesis.

5.1.2 Convergence of twists

Studying the convergence of twists along the flow by a direct geometric method requires bounds on the distance between the hyperbolic and combinatorial seams. Instead of following this direction, we use the hyperbolic (9g - 9 + 3n)-theorem to write the hyperbolic twists in terms of hyperbolic lengths of certain curves, to show that these formulae converge with help of Proposition 5.4, and compare the limit to the expressions for the combinatorial twists underlying the combinatorial (9g - 9 + 3n)-theorem established in Section 2.6.

Proposition 5.6. Let Σ be a bordered surface of type (g, n), fix a seamed pants decomposition and let $(\tau_i)_{i=1}^{3g-3+n}$ be the associated combinatorial twist parameters. For any compact $K \subset \mathfrak{T}_{\Sigma}^{comb}$ and $\mathbb{G} \in K$, we denote $\sigma = sp^{-1}(\mathbb{G})$ and $\sigma^{\beta} = sp^{-1}(\beta\mathbb{G})$. There exists constants $\beta_K \ge 1$ and $c_K > 0$ depending only on K such that, for any $\beta \ge \beta_K$ and $i \in \{1, \ldots, 3g-3+n\}$, we have

$$\left|\frac{\tau_{i}(\sigma^{\beta})}{\beta} - \tau_{i}(\mathbb{G})\right| \leqslant \frac{c_{\mathsf{K}}}{\beta}.$$
(5.9)

Corollary 5.7. *In any fixed seamed pants decomposition, for* $\sigma \in T_{\Sigma}$ *, we have*

$$\lim_{\beta \to \infty} \frac{\tau_{i}(\sigma^{\beta})}{\beta} = \tau_{i}(sp(\sigma)),$$
(5.10)

and the limit is uniform when σ belongs to an arbitrary compact of T_{Σ} .

Before starting the proof of Proposition 5.6, we recall the formulae which allows us for four-holed spheres and one-holed tori to express the change of Fenchel–Nielsen coordinates under a flip of the pair of pants decomposition. They can be found in [37, Theorems 1.i and 2.i], or deduced from [10, Sections 3.3 and 3.4].

Let X be a four-holed sphere and $\sigma \in \mathcal{T}_X(L_1, L_2, L_3, L_4)$. We place ourselves in the situation described in § 2.6.1. Namely, we fix a coordinate system, which in turn defines a simple closed curve γ separating X into a pair of pants having boundary components $(\partial_1 X, \partial_4 X, \gamma)$ and another pair of pants having boundary components $(\partial_1 X, \partial_4 X, \gamma)$ and another pair of pants having boundary components $(\partial_1 X, \partial_2 X, \delta)$ and another pair of pants having boundary components $(\partial_1 X, \partial_2 X, \delta)$ and another pair of pants having boundary components $(\partial_1 X, \partial_2 X, \delta)$ and another pair of pants having boundary components $(\partial_1 X, \partial_2 X, \delta)$ and another pair of pants having boundary components $(\partial_1 X, \partial_2 X, \delta)$ and another pair of pants having boundary components $(\partial_1 X, \partial_2 X, \delta)$ and another pair of pants having boundary components $(\partial_1 X, \partial_2 X, \delta)$ and another pair of pants having boundary components $(\partial_1 X, \partial_2 X, \delta)$ and another pair of pants having boundary components $(\partial_1 X, \partial_2 X, \delta)$ and another pair of pants having boundary components $(\partial_1 X, \partial_2 X, \delta)$ and another pair of pants having boundary components $(\partial_1 X, \partial_2 X, \delta)$ and another pair of pants having boundary components $(\partial_1 X, \partial_2 X, \delta)$ and another pair of pants having boundary components $(\partial_1 X, \partial_2 X, \delta)$ and another pair of pants having boundary components $(\partial_1 X, \partial_2 X, \delta)$ and another pair of pants having boundary components $(\partial_1 X, \partial_2 X, \delta)$ and $(\partial_1 X, \partial_2 X, \delta)$ and $(\partial_1 X, \partial_2 X, \delta)$ and $(\partial_1 X, \partial_2 X, \delta)$ and the parts having boundary components $(\partial_1 X, \partial_2 X, \delta)$ and $(\partial_1 X, \partial_2 X, \delta)$

$$C_{i,j}(\ell) = \cosh^2(\frac{\ell}{2}) + \cosh^2(\frac{L_i}{2}) + \cosh^2(\frac{L_j}{2}) + 2\cosh(\frac{L_i}{2})\cosh(\frac{L_j}{2})\cosh(\frac{\ell}{2}) - 1,$$
(5.11)

the length of δ is then given in terms of Fenchel–Nielsen coordinates by

$$\begin{aligned} \cosh\left(\frac{\ell'(\ell,\tau)}{2}\right)\sinh^{2}\left(\frac{\ell}{2}\right) \\ &= \cosh(\frac{L_{1}}{2})\cosh(\frac{L_{2}}{2}) + \cosh(\frac{L_{3}}{2})\cosh(\frac{L_{4}}{2}) + \cosh(\frac{\ell}{2})\left(\cosh(\frac{L_{1}}{2})\cosh(\frac{L_{3}}{2}) + \cosh(\frac{L_{2}}{2})\cosh(\frac{L_{4}}{2})\right) \quad (5.12) \\ &+ \cosh(\tau)\sqrt{C_{1,4}(\ell)C_{2,3}(\ell)}, \end{aligned}$$

while the length of η is $\ell''(\ell, \tau) = \ell'(\ell, \tau + \ell)$.

Likewise, if T is a one-holed torus, $\sigma \in T_T(L)$, and we are in the situation described in § 2.6.2, the length of δ is given by

$$\cosh\left(\frac{\ell'(\ell,\tau)}{2}\right) = \frac{\cosh(\frac{\tau}{2})}{\sinh(\frac{\ell}{2})} \sqrt{\frac{\cosh(\frac{L}{2}) + \cosh(\ell)}{2}},\tag{5.13}$$

while the length of η is $\ell''(\ell, \tau) = \ell'(\ell, \tau + \ell)$.

Proof of Proposition 5.6. Fix σ in a compact K of \mathfrak{T}_{Σ} , and denote $\mathbb{G} = \operatorname{sp}(\sigma)$. We use repeatedly Proposition 5.4, which implies that for any simple closed curve ν chosen in a fixed finite subset of S_{Σ} , we have

$$\ell_{\mathbb{G}}(\nu) \leqslant \frac{\ell_{\sigma^{\beta}}(\nu)}{\beta} \leqslant \ell_{\mathbb{G}}(\nu) + \frac{c_{\mathsf{K}}}{\beta}$$

for any $\beta \ge \beta_K$ and some constant $c_K > 0$ depending only on the compact K and this finite set. In what follows, c_K, c'_K, \ldots denote positive constants depending on K and whose value may change from line to line. We denote lengths and twists with a superscript β to refer to the hyperbolic quantities measured with respect to σ^{β} , while lengths and twists without superscripts denote the combinatorial quantities measured with respect to G.

As the twist parameters τ_i^{β} and τ_i are computed locally in each piece Σ_i of type (0,4) or (1,1) defined by the seamed pants decomposition as in Section 2.6, we can restrict our attention to each piece separately. On Σ_i we have the curve δ_i defined by the seamed pants decomposition, and for every $k \in \mathbb{Z}$ we consider the curve obtained as the image of δ_i after k Dehn twists along γ_i . Denote by $\ell_i^{(k),\beta}$ its hyperbolic length with respect to σ^{β} , and by $\ell_i^{(k)}$ its combinatorial length with respect to \mathbb{G} . Compared to the notation of Section 2.6,

$$\ell_{i}^{\prime\beta} = \ell_{i}^{(0),\beta}, \quad \ell_{i}^{\prime\prime\beta} = \ell_{i}^{(1),\beta}, \qquad \ell_{i}^{\prime} = \ell_{i}^{(0)}, \quad \ell_{i}^{\prime\prime} = \ell_{i}^{(1)}.$$

As we work with a single piece at a time, we in fact often omit the subscript i. Suppose $\Sigma_i = X$ is a four-holed sphere. With the labelling matching the one described in § 2.6.1, we denote $L_i^\beta = \ell_{\sigma^\beta}(\partial_i X)$ for $i \in \{1, 2, 3, 4\}$. According to Lemma 2.43, we have

$$\ell^{(k)} = \max\left\{L_1 + L_3 - \ell, L_2 + L_4 - \ell, 2|\tau + k\ell| + M_{1,4}(\ell) + M_{2,3}(\ell)\right\}$$
(5.14)

where $M_{i,j}(\ell) = \max\left\{0, L_i - \ell, L_j - \ell, \frac{L_i + L_j - \ell}{2}\right\}$. The open sets

$$\mathbf{U}^{(k)} = \left\{ \mathbb{G} \in \mathfrak{T}_{\Sigma}^{\text{comb}} \mid 2\ell < \tau + k\ell < 4\ell \right\}, \qquad k \in \mathbb{Z}.$$
(5.15)

cover the compact K, so we can select finitely many indices $k_1, \ldots, k_N \in \mathbb{Z}$ to cover K. If $\mathbb{G} \in U^{(k)} \cap K$ for some $k \in \{k_1, \ldots, k_N\}$, the maximum in (5.14) is given by the last argument, that is

$$\ell^{(k)} = 2(\tau + k\ell) + M_{1,4}(\ell) + M_{2,3}(\ell)$$

which we see as an expression of $\tau + k\ell$ in terms of other lengths. We would like to compare it, when β is large, to the expression of $\tau^{\beta} + k\ell^{\beta}$ which we can access via Equation (5.12), namely

$$\cosh\left(\tau^{\beta} + k\ell^{\beta}\right) = \left[\cosh\left(\frac{\ell^{(k),\beta}}{2}\right)\sinh^{2}\left(\frac{\ell^{\beta}}{2}\right) - \cosh\left(\frac{L_{1}^{\beta}}{2}\right)\cosh\left(\frac{L_{2}^{\beta}}{2}\right) - \cosh\left(\frac{L_{3}^{\beta}}{2}\right)\cosh\left(\frac{L_{4}^{\beta}}{2}\right)\right] - \cosh\left(\frac{\ell^{\beta}}{2}\right)\left(\cosh\left(\frac{L_{1}^{\beta}}{2}\right)\cosh\left(\frac{L_{3}^{\beta}}{2}\right) + \cosh\left(\frac{L_{2}^{\beta}}{2}\right)\cosh\left(\frac{L_{4}^{\beta}}{2}\right)\right)\right] C_{1,4}^{-1/2}(\ell^{\beta})C_{2,3}^{-1/2}(\ell^{\beta}).$$

$$(5.16)$$

To obtain an upper bound for the right-hand side of (5.16), we can ignore the negative terms and use as upper bound for the numerator

$$\cosh\left(\frac{\ell^{(k),\beta}}{2}\right)\sinh^2\left(\frac{\ell^{\beta}}{2}\right)\leqslant \frac{1}{4}\;e^{\frac{\ell^{(k),\beta}}{2}+\ell^{\beta}},$$

and as lower bound for the factors in the denominator

$$C_{i,j}(\ell^{\beta}) \geq \frac{1}{4} e^{\max\left\{\ell^{\beta}, L_{i}^{\beta}, L_{j}^{\beta}, \frac{L_{i}^{\beta}+L_{j}^{\beta}+\ell^{\beta}}{2}\right\}} \geq \frac{1}{4} e^{\max\left\{L_{i}^{\beta}, L_{j}^{\beta}, \frac{L_{i}^{\beta}+L_{j}^{\beta}+\ell^{\beta}}{2}\right\}}.$$

Combined with $\operatorname{arcosh}(x) \leq \ln(2x)$ for $x \geq 1$, this results in

$$\begin{split} \frac{|\tau^{\beta} + k\ell^{\beta}|}{\beta} &\leqslant \frac{1}{\beta} \bigg(\ln 2 + \frac{\ell^{(k),\beta}}{2} + \ell^{\beta} - \frac{1}{2} \max\left\{ L_{1}^{\beta}, L_{4}^{\beta}, \frac{L_{1}^{\beta} + L_{4}^{\beta} + \ell^{\beta}}{2} \right\} - \frac{1}{2} \max\left\{ L_{2}^{\beta}, L_{3}^{\beta}, \frac{L_{2}^{\beta} + L_{3}^{\beta} + \ell^{\beta}}{2} \right\} \bigg) \\ &\leqslant \frac{\ell^{(k)}}{2} + \ell - \frac{1}{2} \max\left\{ L_{1}, L_{4}, \frac{L_{1} + L_{4} + \ell}{2} \right\} - \frac{1}{2} \max\left\{ L_{2}, L_{3}, \frac{L_{2} + L_{3} + \ell}{2} \right\} + \frac{c_{K}'}{\beta} \\ &\leqslant \frac{\ell^{(k)}}{2} - \frac{1}{2} \big(M_{1,4}(\ell) + M_{2,3}(\ell) \big) + \frac{c_{K}'}{\beta} \end{split}$$

and thus

$$\frac{|\tau^\beta+k\ell^\beta|}{\beta}\leqslant \tau+k\ell+\frac{c_K'}{\beta}.$$

We now look for a bound from below for (5.16). We first observe that by Equation (5.3), we have $\ell^{\beta} \ge \ell$ for $\beta \ge 1$. Since on the compact K, ℓ is bounded from below by $\varepsilon_{K} > 0$, we deduce that

$$\sinh^2\left(\frac{\ell^{\beta}}{2}\right) \geqslant \mathfrak{m}_{\mathsf{K}} e^{\ell^{\beta}} \qquad \text{with} \qquad \mathfrak{m}_{\mathsf{K}} = \frac{(1-e^{-\epsilon_{\mathsf{K}}})^2}{4} > 0.$$

This leads to a (rather crude) lower bound for the numerator of the right-hand side of (5.16)

$$\cosh\left(\frac{\ell^{(k),\beta}}{2}\right)\sinh^{2}\left(\frac{\ell^{\beta}}{2}\right) - \cosh\left(\frac{L_{1}^{\beta}}{2}\right)\cosh\left(\frac{L_{2}^{\beta}}{2}\right) - \cosh\left(\frac{L_{3}^{\beta}}{2}\right)\cosh\left(\frac{L_{4}^{\beta}}{2}\right) \\
- \cosh\left(\frac{\ell^{\beta}}{2}\right)\left(\cosh\left(\frac{L_{1}^{\beta}}{2}\right)\cosh\left(\frac{L_{3}^{\beta}}{2}\right) + \cosh\left(\frac{L_{2}^{\beta}}{2}\right)\cosh\left(\frac{L_{4}^{\beta}}{2}\right)\right) \\
\geqslant \frac{m_{K}}{2}e^{\frac{\ell^{(k),\beta}}{2} + \ell^{\beta}} - e^{\frac{L_{1}^{\beta} + L_{2}^{\beta}}{2}} - e^{\frac{L_{3}^{\beta} + L_{4}^{\beta}}{2}} - e^{\frac{\ell^{\beta} + L_{1}^{\beta} + L_{3}^{\beta}}{2}} - e^{\frac{\ell^{\beta} + L_{2}^{\beta} + L_{4}^{\beta}}{2}} \\
\geqslant \frac{m_{K}}{4}e^{\frac{\ell^{(k),\beta}}{2} + \ell^{\beta}} + \left(\frac{m_{K}}{4}e^{\frac{\ell^{(k),\beta}}{2} + \ell^{\beta}} - 4e^{\frac{1}{2}\max\{L_{1}^{\beta} + L_{2}^{\beta}, L_{3}^{\beta} + L_{4}^{\beta}, L_{1}^{\beta} + L_{3}^{\beta} + \ell^{\beta}, L_{2}^{\beta} + L_{4}^{\beta} + \ell^{\beta}\}}\right).$$
(5.17)

We split the first term in order to exhibit positivity of our lower bound – which is therefore not useless. Indeed, using Proposition 5.4 and Equation (5.14) on $U^{(k)} \cap K$ we get

$$\frac{\ell^{(\kappa),\beta}}{2} + \ell^{\beta} \ge \max\left\{L_{1}^{\beta} + L_{2}^{\beta}, L_{3}^{\beta} + L_{4}^{\beta}, L_{1}^{\beta} + L_{3}^{\beta} + \ell^{\beta}, L_{2}^{\beta} + L_{4}^{\beta} + \ell^{\beta}\right\} + \beta c_{\kappa}^{\prime\prime}$$

for some constant $c_K'' > 0$. Consequently, there exists $\beta_K \ge 1$ such that for any $\beta \ge \beta_K$ the expression inside the bracket in (5.17) is positive, and can be ignored in the lower bound. To obtain an upper bound for the denominator of (5.16), we write

$$C_{i,j}(\ell^{\beta}) \leqslant 5 e^{\max\left\{\ell^{\beta}, L_{i}^{\beta}, L_{j}^{\beta}, \frac{L_{i}^{\beta}+L_{j}^{\beta}+\ell^{\beta}}{2}\right\}}.$$

Combined with $\operatorname{arccosh}(x) \ge \ln x$, this implies for $\beta \ge \beta_K$

$$\begin{aligned} \frac{|\tau^{\beta} + k\ell^{\beta}|}{\beta} &\geq \frac{1}{\beta} \bigg(\ln\left(\frac{m_{K}}{20}\right) + \frac{1}{2}\ell^{(K),\beta} - \frac{1}{2}\max\left\{\ell^{\beta}, L_{1}^{\beta}, L_{4}^{\beta}, \frac{L_{1}^{\beta} + L_{4}^{\beta} + \ell^{\beta}}{2}\right\} - \frac{1}{2}\max\left\{\ell^{\beta}, L_{2}^{\beta}, L_{3}^{\beta}, \frac{L_{2}^{\beta} + L_{3}^{\beta} + \ell^{\beta}}{2}\right\} \bigg) \\ &\geq \tau + k\ell - \frac{c_{K}^{\prime\prime\prime}}{\beta} \end{aligned}$$

using the arguments we already used for the upper bound. We deduce that on $U^{(k)} \cap K$ and for $\beta \ge \beta_K$

$$\left|\frac{|\tau^{\beta} + k\ell^{\beta}|}{\beta} - (\tau + k\ell)\right| \leqslant \frac{c_{\kappa}}{\beta}.$$
(5.18)

A similar argument shows that on $U^{(k)} \cap K$ and for $\beta \ge \beta_K$, we have

$$\left|\frac{|\tau^{\beta} + (k+1)\ell^{\beta}|}{\beta} - (\tau + (k+1)\ell)\right| \leqslant \frac{c_{\kappa}}{\beta},\tag{5.19}$$

for perhaps larger constants β_{K} , $c_{K} > 0$. We can now conclude by using (5.18)–(5.19) to estimate

$$\frac{\tau^{\beta}}{\beta} = \frac{\beta}{2\ell^{\beta}} \left(\left| \frac{\tau^{\beta} + (k+1)\ell^{\beta}}{\beta} \right|^{2} - \left| \frac{\tau^{\beta} + k\ell^{\beta}}{\beta} \right|^{2} \right) - \frac{(2k+1)\ell^{\beta}}{2\beta}$$

Since $\ell^{\beta} \ge \varepsilon_{K} > 0$ on the compact K, we arrive on $U^{(k)} \cap K$ and for $\beta \ge \beta_{K}$ at the inequality

$$\left|\frac{\tau^{\beta}}{\beta} - \tau\right| \leqslant \frac{c_{\mathsf{K}}}{\beta},\tag{5.20}$$

for perhaps larger constants β_K , $c_K > 0$. These arguments were done for a fixed $k \in \{k_1, ..., k_N\}$, and as $(U^{(k_i)})_{i=1}^N$ cover K, we can find constants β_K and c_K such that the same estimate (5.20) holds uniformly over K for $\beta \ge \beta_K$.

A similar argument can be carried out for Σ_i being a one-holed torus, in the situation described in § 2.6.2. Instead of (5.16) we should estimate $\tau^{\beta} + k\ell^{\beta}$ via the formula

$$\cosh\left(\frac{\tau^{\beta}+k\ell^{\beta}}{2}\right) = \cosh\left(\frac{\ell^{(k),\beta}}{2}\right) \sinh\left(\frac{\ell^{\beta}}{2}\right) \sqrt{\frac{2}{\cosh\left(\frac{L^{\beta}}{2}\right) + \cosh(\ell^{\beta})}}$$

coming from (5.13), and compare it to the expression of $\tau + k\ell$ deduced from Lemma 2.45. As the estimates in this case are much easier and without surprise, we omit them.

We conclude that the desired estimate (5.9) is valid for any fixed $i \in \{1, ..., 3g - 3 + n\}$, and since this set is finite we can choose constants β_K and c_K independently of i: in the end, they only depend on the fixed seamed pants decomposition and on the compact K.

5.2 Application: piecewise linear structure

The combinatorial Fenchel–Nielsen coordinates on $\mathcal{T}_{\Sigma}^{\text{comb}}$ described in Theorem 2.41 depend on a seamed pants decomposition $(\mathcal{P}, \mathcal{S})$. Notice that changing \mathcal{S} amounts to changing $\tau_i(\mathbb{G}) \mapsto \tau_i(\mathbb{G}) + k_i \ell_i(\mathbb{G})$ by some $k_i \in \mathbb{Z}$, while $\ell_i(\mathbb{G})$ remains unchanged. The most interesting case occurs when we change the pair of pants decomposition \mathcal{P} . From [25], these changes are generated by local changes in four-holed spheres and one-holed tori, as presented by Okai in [37] for the hyperbolic case, where the corresponding change of Fenchel–Nielsen coordinate can be found. We now report these hyperbolic formulae ⁹.

FOUR-HOLED SPHERE. For a four-holed sphere X , consider the hyperbolic Fenchel–Nielsen coordinates $(\ell, \tau) \in T_X(L_1, L_2, L_3, L_4)$ relative to the system of curves $(\mathcal{P}, \mathcal{S})$ of Figure 25. Then the change of seamed pants decomposition to $(\mathcal{P}', \mathcal{S}')$ is given by Equation (5.12) and

$$\cosh\left(\tau'(\ell,\tau)\right) = \left[\sinh^{2}\left(\frac{\ell'(\ell,\tau)}{2}\right)\cosh\left(\frac{\ell}{2}\right) - \cosh\left(\frac{L_{1}}{2}\right)\cosh\left(\frac{L_{4}}{2}\right) - \cosh\left(\frac{L_{2}}{2}\right)\cosh\left(\frac{L_{3}}{2}\right) - \cosh\left(\frac{\ell'(\ell,\tau)}{2}\right)\left(\cosh\left(\frac{L_{1}}{2}\right)\cosh\left(\frac{L_{3}}{2}\right) + \cosh\left(\frac{L_{2}}{2}\right)\cosh\left(\frac{L_{4}}{2}\right)\right)\right]C_{1,2}(\ell)^{-1/2}C_{3,4}(\ell)^{-1/2}$$
(5.21)

with $sgn(\tau') = -sgn(\tau)$ and $C_{i,j}(\ell)$ has been defined in (??).

⁹In [37, Theorem 2.ii] there is a misprint in the denominator of the formula giving τ' for the one-holed torus. More precisely, the first $\cosh(\frac{L}{2})$ should not be squared, and the second $\cosh(\frac{L}{2})$ should be replaced with $\cosh(\frac{L}{2}) - 1$. We report the correct formula (5.22) here.



Figure 25: Change in the coordinate system of Σ .



Figure 26: Change in the coordinate system of T.

ONE-HOLED TORUS. For a one-holed torus T, consider the global coordinates $(\ell, \tau) \in T_T(L)$ relative to $(\mathcal{P}, \mathcal{S})$ of Figure 26. Then the change of coordinate system to $(\mathcal{P}', \mathcal{S}')$ is given by by Equation (5.13) and

$$\cosh\left(\frac{\tau'(\ell,\tau)}{2}\right) = \cosh\left(\frac{\ell}{2}\right) \sqrt{\frac{\cosh^2\left(\frac{\tau}{2}\right)\left(\cosh\left(\frac{L}{2}\right) + \cosh\left(\ell\right)\right) - 2\sinh^2\left(\frac{\ell}{2}\right)}{\cosh^2\left(\frac{\tau}{2}\right)\left(\cosh\left(\frac{L}{2}\right) + \cosh\left(\ell\right)\right) + \sinh^2\left(\frac{\ell}{2}\right)\left(\cosh\left(\frac{L}{2}\right) - 1\right)}}$$
(5.22)

with $sgn(\tau') = -sgn(\tau)$.

Using the convergence under the rescaling flow of hyperbolic length and twist parameters to the combinatorial analogues, we can calculate a "tropicalisation" of Okai's formulae.

Proposition 5.8.

SPHERE WITH FOUR BOUNDARY COMPONENTS. For a four-holed sphere X, consider the global coordinates $(\ell, \tau) \in T_X^{\text{comb}}(L_1, L_2, L_3, L_4)$ relative to the system of curves $(\mathcal{P}, \mathcal{S})$ of Figure 25. Then the change of coordinate system to $(\mathcal{P}', \mathcal{S}')$ is given by Equation (2.29) and

$$|\tau'(\ell,\tau)| = \frac{1}{2} \left| 2|\tau| + \ell + M_{1,4}(\ell) + M_{2,3}(\ell) - \ell'(\ell,\tau) - M_{1,2}(\ell') - M_{3,4}(\ell') \right|$$
(5.23)

 $\textit{with } sgn(\tau') = - \, sgn(\tau). \textit{ We recall here that } M_{i,j}(\ell) = max \, \big\{0, L_i - \ell, L_j - \ell, \frac{L_i + L_j - \ell}{2} \big\}.$

TORUS WITH ONE BOUNDARY COMPONENT. For a one-holed torus T, consider the global coordinates $(\ell, \tau) \in \mathcal{T}_{T}^{\text{comb}}(L)$ relative to the system of curves $(\mathcal{P}, \mathcal{S})$ of Figure 26. Then the change of coordinate system to $(\mathcal{P}', \mathcal{S}')$ is given by Equation (2.31) and

$$|\tau'(\ell,\tau)| = \left|\ell - \left[\frac{L}{2} - \ell'(\ell,\tau)\right]_+\right|$$
(5.24)

with $\operatorname{sgn}(\tau') = -\operatorname{sgn}(\tau)$.



Figure 27: The graphs of $\ell'(\ell, \tau)$ and $\tau'(\ell, \tau)$, with L = 2.

Proof. The formulae follow from Equations (5.21)–(5.22) by direct computation, together with the convergence of length and twist parameters and using relations of the form

$$\forall a, b \in \mathbb{R}_+, \qquad \lim_{\beta \to \infty} \frac{1}{\beta} \ln \left(e^{\beta a} + e^{\beta b} \right) = \max\{a, b\}.$$

The reader can check that the limit of Equations (5.12) and (5.13) coincide with Equations (2.29) and (2.31). \Box

Corollary 5.9. Consider a bordered surface Σ of type (g, n) and fix two seamed pants decompositions $(\mathcal{P}, \mathcal{S})$ and $(\mathcal{P}', \mathcal{S}')$, defining combinatorial Fenchel–Nielsen coordinates $\Phi, \Phi': \mathbb{T}_{\Sigma}^{\text{comb}} \to (\mathbb{R}_+ \times \mathbb{R})^{3g-3+n}$. Then the change of coordinate $\Phi' \circ \Phi^{-1}$ between open subsets of $(\mathbb{R}_+ \times \mathbb{R})^{3g-3+n}$ is piecewise linear. In particular, the combinatorial Teichmüller space $\mathbb{T}_{\Sigma}^{\text{comb}}$ is endowed with a canonical piecewise linear structure.

Remark 5.10. It should be possible to write a proof of Proposition 5.8, and thus Corollary 5.9, using purely combinatorial-geometric arguments (without resorting to hyperbolic geometry). However, given the complexity of Formulae (5.23)–(5.24) it may be challenging to do so in every cell. The proof of the hyperbolic formulae by Okai rely on the description of the Teichmüller space via the SL(2, \mathbb{R}) character variety. Having a similar representation-theoretic approach to the combinatorial Teichmüller space could give a direct way to prove Proposition 5.8 and would be of independent interest.

Remark 5.11. The above transformations are not continuous on the whole of $\mathbb{R}_+ \times \mathbb{R}$. The locus of discontinuity actually identifies a subset of the non-admissible twists, *i.e.* the creation of saddle connections in the measured foliation perspective. For instance, the plot of ℓ' and τ' on $\mathbb{R}_+ \times \mathbb{R} \cong \mathfrak{T}_T^{\text{comb}}(L)$ is illustrated in Figure 27. Notice that along the line

$$\mathfrak{l} = \left[\frac{\mathrm{L}}{2}, +\infty\right) \times \{0\},\$$

the function ℓ' is identically zero, while τ' has a discontinuity. But \mathfrak{l} is not in the image of $\mathfrak{T}_{\mathsf{T}}^{\text{comb}}(\mathsf{L})$ under the map Φ of Theorem 2.41. Thus, having $\ell' = 0$ and τ' discontinuous along \mathfrak{l} is not contradictory.

5.3 Geometric recursion in the flow

The geometric recursion in the hyperbolic and in the combinatorial setting produces functions respectively on the moduli spaces \mathcal{M}_{Σ} and $\mathcal{M}_{\Sigma}^{comb}$. The main result of this section is that the flow in the $\beta \to \infty$ limit takes the hyperbolic GR to the combinatorial GR.

5.3.1 Rescaling initial data

Before considering the behaviour of geometric recursion amplitudes in the flow, we discuss the rescaling of initial data.

Definition 5.12. Let $(A_{\beta}, B_{\beta}, C_{\beta})_{\beta \ge 1}$ be a family of triples of measurable functions on \mathbb{R}^3_+ . We say it is *uniformly (strongly) admissible* if the constants in the (strong) admissibility can be chosen to be independent of $\beta \ge 1$.

We remark that here, following the remark of Section 4.3, the triple (A, B, C) is completed by a natural choice of D. We introduce the rescaling operator, acting on a function $\phi \colon \mathbb{R}^k_+ \to \mathbb{R}$ by

$$\rho_{\beta*}\phi(L) = \phi(\beta L). \tag{5.25}$$

We notice two basic properties of this rescaling. Firstly, limits of uniformly admissible rescaled initial data (if they exist) are automatically admissible.

Lemma 5.13. Let $(A_{\beta}, B_{\beta}, C_{\beta})_{\beta \ge 1}$ be initial data such that $\rho_{\beta*}(A_{\beta}, B_{\beta}, C_{\beta})$ is uniformly admissible and converges to a pointwise limit $(\hat{A}, \hat{B}, \hat{C})$. Then $(\hat{A}, \hat{B}, \hat{C})$ is admissible.

Proof. This is clear by taking the $\beta \to \infty$ limit in the inequalities specified by the uniform admissibility. \Box

Secondly, if we rescale initial data that do not depend on β , the limit (if it exists) must be supported on small pairs of pants.

Lemma 5.14. *If* (A, B, C) *is admissible, then* $\rho_{\beta*}(A, B, C)$ *is uniformly admissible, with same bounding constants. Besides, if* $\rho_{\beta*}(B, C)$ *has a pointwise limit* (\hat{B}, \hat{C}) *when* $\beta \to \infty$ *, then*

$$\ell > L_1 + L_2 \implies \hat{B}(L_1, L_2, \ell) = 0, \quad and \quad \ell + \ell' > L_1 \implies \hat{C}(L_1, \ell, \ell') = 0.$$
(5.26)

Proof. The bound on $\rho_{\beta*}A$ is clear. For the bound on B, we write

$$\sup_{L_{1},L_{2},\ell \ge \epsilon} |B(\beta L_{1},\beta L_{2},\beta \ell)| (1 + [\ell - L_{1} - L_{2}]_{+})^{s} \\ \leq \sup_{L_{1},L_{2},\ell \ge \beta \epsilon} |B(L_{1},L_{2},\ell)| (1 + [\ell - L_{1} - L_{2}]_{+})^{s} \left(\frac{1 + [(\ell - L_{1} - L_{2})/\beta]_{+}}{1 + [\ell - L_{1} - L_{2}]_{+}}\right)^{s}.$$

Since $\beta \ge 1$, we can bound the last expression by the supremum over $L_1, L_2, \ell \ge \epsilon$. We also observe that $t \mapsto \frac{1+[t/\beta]_+}{1+[t]_+}$ is equal to 1 for $t \le 0$ and is decreasing for t > 0, hence it is uniformly bounded by 1. Using the initial bound for B we get the desired bound for $\rho_{\beta*}B$. The bound for $\rho_{\beta*}C$ is proved similarly.

To establish the vanishing property for \hat{B} , we fix (L_1, L_2, ℓ) such that $\ell - (L_1 + L_2) \ge \epsilon$ and min $\{L_1, L_2, \ell\} \ge \epsilon > 0$ for some $\epsilon > 0$. Specialising the admissibility condition for B at s = 1 and rescaled lengths, we have

$$|\mathsf{B}(\beta \mathsf{L}_1, \beta \mathsf{L}_2, \beta \ell)| \leqslant \frac{\mathsf{M}_{\epsilon, 1}}{1 + \beta \epsilon}$$

Taking the limit $\beta \to \infty$ yields the claim. The vanishing property of \hat{C} is proved similarly.

Remark 5.15. If (A, B, C) are continuous functions on \mathbb{R}^3_+ , one easily checks that $(\rho_{\beta*}A, \rho_{\beta*}B, \rho_{\beta*}C)_{\beta \ge 1}$ forms an equicontinuous family. By Arzela–Ascoli theorem, for any fixed compact, it must admit uniformly converging subsequences, and the vanishing properties in Lemma 5.14 must hold for any limit point. Therefore, the assumption that $\rho_{\beta*}(A, B, C)$ converges is rather weak.

5.3.2 Rescaling geometric recursion amplitudes

Theorem 5.16. Let $(A_{\beta}, B_{\beta}, C_{\beta})_{\beta \ge 1}$ be initial data such that $\rho_{\beta*}(A_{\beta}, B_{\beta}, C_{\beta})$ is uniformly admissible and converges uniformly on any compact to a limit $(\hat{A}, \hat{B}, \hat{C})$. Let us denote by $\Omega_{\Sigma;\beta}$ the result of the $Mes(T_{\Sigma})$ -valued GR with initial data $(A_{\beta}, B_{\beta}, C_{\beta})$, and by $\hat{\Xi}_{\Sigma}$ the result of the $Mes(T_{\Sigma}^{comb})$ -valued GR with initial data $(\hat{A}, \hat{B}, \hat{C})$. We have for any bordered surface Σ and $\sigma \in T_{\Sigma}$

$$\lim_{\beta \to \infty} \Omega_{\Sigma;\beta}(\sigma^{\beta}) = \widehat{\Xi}_{\Sigma}(\operatorname{sp}(\sigma))$$
(5.27)

and the convergence is uniform for σ in any compact of T_{Σ} . Besides, there exists $t \ge 0$ depending only on the topology of Σ , such that, for any $\varepsilon > 0$ there exists $M_{\varepsilon} > 0$ for which we have, for any $\beta \ge 1$, any $\sigma \in T_{\Sigma}$ such that $sys_{sp(\sigma)} \ge \varepsilon$,

$$\left|\Omega_{\Sigma;\beta}(\sigma^{\beta})\right| \leqslant M_{\varepsilon} \prod_{b \in \pi_{0}(\partial \Sigma)} \left(1 + \ell_{sp(\sigma)}(b)\right)^{t}$$
(5.28)

and the same inequality holds for the limit $\widehat{\Xi}_{\Sigma}(sp(\sigma))$.

Remark 5.17. If we specialise $(A_{\beta}, B_{\beta}, C_{\beta})$ to (A^{M}, B^{M}, C^{M}) appearing in Theorem 4.15, this gives another proof of the combinatorial Mirzakhani–McShane of Theorem 4.16 or its equivalent form Corollary 4.17. Indeed, the hyperbolic Mirzakhani–McShane identities immediately show that $\Omega_{\Sigma}^{M}(\sigma^{\beta}) = 1$ for any $\sigma \in \mathcal{T}_{\Sigma}$ and $\beta \ge 1$, and the convergence

$$\lim_{\beta \to \infty} \frac{\mathsf{F}(\beta x)}{\beta} = \lim_{\beta \to \infty} \frac{2\ln(1 + e^{\beta x/2})}{\beta} = [x]_+,$$

which is uniform on any compact of $\{-\infty\} \cup \mathbb{R}$, implies that

$$\lim_{\beta\to\infty}\rho_{\beta*}(A^M,B^M,C^M)=(A^K,B^K,C^K)$$

uniformly on any compact of \mathbb{R}^3_+ . Thus, the $\beta \to \infty$ limit of $\Omega^M_{\Sigma}(\sigma^{\beta})$ – which must be the constant function 1 – converges to the function $\Xi^K_{\Sigma}(sp(\sigma))$ satisfying the geometric recursion for the initial data (A^K, B^K, C^K) .

Proof of Theorem 5.16. It is enough to prove the result for connected surfaces, and we proceed by induction on the Euler characteristic. Throughout the proof, we denote $\mathbb{G} = \operatorname{sp}(\sigma)$. We recall that for any boundary component b, we have $\ell_{\sigma^{\beta}}(b) = \beta \ell_{\mathbb{G}}(b) = \beta \ell_{\sigma}(b)$. For a pair of pants P, we have

$$\Omega_{\mathsf{P};\beta}(\sigma^{\beta}) = \mathsf{A}_{\beta}(\vec{\ell}_{\sigma^{\beta}}(\partial\mathsf{P})) = \mathsf{A}_{\beta}(\beta\vec{\ell}_{\sigma}(\partial\mathsf{P})) = \mathsf{A}_{\beta}(\beta\vec{\ell}_{\mathbb{G}}(\partial\mathsf{P})),$$

which by assumption converges uniformly on any compact to $\hat{A}(\vec{\ell}_{\mathbb{G}}(\partial P)) = \widehat{\Xi}_{P}(\mathbb{G})$, and is bounded by a constant independent of β on any ϵ -thick part of \mathcal{T}_{P}^{comb} .

For a torus with one boundary component T, we have by the convention in Section 4.3

$$\Omega_{\mathsf{T};\beta}(\sigma^{\beta}) = \sum_{\gamma \in \mathsf{S}^{\circ}_{\mathsf{T}}} C_{\beta} \big(\beta \ell_{\mathbb{G}}(\partial \mathsf{T}), \ell_{\sigma^{\beta}}(\gamma), \ell_{\sigma^{\beta}}(\gamma) \big).$$
(5.29)

Let K be a compact subset of $\mathfrak{T}_{\Sigma}^{comb}$ and ε a lower bound of the systole on K. Let $\gamma \in S_T^{\circ}$ and $\mathbb{G} \in K$. Since the function $\ell_{\mathbb{G}}(\gamma)$ is bounded from below and from above on K, Proposition 5.5 implies that $\beta^{-1}\ell_{\sigma^{\beta}}(\gamma)$ converges to $\ell_{\mathbb{G}}(\gamma)$ uniformly for $\sigma \in sp^{-1}(K)$. By uniform convergence of $\rho_{\beta*}C_{\beta}$ on that compact, we deduce that

$$\lim_{\beta \to \infty} C_{\beta} \big(\beta \ell_{\mathbb{G}}(\partial T), \ell_{\sigma^{\beta}}(\gamma), \ell_{\sigma^{\beta}}(\gamma) \big) = \hat{C} \big(\ell_{\mathbb{G}}(T), \ell_{\mathbb{G}}(\gamma), \ell_{\mathbb{G}}(\gamma) \big)$$

uniformly for $\sigma \in K$. Next, we would like to bound each term in (5.29) by a summable (over γ) quantity depending only on K and not on β . If this holds, we can conclude by the (Banach-valued) dominated convergence theorem that $\Omega_{T;\beta}(\sigma^{\beta})$ converges to $\widehat{\Xi}_{T}(\mathbb{G})$ when $\beta \to \infty$, uniformly for $\sigma \in K$.

To prove the bound, we notice that by uniform admissibility we have for any s > 0

$$\left|C_{\beta}\left(\beta\ell_{\mathbb{G}}(\partial T),\ell_{\sigma^{\beta}}(\gamma),\ell_{\sigma^{\beta}}(\gamma)\right)\right| \leqslant \frac{M_{\epsilon,s}}{\left(1+\left[2\beta^{-1}\ell_{\sigma^{\beta}}(\gamma)-\ell_{\mathbb{G}}(\partial T)\right]_{+}\right)^{s}} \leqslant \frac{M_{\epsilon,s}}{\left(1+\left[2\ell_{\mathbb{G}}(\gamma)-\ell_{\mathbb{G}}(\partial T)\right]_{+}\right)^{s}},$$

where the second inequality used the lower bound $\beta \ell_{\mathbb{G}}(\gamma) \leq \ell_{\sigma^{\beta}}(\gamma)$ of Proposition 5.4. Since the number of small pairs of pants is bounded by the number of oriented edges, there are at most 6 curves $\gamma \in S^{\circ}_{\mathsf{T}}$ for which the denominator is equal to 1. Hence

$$\sum_{\gamma \in S^{\circ}_{\mathsf{T}}} \frac{1}{\left(1 + \left[2\ell_{\mathbb{G}}(\gamma) - \ell_{\mathbb{G}}(\mathfrak{d}\mathsf{T})\right]_{+}\right)^{s}} \leqslant 6 + \sum_{\mathsf{L} > \ell_{\mathsf{G}}(\mathfrak{d}\mathsf{T})/2} \frac{\#\{\gamma \in S^{\circ}_{\mathsf{T}} \mid \ell_{\mathsf{G}}(\gamma) \leqslant \mathsf{L}+1\}}{\left(1 + 2\mathsf{L} - \ell_{\mathsf{G}}(\mathfrak{d}\mathsf{T})\right)^{s}}.$$

Invoking Proposition 6.3, the numerator is bounded by $m_{\varepsilon}(L+1)^2$ for some constant m_{ε} depending on ε only. Therefore

$$\left|\Omega_{\mathsf{T};\beta}(\sigma^{\beta})\right| \leqslant \mathsf{M}_{\varepsilon,s} \bigg(6 + \mathfrak{m}_{\varepsilon} \sum_{L \geqslant 0} \frac{(1 + \ell_{\mathbb{G}}(\partial\mathsf{T})/2 + L)^2}{\left(1 + 2L\right)^s} \bigg)$$

By choosing s = 4 we find that $|\Omega_{T;\beta}(\sigma^{\beta})|$ is bounded by a polynomial of degree 2 in $\ell_{\mathbb{G}}(\partial T)$, whose coefficients are independent of β but may depend on ϵ . This proves the theorem for $\Sigma = T$.

Now let Σ be a connected bordered surface with $\chi_{\Sigma} < -1$, and assume the thesis for all Σ' such that $\chi_{\Sigma'} > \chi_{\Sigma}$. Let K be a compact subset of $\mathcal{T}_{\Sigma}^{\text{comb}}$ and ε a lower bound on the systole on K. The geometric recursion gives

$$\Omega_{\Sigma;\beta}(\sigma^{\beta}) = \sum_{[P]\in\mathfrak{P}_{\Sigma}} X_{P;\beta}(\vec{\ell}_{\sigma^{\beta}}(\partial P)) \Omega_{\Sigma-P;\beta}(\sigma^{\beta}|_{\Sigma-P}).$$

For each $[P] \in \mathfrak{P}_{\Sigma}$, we can repeat the previous arguments to show that $X_P(\tilde{\ell}_{\sigma^\beta}(\partial P))$ is converging to $\hat{X}_{P;\beta}(\tilde{\ell}_{\mathbb{G}}(\partial P))$ uniformly for $\mathbb{G} \in K$. If the initial data were the one of Mirzakhani (4.22), the factor $\Omega_{\Sigma-P;\beta}$ would be the constant function 1. Then, we could finish the proof very similarly to the case of $\Sigma = T$, and this would prove a second time the combinatorial Mirzakhani–McShane identities (Theorem 4.16). However, in the general case, we only know by induction hypothesis that $\Omega_{\Sigma-P;\beta}(\tilde{\sigma}^{\beta})$ converges to $\widehat{\Xi}_{\Sigma-P}(\tilde{\mathbb{G}})$ uniformly for $\tilde{\mathbb{G}} = sp(\tilde{\sigma})$ in any compact of $\mathfrak{T}_{\Sigma-P}^{comb}$. To employ this information, we shall compare the two hyperbolic structures $\sigma^{\beta}|_{\Sigma-P}$ and $\tilde{\sigma}^{\beta} \coloneqq sp^{-1}(\beta \mathbb{G}|_{\Sigma-P})$, by means of their lengths functions. Since the combinatorial cutting does not decrease the systole, Proposition 5.4 on Σ and $\Sigma-P$ respectively imply for $\beta \ge \beta_{\varepsilon}$ and $\gamma \in S_{\Sigma-P}^{\circ}$

$$\begin{split} \ell_{\mathbb{G}}(\gamma) &\leqslant \frac{\ell_{\sigma^{\beta}}(\gamma)}{\beta} \leqslant \ell_{\mathbb{G}}(\gamma) \left(1 + \frac{\kappa_{\varepsilon}}{\beta}\right), \\ \ell_{\mathbb{G}|_{\Sigma-P}}(\gamma) &\leqslant \frac{\ell_{\tilde{\sigma}^{\beta}}(\gamma)}{\beta} \leqslant \ell_{\mathbb{G}|_{\Sigma-P}}(\gamma) \left(1 + \frac{\kappa_{\varepsilon}}{\beta}\right), \end{split}$$

where we can use for κ_{ϵ} , β_{ϵ} the constants provided by Proposition 5.4 for Σ . The combinatorial and hyperbolic cutting procedures are such that

$$\forall \gamma \in S^{\circ}_{\Sigma-P} \qquad \ell_{\mathbb{G}|_{\Sigma-P}}(\gamma) = \ell_{\mathbb{G}}(\gamma) \quad \text{and} \quad \ell_{\sigma^{\beta}|_{\Sigma-P}}(\gamma) = \ell_{\sigma^{\beta}}(\gamma).$$

Therefore

$$\forall \gamma \in S^{\circ}_{\Sigma-P} \qquad \left(1 + \frac{\kappa_{\varepsilon}}{\beta}\right)^{-1} \ell_{\tilde{\sigma}^{\beta}}(\gamma) \leqslant \ell_{\sigma^{\beta}|_{\Sigma-P}}(\gamma) \leqslant \ell_{\tilde{\sigma}^{\beta}}(\gamma) \left(1 + \frac{\kappa_{\varepsilon}}{\beta}\right).$$

Further, for any component of $\partial P \cap \Sigma^{\circ}$ and $\partial \Sigma \cap \partial(\Sigma - P)$, we have exactly $\ell_{\sigma^{\beta}|_{\Sigma-P}}(\gamma) = \ell_{\tilde{\sigma}^{\beta}}(\gamma)$, so that the above bounds extend to all $\gamma \in S_{\Sigma-P}$. From our description of the topology of combinatorial Teichmüller spaces (Theorem 2.20), we deduce that $\beta^{-1} \operatorname{sp}(\sigma^{\beta}|_{\Sigma-P})$ remains in a compact of $\mathcal{T}_{\Sigma-P}^{\operatorname{comb}}$ independent of β when $\mathbb{G} \in K$, and that it converges to $\mathbb{G}|_{\Sigma-P}$ uniformly for $\mathbb{G} \in K$. By the induction hypothesis, we have

$$\lim_{\beta \to \infty} \Omega_{\Sigma - P; \beta}(\sigma^{\beta}|_{\Sigma - P}) = \widehat{\Xi}_{\Sigma - P}(\mathbb{G}|_{\Sigma - P})$$

and the convergence is uniform for $\mathbb{G} \in K$. Supplemented with a summable (over $[P] \in \mathcal{P}_{\Sigma}$) bound whose derivation is similar to the case $\Sigma = T$ and therefore omitted, this proves the theorem for Σ , and thus in full generality by induction.

5.4 Topological recursion in the flow

By Theorem 5.2, we know that the Weil–Petersson measure flows in a pointwise sense, in the $\beta \rightarrow \infty$ limit ,to the Kontsevich measure. This implies that the Weil–Petersson volumes flow (by rescaling the boundary lengths by β) to the Kontsevich volumes. More generally, we are able to prove an analogue of Theorem 5.16 after integration over the moduli spaces.

Theorem 5.18. Let $(A_{\beta}, B_{\beta}, C_{\beta})_{\beta \ge 1}$ be initial data such that $\rho_{\beta*}(A_{\beta}, B_{\beta}, C_{\beta})$ is uniformly strongly admissible and converges uniformly on any subset of the form $(0, M]^3 \subset \mathbb{R}^3_+$ to a limit $(\hat{A}, \hat{B}, \hat{C})$. Then, $(\hat{A}, \hat{B}, \hat{C})$ is strongly admissible and

$$\lim_{\beta \to \infty} \frac{V\Omega_{g,n;\beta}(\beta L)}{\beta^{6g-6+2n}} = V\widehat{\Xi}_{g,n}(L),$$
(5.30)

with uniform convergence for L in any subset of the form $(0, M]^n \subset \mathbb{R}^n_+$.

Proof. We prove the result by induction on 2g-2+n > 0. For (g, n) = (0, 3), we have $V\Omega_{0,3;\beta}(\beta L) = A_{\beta}(\beta L)$, which converges uniformly on any compact of \mathbb{R}^3 intersected with \mathbb{R}^3_+ to $\hat{A}(L) = V\hat{\Xi}_{0,3}(L)$. For (g, n) = (1, 1) we have

$$\frac{V\Omega_{1,1;\beta}(\beta L_1)}{\beta^2} = \int_{\mathbb{R}_+} C_{\beta}(\beta L_1, \ell, \ell) \frac{\ell}{\beta} \frac{d\ell}{\beta} = \int_{\mathbb{R}_+} C_{\beta}(\beta L_1, \beta \ell, \beta \ell) \ell d\ell$$

Note $C_{\beta}(\beta L_1, \beta \ell, \beta \ell)$ converges uniformly on any $(0, M]^2$ to $\hat{C}(L_1, \ell, \ell)$. Strong uniform admissibility means we can bound the integrand by an integrable function independent of β . Moreover, the uniformity of the convergence around zero implies that we can exchange the integral and the limit, so that

$$\lim_{\beta \to \infty} \frac{V\Omega_{1,1;\beta}(\beta L_1)}{\beta^2} = \lim_{\beta \to \infty} \int_{\mathbb{R}_+} C_{\beta}(\beta L_1, \beta \ell, \beta \ell) \ell d\ell = \int_{\mathbb{R}_+} \hat{C}(L_1, \ell, \ell) \ell d\ell = V \widehat{\Xi}_{1,1}(L_1),$$

and the convergence is uniform on any (0, M]. This proves the two base cases.

The general argument follows along the same lines as the (g, n) = (1, 1) case. Assume the result for (g', n') such that 2g' - 2 + n' < 2g - 2 + n. The topological recursion for $V\Omega_{g,n;\beta}(\beta L)$ yields

$$\begin{split} \frac{V\Omega_{g,n;\beta}(\beta L_1,\ldots,\beta L_n)}{\beta^{6g-6+2n}} &= \sum_{m=2}^n \int_{\mathbb{R}_+} B_\beta(\beta L_1,\beta L_m,\ell) \frac{V\Omega_{g,n-1;\beta}(\ell,\beta L_2,\ldots,\widehat{\beta L_m},\ldots,\beta L_n)}{\beta^{6g-6+2(n-1)}} \frac{\ell}{\beta} \frac{d\ell}{\beta} \\ &\quad + \frac{1}{2} \int_{\mathbb{R}_+^2} C_\beta(\beta L_1,\ell,\ell') \bigg(\frac{V\Omega_{g-1,n+1;\beta}(\ell,\ell',\beta L_2,\ldots,\beta L_n)}{\beta^{6(g-1)-6+2(n+1)}} \\ &\quad + \sum_{\substack{J \sqcup J' = \{L_2,\ldots,L_m\}\\h+h' = g}} \frac{V\Omega_{h,1+\#J;\beta}(\ell,\beta J)}{\beta^{6h-6+2(1+\#J)}} \frac{V\Omega_{h',1+\#J';\beta}(\ell',\beta J')}{\beta^{6h'-6+2(1+\#J')}} \bigg) \frac{\ell\ell'}{\beta^2} \frac{d\ell\,d\ell'}{\beta^2}. \end{split}$$

Now rescaling the integration variables, we get

$$\begin{split} \frac{V\Omega_{g,n;\beta}(\beta L_1,\ldots,\beta L_n)}{\beta^{6g-6+2n}} &= \sum_{m=2}^n \int_{\mathbb{R}_+} B_\beta(\beta L_1,\beta L_m,\beta\ell) \frac{V\Omega_{g,n-1;\beta}(\beta\ell,\beta L_2,\ldots,\widehat{\beta L_m},\ldots,\beta L_n)}{\beta^{6g-6+2(n-1)}} \ell d\ell \\ &\quad + \frac{1}{2} \int_{\mathbb{R}_+^2} C_\beta(\beta L_1,\beta\ell,\beta\ell') \bigg(\frac{V\Omega_{g-1,n+1;\beta}(\beta\ell,\beta\ell',\beta L_2,\ldots,\beta L_n)}{\beta^{6(g-1)-6+2(n+1)}} \\ &\quad + \sum_{\substack{J \sqcup J' = \{L_2,\ldots,L_m\}\\h+h' = g}} \frac{V\Omega_{h,1+\#J;\beta}(\beta\ell,\beta J)}{\beta^{6h-6+2(1+\#J)}} \frac{V\Omega_{h',1+\#J';\beta}(\beta\ell',\beta J')}{\beta^{6h'-6+2(1+\#J')}} \bigg) \ell \ell' d\ell \, d\ell'. \end{split}$$

By our induction assumption, all of the $\beta^{-(6g'-6+2n')}V\Omega_{g',n';\beta}$'s appearing on the right-hand side of the equation converge uniformly on any $(0, M]^n$. For $X_\beta \in \{B_\beta, C_\beta\}$, we assumed that $X_\beta(\beta L)$ converges to $\hat{X}(L)$ uniformly on any $(0, M]^3$ and moreover that $X_\beta(\beta L)$ is uniformly admissible. Therefore using the strong uniform admissibility we can bound the integrals around infinity by integrable functions independently of β and that can be uniformly chosen on compact sets of L. The uniformity in compacts around zero then implies that we can interchange the integral and the limit, which again by our induction assumption reproduces the topological recursion for $V\hat{\Xi}_{g,n}(L)$ uniformly on any $(0, M]^n$.

Corollary 5.19. If $\rho_{\beta*}(A, B, C)$ is uniformly strongly admissible and converges uniformly on any compact of \mathbb{R}^3 intersected with \mathbb{R}^3_+ to a limit $(\hat{A}, \hat{B}, \hat{C})$, then $(\hat{A}, \hat{B}, \hat{C})$ is strongly admissible and

$$\lim_{\beta \to \infty} \frac{V\Omega_{g,n}(\beta L)}{\beta^{6g-6+2n}} = V\widehat{\Xi}_{g,n}(L).$$
(5.31)

In particular, if $V\Omega_{g,n}(L)$ is a polynomial, then $V\Omega_{g,n}(L)$ is of degree at most 6g - 6 + 2n and $V\widehat{\Xi}_{g,n}(L)$ is the homogeneous component of degree 6g - 6 + 2n.

6 Enumerative geometry of multicurves

In [33], Mirzakhani defines a function on the hyperbolic moduli space with cusps that counts the number of multicurves with length bounded by some parameter $t \in \mathbb{R}_+$. Moreover, she calculates the integral of this function over the moduli space as a sum over stable graphs. She then considers the behaviour of this integral as the parameter tends to infinity, and in [34] she shows that the limiting value gives the Masur– Veech volume of the moduli space of quadratic differentials.

In this section, after explaining how to parametrise multicurves via embedded ribbon graphs, we briefly recall Mirzakhani's work and then give the natural extension to the combinatorial setting. We then describe twisted geometric recursion amplitudes, and by considering a special family of such functions we illustrate how to recover additive statistics of multicurves by taking Laplace transform. This gives an interpretation of the Masur–Veech polynomials in [1] as a Laplace transform of averages of additive statistics. Moreover, this suggest a new family of polynomials, calculated by topological recursion, that calculate averages of additive statistics of hyperbolic length.

6.1 Parametrising multicurves

Consider an open cell $\mathfrak{Z}_{\Sigma,G}$ on the combinatorial Teichmüller space $\mathfrak{T}_{\Sigma}^{comb}$, that is a trivalent ribbon graph G together with a marking f: $\Sigma \to |G|$. Assigning to each edge $e \in E_G$ the number of times a non-backtracking representative on G of $c \in M_{\Sigma}$ passes through it, we obtain a map

$$\mathfrak{m}_{\mathfrak{Z}_{\Sigma,G}}\colon \mathcal{M}_{\Sigma} \longrightarrow \mathbb{N}^{\mathsf{E}_{\mathsf{G}}}.\tag{6.1}$$

We show that in fact this map gives a parametrisation of M_{Σ} . Let us first introduce some notation to describe its image.

Definition 6.1. Given a trivalent ribbon graph G, a *corner* is an ordered triple $\Delta = (e, e', e'')$ where e, e', e'' are edges incident to a vertex in the cyclic order. Equivalently, a corner consists of a vertex v together with the choice of an incident edge e. We say that a corner belongs to a face $\mathfrak{f} \in F_G$ if e' and e'' are edges around that face. We denote by $\mathfrak{C}(\mathfrak{f})$ the set of corners belonging to \mathfrak{f} , and by \mathfrak{C}_G the set of all corners. If we have an assignment of real numbers $(x_e)_{e \in E_G}$ and $\Delta = (e, e', e'')$ is a corner, we set $x_\Delta = x_{e''} + x_{e'} - x_e$.

Lemma 6.2. The map $\mathfrak{m}_{\mathfrak{I}_{\Sigma,G}}$ is a bijection between M_{Σ} and the set

$$Z_{G} = \left\{ m \in \mathbb{N}^{\mathbb{E}_{G}} \mid \begin{array}{c} \forall \Delta \in \mathfrak{C}_{G}, \ \mathfrak{m}_{\Delta} \in 2\mathbb{N} \\ \forall \mathfrak{f} \in \mathbb{F}_{G}, \ \exists \Delta \in \mathfrak{C}(\mathfrak{f}) \ s.t. \ \mathfrak{m}_{\Delta} = 0 \end{array} \right\}.$$
(6.2)

Proof. As $\mathfrak{m}_{\mathfrak{I}_{\Sigma,G}}$ is additive under union, it is enough to prove that for any simple closed curve $\gamma \in S_{\Sigma}^{\circ}$, $\mathfrak{m}_{\mathfrak{I}_{\Sigma,G}}(\gamma) \in Z_{G}$ and there is a unique multicurve corresponding to each $\mathfrak{m} \in Z_{G}$.

For the first part, we decompose the geometric realisation |G| into strips S_e for each edge e and small triangular neighbourhoods \mathcal{N}_v of each vertex the vertices as in Figure 28a, and pullback this structure to Σ via f. If $\gamma \in S_{\Sigma}^{\circ}$ is a simple closed curve in Σ , we can isotope γ to a non-backtracking simple representative γ that has m_e parallel paths in the strip corresponding to $e \in E_G$. At each vertex v, it is possible to draw pairwise non-intersecting arcs connecting inside \mathcal{N}_v the endpoints of γ in $\partial \mathcal{N}_v$ in a non-backtracking way if and only if $m_{\Delta} \in 2\mathbb{N}$ for each corner Δ . When these conditions hold, there is in fact a unique way (up to isotopy relative to $\partial \mathcal{N}_v$) to draw such arcs as in Figure 28b. Namely, we can label the points in $\partial \mathcal{N}_v \cap \gamma \cap S_e$ by $p_{e,1}, \ldots, p_{e,m_e}$ following the cyclic order around v. Then, $p_{e,i}$ must be connected to

- $p_{e'',\mathfrak{m}_{e''}+1-\mathfrak{i}}$ for $1 \leq \mathfrak{i} \leq \frac{1}{2}\mathfrak{m}_{\Delta'}$
- $p_{e',\frac{1}{2}\mathfrak{m}_{\Delta'}+1-\mathfrak{i}}$ for $\frac{1}{2}\mathfrak{m}_{\Delta'} < \mathfrak{i} \leq \frac{1}{2}(\mathfrak{m}_{\Delta'}+\mathfrak{m}_{\Delta''}) = \mathfrak{m}_e$.



Figure 28: Decomposition of Σ into strips S_e around edges and triangular neighbourhoods \mathcal{N}_{ν} around vertices.

This proves that $\mathfrak{m}_{\mathfrak{Z}_{\Sigma,G}}$ is injective on S_{Σ}° and its image is included in $\{\mathfrak{m} \in \mathbb{N}^{E_{G}} \mid \mathfrak{m}_{\Delta} \in 2\mathbb{N}\}$.

Let $(e_i)_{i \in \mathbb{Z}/N\mathbb{Z}}$ be the sequence of edges around a face \mathfrak{f} , and Δ_i be the corner containing both e_i and e_{i+1} . Then, the $\frac{1}{2}\mathfrak{m}_{\Delta_i}$ arcs in $\gamma \cap \mathscr{S}_{e_i}$ which are closest to \mathfrak{f} are connected to the $\frac{1}{2}\mathfrak{m}_{\Delta_i}$ arcs in $\gamma \cap \mathscr{S}_{e_{i+1}}$ which are the closest to \mathfrak{f} . In particular, the $\frac{1}{2}\mathfrak{m}_{i} \{\mathfrak{m}_{\Delta_i}\}$ arcs which are (in each strip around \mathfrak{f}) the closest to \mathfrak{f} are connected and form loops, which are homotopic to the boundary component of Σ that \mathfrak{f} represents. By definition of S_{Σ}° , we must have $\mathfrak{min}_i \{\mathfrak{m}_{\Delta_i}\} = 0$. This proves that $\mathfrak{m}_{\mathfrak{I}_{\Sigma,G}}(S_{\Sigma}^{\circ}) \subset Z_G$ and the first part of the claim.

Conversely, if we are given $m \in Z_G$, we draw m_e parallel arcs in S_e , and connect them inside each \mathcal{N}_v in the unique non-intersecting and non-backtracking (as explained above) way. We obtain a collection of simple closed curves, none of them being homotopic to a boundary component of Σ .

As a consequence, we obtain a polynomial growth of the number of essential simple closed curve of bounded combinatorial length.

Proposition 6.3. Let Σ be a connected bordered surface of type (g, n) and $\varepsilon > 0$. For any $\mathbb{G} \in \mathbb{T}_{\Sigma}^{comb}$ such that $sys_{\mathbb{G}} \ge \varepsilon$, and any t > 0, we have

$$\#\left\{\gamma\in S_{\Sigma}^{\circ}\mid \ell_{\mathbb{G}}(\gamma)\leqslant t\right\}\leqslant \mathfrak{m}_{\varepsilon}\,t^{6g-6+2n},\tag{6.3}$$

for some positive constant $\mathfrak{m}_{\varepsilon}$ depending on ε and the topology $(\mathfrak{g},\mathfrak{n})$ of the surface.

Proof. Let $\gamma \in S_{\Sigma}^{\circ}$ and denote by $E_{\mathbb{G}}(\gamma)$ the number (with multiplicity) of edges γ passes through. We recall from the proof of Proposition 5.4 that for sys₆ $\geq \epsilon$ we have

$$\mathsf{E}_{\mathbb{G}}(\gamma) \leqslant \frac{(6g-6+3n)^2}{\varepsilon}\, \ell_{\mathbb{G}}(\gamma)$$

If the underlying graph G is trivalent, from this bound and Lemma 6.2 we deduce

$$\#\left\{ \gamma \in S_{\Sigma}^{\circ} \mid \ell_{\mathbb{G}}(\gamma) \leqslant t \right\} \leqslant \#\left\{ x \in \mathsf{Z}_{\mathsf{G}} \mid \sum_{e \in \mathsf{E}_{\mathsf{G}}} x_{e} \leqslant \frac{(6g - 6 + 3n)^{2}}{\varepsilon} t \right\}$$

and the claim follows, as the latter is the set of integer points with bounded L¹-norm in a polytope of dimension 6g - 6 + 2n. If G is not trivalent, we resolve it in an arbitrary way into a trivalent ribbon graph G' and the same argument works with G'.

Remark 6.4. A second proof of Proposition 6.3 can be obtained by invoking that a similar result holds on T_{Σ} with respect to hyperbolic lengths – [40, 33] for punctured surfaces, extended to bordered surfaces in [2, Section 7.2] – and flowing it to the combinatorial setting via Proposition 5.4.

6.2 Unit balls in measured foliations: the hyperbolic case

Let Σ be a connected bordered surface of type (g, n). In Section 6.1, we considered the space MF_{Σ}^{\star} parametrising measured foliations on Σ up to Whitehead equivalence, allowing all types of boundary behaviours. We consider now the subset $MF_{\Sigma} \subset MF_{\Sigma}^{\star}$ that contains only those measured foliations whose boundary is made up of a union of singular leaves (that is, points at the boundary are of parallel type, *cf.* Figure 7c and 7f). In other words, on each boundary component there is at least one singularity, and the singularities on the boundary are connected by singular leaves. For convenience, we include in MF_{Σ} the empty foliation. Notice that MF_{Σ} is disjoint from the image of $\mathbb{T}_{\Sigma}^{\text{comb}}$ in MF_{Σ}^{\star} .

The space MF_{Σ} has dimension 6g-6+2n and admits a canonical piecewise linear structure. It also admits an integral structure given by M_{Σ} , the set of multicurves in which components are not allowed to be homotopic to boundary components of Σ . We denote by μ_{Th} the Thurston measure on MF_{Σ} associated to this piecewise linear integral structure¹⁰. For an open set $A \subset MF_{\Sigma}$, we have

$$\mu_{Th}(A) = \lim_{t \to +\infty} \frac{\#(t \cdot A \cap M_{\Sigma})}{t^{6g-6+2n}}.$$

The Thurston measure allows one to define a counting of multicurves associated to any length function on MF_{Σ} . In fact, MF_{Σ} is a completion of the set of formal \mathbb{Q}_+ -linear combinations of simple closed curves.

In the hyperbolic context, for $\sigma \in \mathfrak{T}_{\Sigma}$ the hyperbolic length function $\ell_{\sigma}: M_{\Sigma} \to \mathbb{R}_+$ has a unique continuous extension $\ell_{\sigma}: MF_{\Sigma} \to \mathbb{R}_{\geq 0}$ that is compatible with the piecewise linear structure. Mirzakhani [33] introduces a function measuring the volume of the unit ball in MF_{Σ} with respect to ℓ_{σ} .

Definition 6.5. The function $\mathscr{B}_{\Sigma} : \mathscr{T}_{\Sigma} \to [0, +\infty]$ is defined by

$$\mathscr{B}_{\Sigma}(\sigma) = \mu_{\mathrm{Th}} \big(\{ \mathfrak{F} \in \mathrm{MF}_{\Sigma} \mid \ell_{\sigma}(\mathfrak{F}) \leqslant 1 \} \big). \tag{6.4}$$

It is manifestly Mod_{Σ}-invariant, hence descends to a function $\mathcal{B}_{q,n}$ on $\mathcal{M}_{q,n}$ for Σ of type (g, n).

By definition of the Thurston measure, this function describes the asymptotic growth of the number of multicurves of length $\leq t$, when $t \to \infty$, namely

$$\mathcal{B}_{\Sigma}(\sigma) = \lim_{t \to +\infty} \frac{\#\{c \in M_{\Sigma} \mid \ell_{\sigma}(c) \leq t\}}{t^{6g-6+2n}}.$$
(6.5)

The main properties of \mathscr{B}_{Σ} established by Mirzakhani for punctured surfaces – *i.e.* on $\mathcal{M}_{g,n}(0)$ – can easily be generalised to the case of bordered surfaces.

Theorem 6.6. [33, Proposition 3.2, Theorem 3.3] The function \mathcal{B}_{Σ} takes values in \mathbb{R}_+ , is continuous on \mathfrak{T}_{Σ} , and $\mathcal{B}_{q,n}$ is integrable on $\mathcal{M}_{q,n}(L)$ with respect to μ_{WP} .

We remark that finiteness of \mathscr{B}_{Σ} comes from the hyperbolic analogue – known since [40] for punctured surfaces and extended to bordered surfaces in [2, Theorem 7.2] – of Proposition 6.3.

More generally, the same properties hold for the counting function

$$\mathcal{N}_{\Sigma}(\sigma; t) = \#\{ c \in \mathcal{M}_{\Sigma} \mid \ell_{\sigma}(c) \leqslant t \},$$
(6.6)

and Mirzakhani computed in [33, Section 5] the integral of $\mathcal{N}_{g,n}(X;t)$ over $\mathcal{M}_{g,n}(L)$ as a sum over stable graphs. Here we present an equivalent version of her statement, in a form that suits better our next purpose.

 $^{^{10}}$ On MF_{Σ}, the normalisation by counting lattice points differs by a power of 2 from the normalisation obtained by taking the top power of the Thurston symplectic form.

Lemma 6.7. The integral of $\mathcal{N}_{g,n}(X;t)$ over $\mathcal{M}_{g,n}(L)$ is a polynomial in $t^2, L_1^2, \ldots, L_n^2$. Furthermore, its Laplace transform in the cutoff variable t is given by

$$\int_{\mathbb{R}_{+}} \left(\int_{\mathcal{M}_{g,n}(L)} \mathcal{N}_{g,n}(X;t) \, d\mu_{WP}(X) \right) e^{-st} dt$$

$$= \frac{1}{s} \sum_{\Gamma \in \mathfrak{G}_{g,n}} \frac{1}{\#\operatorname{Aut}(\Gamma)} \int_{\mathbb{R}_{+}^{\mathsf{E}_{\Gamma}}} \prod_{\nu \in V_{\Gamma}} V_{g(\nu),n(\nu)}^{WP} \left((\ell_{e})_{e \in \mathsf{E}_{\nu}}, (\mathsf{L}_{\lambda})_{\lambda \in \Lambda_{\nu}} \right) \prod_{e \in \mathsf{E}_{\Gamma}} \frac{\ell_{e} \, d\ell_{e}}{e^{s\ell_{e}} - 1}$$

$$(6.7)$$

and it is a polynomial of degree 6g - 5 + 2n in s^{-1} .

As a consequence, using the final value theorem for the Laplace transform, we have an expression of the integral of $\mathcal{B}_{q,n}$ over $\mathcal{M}_{q,n}(L)$ as a sum over stable graphs (see also [1, 17]).

Theorem 6.8. [33, Theorem 5.3] The integral of $\mathcal{B}_{g,n}$ is computed as

$$(6g - 6 + 2n)! \int_{\mathcal{M}_{g,n}(L)} \mathcal{B}_{g,n} d\mu_{WP}$$

$$= \sum_{\Gamma \in \mathcal{G}_{g,n}} \frac{1}{\# \operatorname{Aut}(\Gamma)} \int_{\mathbb{R}^{E_{\Gamma}}_{+}} \prod_{\nu \in V_{\Gamma}} V^{K}_{g(\nu),n(\nu)} ((\ell_{e})_{e \in E_{\nu}}, (0_{\lambda})_{\lambda \in \Lambda_{\nu}}) \prod_{e \in E_{\Gamma}} \frac{\ell_{e} d\ell_{e}}{e^{\ell_{e}} - 1}.$$
(6.8)

Notice the presence of Kontsevich volumes as vertices weights. This comes from the fact that the top coefficient of s^{-1} in equation (6.7) is given by extraction of the largest possible power of the l_e , and that the top degree of $V_{g,n}^{WP}$ is equal to $V_{g,n}^{K}$. We return in § 6.4.2 to the fact that the right-hand side of Equation (6.7) can be calculated by the topological recursion.

6.3 Unit balls in measured foliations: the combinatorial case

Following the steps of the previous paragraph, we can also study the volume of the combinatorial unit balls in MF_{Σ} . For $\mathbb{G} \in \mathbb{T}_{\Sigma}^{comb}$, we have a combinatorial length function $\ell_{\mathbb{G}} \colon MF_{\Sigma} \to \mathbb{R}_{\geq 0}$, induced by the one on the larger space of measured foliations MF_{Σ}^{\star} and already used in Section 2.3 to define the combinatorial length of curves.

Definition 6.9. The function $\mathscr{B}_{\Sigma}^{comb} : \mathscr{T}_{\Sigma}^{comb} \to [0, +\infty]$ is defined by

$$\mathscr{B}_{\Sigma}^{\text{comb}}(\mathbb{G}) = \mu_{\text{Th}}(\{\mathscr{F} \in \text{MF}_{\Sigma} \mid \ell_{\mathbb{G}}(\mathscr{F}) \leqslant 1\}).$$
(6.9)

By construction, this function measures the asymptotic growth of the number of multicurves with respect to combinatorial lengths

$$\mathscr{B}_{\Sigma}^{\text{comb}}(\mathbb{G}) = \lim_{t \to +\infty} \frac{\#\{c \in M_{\Sigma} \mid \ell_{\mathbb{G}}(c) \leq t\}}{t^{6g-6+2n}}.$$
(6.10)

In fact, it can be obtained as a limit of the hyperbolic setting via the rescaling flow.

Lemma 6.10. For any connected bordered surface Σ of type (g, n), we have

$$\lim_{\beta \to \infty} \beta^{6g-6+2n} \mathcal{B}_{\Sigma}(\sigma^{\beta}) = \mathcal{B}_{\Sigma}^{\text{comb}}(\text{sp}(\sigma))$$
(6.11)

and the limit is uniform for $sp(\sigma)$ in any thick part of T_{Σ}^{comb} .

Proof. Let $\varepsilon > 0$ and $\mathbb{G} \in \mathbb{T}_{\Sigma}^{comb}$ such that $sys_{\mathbb{G}} \ge \varepsilon$. We denote $\sigma = sp^{-1}(\mathbb{G})$ and $\sigma^{\beta} = sp^{-1}(\beta\mathbb{G})$. Since the length of a multicurve is the sum of lengths of its connected components, Proposition 5.4 implies that for any $c \in M_{\Sigma}$, we have $\beta \ell_{\mathbb{G}}(c) \le \ell_{\sigma^{\beta}}(c) \le (\beta + \kappa_{\varepsilon})\ell_{\mathbb{G}}(c)$. Therefore

$$\left\{ \left. c \in M_{\Sigma} \; \right| \; \ell_{\mathbb{G}}(c) \leqslant \frac{t}{\beta + \kappa_{\varepsilon}} \; \right\} \subseteq \left\{ \left. c \in M_{\Sigma} \; \right| \; \ell_{\sigma^{\beta}}(c) \leqslant t \; \right\} \subseteq \left\{ \left. c \in M_{\Sigma} \; \right| \; \ell_{\sigma^{\beta}}(c) \leqslant \frac{t}{\beta} \; \right\}$$

and thus

$$\frac{\mathscr{B}_{\Sigma}^{\operatorname{comb}}(\mathbb{G})}{(\beta+\kappa_{\varepsilon})^{6g-6+2n}} \leqslant \mathscr{B}_{\Sigma}(\sigma^{\beta}) \leqslant \frac{\mathscr{B}_{\Sigma}^{\operatorname{comb}}(\mathbb{G})}{\beta^{6g-6+2n}}.$$

Multiplying by $\beta^{6g-6+2n}$ and taking the limit $\beta \to \infty$ yields the result.

We shall now establish the combinatorial analogues of Theorem 6.6, Lemma 6.7 and Theorem 6.8.

We first notice that $\mathscr{B}_{\Sigma}^{\text{comb}}$ takes values in \mathbb{R}_+ , as a consequence of Proposition 6.3, and is continuous since the length function $\ell_* \colon \mathfrak{T}_{\Sigma}^{\text{comb}} \to \mathbb{R}_+^{S_{\Sigma}}$ is continuous. The same holds for the counting function

$$\mathcal{N}_{\Sigma}^{\text{comb}}(\mathbb{G}; \mathsf{t}) = \#\{ c \in \mathsf{M}_{\Sigma} \mid \ell_{\mathbb{G}}(c) \leqslant \mathsf{t} \}.$$
(6.12)

The integrability result is more delicate. We derive it from the following two statements: the integration formulae of Lemma 6.11 and Theorem 6.12.

Lemma 6.11. The integral of $\mathbb{N}_{g,n}^{\text{comb}}(\mathbf{G}; t)$ over $\mathbb{M}_{g,n}^{\text{comb}}(L)$ is a polynomial in $t^2, L_1^2, \ldots, L_n^2$. Furthermore, its Laplace transform in the cut-off variable t is given by

$$\int_{\mathbb{R}_{+}} \left(\int_{\mathcal{M}_{g,n}^{\text{comb}}(L)} \mathcal{N}_{g,n}^{\text{comb}}(\mathbf{G};t) \, d\mu_{K}(\mathbf{G}) \right) e^{-st} dt = \frac{1}{s} \sum_{\Gamma \in \mathfrak{S}_{g,n}} \frac{1}{\# \text{Aut}(\Gamma)} \int_{\mathbb{R}_{+}^{E_{\Gamma}}} \prod_{\nu \in V_{\Gamma}} V_{g(\nu),n(\nu)}^{K} \left((\ell_{e})_{e \in E_{\nu}}, (L_{\lambda})_{\lambda \in \Lambda_{\nu}} \right) \prod_{e \in E_{\Gamma}} \frac{\ell_{e} \, d\ell_{e}}{e^{s\ell_{e}} - 1}.$$
(6.13)

Proof. The proof is a variation on the ideas of [33, Section 5], adapted to our notation and the combinatorial setting. Consider the class of a multicurve on Σ under the action of the mapping class group. It is determined by the data (Γ , a) of a stable graph $\Gamma \in \mathcal{G}_{g,n}$, together with an integral tuple $a \in \mathbb{Z}_{+}^{E_{\Gamma}}$ giving the multiplicity of each component. We then consider the frequency of the multicurve in the mapping class orbit (Γ , a) with combinatorial length bounded by t:

$$s_{\Sigma}^{\text{comb}}(\mathbb{G};\Gamma,\mathfrak{a},\mathfrak{t}) = \#\{\mathfrak{c}\in(\Gamma,\mathfrak{a}) \mid \ell_{\mathbb{G}}(\mathfrak{c})\leqslant\mathfrak{t}\}.$$

It is a nonnegative mapping class group invariant function of G. Thus, we can apply the integration formula of Proposition 3.13 to the function $s_{g,n}^{\text{comb}}$ induced on the moduli space $\mathcal{M}_{g,n}^{\text{comb}}(L)$ (see also Remarks 3.15 and 3.16). With the notation of Section 3.3, consider the function $f: \mathbb{R}^n_+ \times \mathbb{R}^{E_{\Gamma}}_+ \to \mathbb{R}$ given by $\Xi_{\Gamma} \equiv 1$ and

$$f(L, \ell) = \begin{cases} 1 & \text{if } \sum_{e \in E_{\Gamma}} a_e \ell_e \leq t, \\ 0 & \text{otherwise.} \end{cases}$$

Then we get $\Xi_{\Sigma}^{f,\Gamma}(\mathbb{G})=\mathbb{N}_{\Sigma}^{comb}(\mathbb{G};t),$ and in particular

$$\int_{\mathcal{M}_{g,n}^{comb}(\mathsf{L})} s_{g,n}^{comb}(\mathbf{G};\Gamma,\mathfrak{a},t) \, d\mu_{\mathsf{K}}(\mathbf{G}) = \frac{1}{\#\operatorname{Aut}(\Gamma,\mathfrak{a})} \int_{\langle\mathfrak{a},\mathfrak{l}\rangle \leqslant \mathfrak{t}} \prod_{\nu \in V_{\Gamma}} V_{g(\nu),n(\nu)}^{\mathsf{K}} \left((\ell_{e})_{e \in \mathsf{E}_{\nu}}, (\mathsf{L}_{\lambda})_{\lambda \in \Lambda_{\nu}} \right) \prod_{e \in \mathsf{E}_{\Gamma}} \ell_{e} \, d\ell_{e},$$

where $\langle a, \ell \rangle = \sum_{e \in E_{\Gamma}} a_e \ell_e$. It is easy to see that such integrals are polynomial in $t^2, L_1^2, \ldots, L_n^2$. Furthermore, notice that for any polynomial $p(\ell)$ in the variables $(\ell_e^2)_{e \in E_{\Gamma}}$, we have

$$\int_{\mathbb{R}_+} \left(\int_{\langle a,\ell \rangle \leqslant t} \mathfrak{p}(\ell) \prod_{e \in \mathsf{E}_{\Gamma}} \ell_e \, d\ell_e \right) e^{-st} dt = \frac{1}{s} \int_{\mathbb{R}_+^{\mathsf{E}_{\Gamma}}} \mathfrak{p}(\ell) \prod_{e \in \mathsf{E}_{\Gamma}} e^{-s\mathfrak{a}_e \ell_e} \ell_e \, d\ell_e.$$

From this, together with the relation

$$\mathcal{N}_{g,n}^{\text{comb}}(\mathbf{G}; t) = \sum_{\Gamma \in \mathfrak{G}_{g,n}} \sum_{a \in \mathbb{Z}_{+}^{\mathbb{E}_{\Gamma}}} \frac{\# \text{Aut}(\Gamma, a)}{\# \text{Aut}(\Gamma)} s_{g,n}^{\text{comb}}(\mathbf{G}; \Gamma, a, t)$$

and Tonelli's theorem, we find

$$\begin{split} &\int_{\mathbb{R}_{+}} \left(\int_{\mathcal{M}_{g,n}^{comb}(\mathbf{L})} \mathcal{N}_{g,n}^{comb}(\mathbf{G}; t) \, d\mu_{K}(\mathbf{G}) \right) e^{-st} dt \\ &= \sum_{\Gamma \in \mathfrak{G}_{g,n}} \frac{1}{\# \operatorname{Aut}(\Gamma)} \sum_{\alpha \in \mathbb{Z}_{+}^{E_{\Gamma}}} \int_{\mathbb{R}_{+}} \left(\int_{\langle \alpha, l \rangle \leqslant t} \prod_{\nu \in V_{\Gamma}} V_{g(\nu), n(\nu)}^{K} \left((\ell_{e})_{e \in E_{\nu}}, (L_{\lambda})_{\lambda \in \Lambda_{\nu}} \right) \prod_{e \in E_{\Gamma}} \ell_{e} \, d\ell_{e} \right) e^{-st} dt \\ &= \frac{1}{s} \sum_{\Gamma \in \mathfrak{G}_{g,n}} \frac{1}{\# \operatorname{Aut}(\Gamma)} \sum_{\alpha \in \mathbb{Z}_{+}^{E_{\Gamma}}} \int_{\mathbb{R}_{+}^{E_{\Gamma}}} \prod_{\nu \in V_{\Gamma}} V_{g(\nu), n(\nu)}^{K} \left((\ell_{e})_{e \in E_{\nu}}, (L_{\lambda})_{\lambda \in \Lambda_{\nu}} \right) \prod_{e \in E_{\Gamma}} e^{-s\alpha_{e}\ell_{e}} \ell_{e} \, d\ell_{e} \\ &= \frac{1}{s} \sum_{\Gamma \in \mathfrak{G}_{g,n}} \frac{1}{\# \operatorname{Aut}(\Gamma)} \int_{\mathbb{R}_{+}^{E_{\Gamma}}} \prod_{\nu \in V_{\Gamma}} V_{g(\nu), n(\nu)}^{K} \left((\ell_{e})_{e \in E_{\nu}}, (L_{\lambda})_{\lambda \in \Lambda_{\nu}} \right) \prod_{e \in E_{\Gamma}} \frac{\ell_{e} \, d\ell_{e}}{e^{s\ell_{e}} - 1}. \end{split}$$

Theorem 6.12. The integral of $\mathcal{B}_{g,n}^{comb}$ is computed as

$$(6g - 6 + 2n)! \int_{\mathcal{M}_{g,n}^{\text{comb}}(L)} \mathcal{B}_{g,n}^{\text{comb}} d\mu_{K}$$

$$= \sum_{\Gamma \in \mathfrak{G}_{g,n}} \frac{1}{\# \operatorname{Aut}(\Gamma)} \int_{\mathbb{R}_{+}^{\mathsf{E}_{\Gamma}}} \prod_{\nu \in V_{\Gamma}} V_{g(\nu),n(\nu)}^{\mathsf{K}} ((\ell_{e})_{e \in \mathsf{E}_{\nu}}, (0_{\lambda})_{\lambda \in \Lambda_{\nu}}) \prod_{e \in \mathsf{E}_{\Gamma}} \frac{\ell_{e} \, d\ell_{e}}{e^{\ell_{e}} - 1}.$$
(6.14)

Proof. From Lemma 6.11, with the change of variable $\ell_e \mapsto \ell_e/s$, we have

$$\begin{split} &\int_{\mathbb{R}_{+}} \left(\int_{\mathcal{M}_{g,n}^{\text{comb}}(\mathbf{L})} \mathcal{N}_{g,n}^{\text{comb}}(\mathbf{G}; t) \, d\mu_{K}(\mathbf{G}) \right) e^{-st} dt \\ &= \frac{1}{s^{6g-5+2n}} \left(\sum_{\Gamma \in \mathcal{G}_{g,n}} \frac{1}{\# \text{Aut}(\Gamma)} \int_{\mathbb{R}_{+}^{E_{\Gamma}}} \prod_{\nu \in V_{\Gamma}} V_{g(\nu),n(\nu)}^{K} \big((\ell_{e})_{e \in E_{\nu}}, (0_{\lambda})_{\lambda \in \Lambda_{\nu}} \big) \prod_{e \in E_{\Gamma}} \frac{\ell_{e} \, d\ell_{e}}{e^{\ell_{e}} - 1} + O(s) \right). \end{split}$$

Here we used the relation $\#E_{\Gamma} + \sum_{\nu \in V_{\Gamma}} (3g(\nu) - 3 + n(\nu)) = 3g - 3 + n$. From properties of the Laplace transform, we find

$$\begin{split} &\int_{\mathbb{R}_{+}} \left(\int_{\mathcal{M}_{g,n}^{\text{comb}}(\mathsf{L})} \frac{\mathcal{N}_{g,n}^{\text{comb}}(\mathsf{G};t)}{t^{6g-6+2n}} \, d\mu_{K}(\mathsf{G}) \right) e^{-st} dt \\ &= \frac{1}{s} \left(\frac{1}{(6g-6+2n)!} \sum_{\Gamma \in \mathcal{G}_{g,n}} \frac{1}{\#\operatorname{Aut}(\Gamma)} \int_{\mathbb{R}_{+}^{\mathsf{E}_{\Gamma}}} \prod_{\nu \in V_{\Gamma}} V_{g(\nu),n(\nu)}^{\mathsf{K}} \big((\ell_{e})_{e \in \mathsf{E}_{\nu}}, (0_{\lambda})_{\lambda \in \Lambda_{\nu}} \big) \prod_{e \in \mathsf{E}_{\Gamma}} \frac{\ell_{e} \, d\ell_{e}}{e^{\ell_{e}} - 1} + O(s) \right), \end{split}$$

and from the final value theorem for the Laplace transform, we have the thesis.

Note that Lemma 6.11 and Theorem 6.12 are valid as equalities in $[0, +\infty]$ as we only used Tonelli's theorem. Since V^K are polynomials, the right-hand sides are actually finite so we can deduce integrability.

Corollary 6.13. The functions $\mathcal{N}_{g,n}^{comb}$ and $\mathcal{B}_{g,n}^{comb}$ are integrable on $\mathcal{M}_{g,n}(L)$ with respect to μ_K .

By comparing Equations (6.8) and (6.14), we also deduce that the integrals of the \mathcal{B} -functions are the same in the combinatorial and in the hyperbolic world and independently of boundary lengths.

Corollary 6.14. The following equality holds

$$\int_{\mathcal{M}_{g,n}(L)} \mathcal{B}_{g,n} \, d\mu_{WP} = \int_{\mathcal{M}_{g,n}^{comb}(L)} \mathcal{B}_{g,n}^{comb} \, d\mu_{K}.$$
(6.15)

Furthermore, both quantities are independent of $L \in \mathbb{R}^{n}_{+}$ *.*

6.4 Geometric recursion for combinatorial length statistics of multicurves

We can rephrase the results of the previous paragraph by saying that we were able to compute combinatorial length statistics of multicurves with respect to the Heaviside function $\theta(t-\ell)$. Indeed the combinatorial Mirzakhani's counting function $\mathcal{N}_{\Sigma}^{\text{comb}}$ can be rewritten as

$$\mathcal{N}_{\Sigma}^{comb}(\mathbb{G};t) = \sum_{c \in \mathcal{M}_{\Sigma}} \theta\big(t - \ell_{\mathbb{G}}(c)\big).$$

In this paragraph, we see the overlap between twisted geometric recursion amplitudes and additive statistics of lengths of multicurves. In the last part of the paragraph, we connect this to the counting of N_{Σ}^{comb} via the Laplace transform.

6.4.1 Geometric recursion and statistics of multicurves

An immediate consequence of Section 4.5 is a GR formula for the combinatorial length statistic of multicurves with exponentially decaying weight: Take $\Xi_{\Sigma} = \Xi_{\Sigma}^{K} \equiv 1$ to be the Kontsevich amplitudes, and $F(\ell; s) = e^{-s\ell}$ for multicurves, or equivalently $f(\ell; s) = \frac{1}{e^{s\ell}-1}$ for primitive multicurves (*cf.* Remark 4.21). This choice is of particular interest, because for every $\mathbb{G} \in \mathcal{T}_{\Sigma}^{comb}$ and every multicurve $c \in M_{\Sigma}$,

$$\prod_{\gamma \in \pi_0(\mathbf{c})} e^{-s\ell_{\mathsf{G}}(\gamma)} = e^{-s\ell_{\mathsf{G}}(\mathbf{c})}.$$

Hence, the specialisation of Theorem 4.20 to such choices gives the following result.

Corollary 6.15. The family of combinatorial GR initial data

$$\begin{aligned} A(L_{1}, L_{2}, L_{3}; s) &= 1, \\ B(L_{1}, L_{2}, \ell; s) &= B^{K}(L_{1}, L_{2}, \ell) + \frac{1}{e^{s\ell} - 1}, \\ C(L_{1}, \ell, \ell'; s) &= C^{K}(L_{1}, \ell, \ell') + \frac{B^{K}(L_{1}, \ell, \ell')}{e^{s\ell} - 1} + \frac{B^{K}(L_{1}, \ell', \ell)}{e^{s\ell'} - 1} + \frac{1}{(e^{s\ell} - 1)(e^{s\ell'} - 1)}, \\ D_{T}(\mathbb{G}; s) &= \sum_{c \in M_{T}} e^{-s\ell_{G}(c)}, \end{aligned}$$
(6.16)

are strongly admissible, and the associated GR amplitudes $\Xi_{\Sigma}(\mathbb{G}; s)$ *satisfy*

$$\Xi_{\Sigma}(\mathbb{G};s) = \sum_{c \in M_{\Sigma}} e^{-s\ell_{\mathbb{G}}(c)}.$$
(6.17)

Furthermore, the TR amplitudes $V\Xi_{g,n}(L; s)$ *satisfy*

$$V\Xi_{g,n}(L;s) = \sum_{\Gamma \in \mathfrak{S}_{g,n}} \frac{1}{\#\operatorname{Aut}(\Gamma)} \int_{\mathbb{R}^{+}_{+}} \prod_{\nu \in V_{\Gamma}} V_{g(\nu),n(\nu)}^{K} \left((\ell_{e})_{e \in E(\nu)}, (L_{\lambda})_{\lambda \in \Lambda(\nu)} \right) \prod_{e \in E_{\Gamma}} \frac{\ell_{e} d\ell_{e}}{e^{s\ell_{e}} - 1}.$$
(6.18)

6.4.2 Connection with the combinatorial Mirzakhani's counting function

Taking the Laplace transform in the cut-off parameter t of the counting function $\mathcal{N}_{\Sigma}^{comb}(\mathbb{G};t)$ and its average $\int_{\mathcal{M}_{g,n}^{comb}(L)} \mathcal{N}_{g,n}^{comb}(\mathbf{G};t) d\mu_{K}(\mathbf{G})$, and comparing them with Equations (6.17) and (6.18), we get the following corollary, which retrieves Lemma 6.11.

Theorem 6.16. The Laplace transform of the counting functions

$$\Xi_{\Sigma}(\mathbb{G};s) = s \int_{\mathbb{R}_{+}} \mathcal{N}_{\Sigma}^{\text{comb}}(\mathbb{G};t) e^{-st} dt$$
(6.19)

are computed by GR, with initial data (6.16). Further, their averages over the combinatorial moduli spaces

$$V\Xi_{g,n}(L;s) = s \int_{\mathbb{R}_+} \left(\int_{\mathcal{M}_{g,n}^{\text{comb}}(L)} \mathcal{N}_{g,n}^{\text{comb}}(\mathbf{G};t) \, d\mu_K(\mathbf{G}) \right) e^{-st} dt$$
(6.20)

are computed by TR, and by the sum over stable graphs (6.13).

In [1] the authors introduced some polynomials $V_{g,n}^{MV}(L)$, indexed by (g, n), and called them Masur–Veech polynomials. They are computed by topological recursion, and their constant terms are the Masur–Veech volumes of the principal strata of the moduli spaces of quadratic differentials (*cf.* next section). Here we propose to upgrade the definition, and consider the 1-parameter family of polynomials

$$V_{g,n}^{MV}(L;s) = \sum_{\Gamma \in \mathcal{G}_{g,n}} \frac{1}{\#Aut(\Gamma)} \int_{\mathbb{R}_{+}^{E_{\Gamma}}} \prod_{\nu \in V_{\Gamma}} V_{g(\nu),n(\nu)}^{K} \left((\ell_{e})_{e \in E(\nu)}, (L_{\lambda})_{\lambda \in \Lambda(\nu)} \right) \prod_{e \in E_{\Gamma}} \frac{\ell_{e} d\ell_{e}}{e^{s\ell_{e}} - 1}, \tag{6.21}$$

which satisfy $V_{g,n}^{MV}(L;s) = s^{-6g+6-2n} V_{g,n}^{MV}(sL)$. The above argument shows that such polynomials coincide with (6.20), providing therefore a more direct geometric interpretation for them.

Remark 6.17. As mentioned previously, the hyperbolic quantities $\mathcal{N}_{g,n}(\sigma; t)$ and their averages can be calculated by GR/TR. Namely, the family of hyperbolic GR initial data

$$\begin{aligned} A(L_{1}, L_{2}, L_{3}; s) &= 1, \\ B(L_{1}, L_{2}, \ell; s) &= B^{M}(L_{1}, L_{2}, \ell) + \frac{1}{e^{s\ell} - 1}, \\ C(L_{1}, \ell, \ell'; s) &= C^{M}(L_{1}, \ell, \ell') + \frac{B^{M}(L_{1}, \ell, \ell')}{e^{s\ell} - 1} + \frac{B^{M}(L_{1}, \ell', \ell)}{e^{s\ell'} - 1} + \frac{1}{(e^{s\ell} - 1)(e^{s\ell'} - 1)'}, \\ D_{T}(\sigma; s) &= \sum_{c \in M_{T}} e^{-s\ell_{\sigma}(c)}, \end{aligned}$$
(6.22)

are strongly admissible, and the associated GR amplitudes $\Omega_{\Sigma}(\sigma; s)$ satisfy

$$\Omega_{\Sigma}(\sigma;s) = \sum_{c \in M_{\Sigma}} e^{-s\ell_{\sigma}(c)} = s \int_{\mathbb{R}_{+}} \mathcal{N}_{\Sigma}(\sigma;t) e^{-st} dt.$$
(6.23)

Furthermore, the TR amplitudes $V\Omega_{g,n}(L;s)$ satisfy

$$V\Omega_{g,n}(L;s) = \sum_{\Gamma \in \mathfrak{S}_{g,n}} \frac{1}{\#\operatorname{Aut}(\Gamma)} \int_{\mathbb{R}_{+}^{E_{\Gamma}}} \prod_{\nu \in V_{\Gamma}} V_{g(\nu),n(\nu)}^{WP} \left((\ell_{e})_{e \in E(\nu)}, (L_{\lambda})_{\lambda \in \Lambda(\nu)} \right) \prod_{e \in E_{\Gamma}} \frac{\ell_{e} d\ell_{e}}{e^{s\ell_{e}} - 1}$$

$$= s \int_{\mathbb{R}_{+}} \left(\int_{\mathcal{M}_{g,n}(L)} \mathfrak{N}_{g,n}(\sigma;t) \, d\mu_{WP}(\sigma) \right) e^{-st} dt.$$
(6.24)

The quantities $V\Omega_{g,n}(L;s)$ should be thought of as the hyperbolic analogue of the 1-parameter family of Masur–Veech polynomials of (6.21). Notice that, as the Weil–Petersson volumes are not homogeneous, the dependence on *s* is more complicated in this case.

6.5 Combinatorial formulae for Masur–Veech volumes

If Σ is a smooth surface of genus g with n > 0 labelled *punctures*, we denote \mathfrak{T}_{Σ} (resp. $\mathfrak{M}_{g,n}$) the corresponding Teichmüller space (resp. moduli space). On the bundle $Q\mathfrak{T}_{\Sigma} \to \mathfrak{T}_{\Sigma}$ of meromorphic quadratic differentials with simple poles at the punctures, Masur [30] and Veech [42] introduced a piecewise linear structure via period coordinates, and an associated measure μ_{MV} which is Mod_{Σ} -invariant. This induces a measure on the total space of the bundle $Q^1\mathfrak{M}_{g,n} \to \mathfrak{M}_{g,n}$ of quadratic differentials of unit area, by taking for a measurable set $Y \subseteq Q^1\mathfrak{M}_{g,n}$

$$\mu_{MV}^{1}(Y) = (12g - 12 + 4n) \, \mu_{MV}(\{ tq \mid t \in (0, \frac{1}{2}) \text{ and } q \in Y \}).$$

The total mass of this measure is finite, and is called *Masur–Veech volume* (for the open stratum of the moduli space of quadratic differentials):

$$MV_{g,n} = \mu_{MV}^1(Q^1\mathfrak{M}_{g,n}).$$
(6.25)

Many works have been recently devoted to the understanding of $MV_{g,n}$ and closely related quantities, motivated (and related to) dynamical questions on the space of measured foliations (and on the Teichmüller space). It is sometimes convenient to change normalisations and divide $MV_{g,n}$ by the constants

$$a_{g,n} = 2^{4g-2+n}(4g-4+n)! \cdot (6g-6+2n)$$
 or $a'_{g,n} = \frac{2^{4g-2+n}(4g-4+n)!}{(6g-7+2n)!}$

Our present work links the combinatorial geometry to $MV_{g,n}$, adding on previously known approaches. They can be summarised in the following way.

Proposition 6.18. The following quantities are all equal:

- (i) $MV_{g,n}$ defined by (6.25).
- (ii) $a'_{q,n}$ times the sum over stable graphs in Theorem 6.8.
- (iii_0) $\mathfrak{a}_{g,n} \int_{\mathfrak{M}_{g,n}} \mathscr{B}_{\Sigma} d\mu_{WP}$.
- (iii) $\mathfrak{a}_{g,n} \int_{\mathcal{M}_{g,n}(L)} \mathscr{B}_{\Sigma} d\mu_{WP}$, for any $L \in \mathbb{R}^{n}_{+}$.
- (iv) $a'_{g,n}$ times the constant term of the hyperbolic TR amplitudes results by twisting the Kontsevich initial data by $f(\ell) = \frac{1}{e^{\ell}-1}$.
- (iv₊) $a'_{g,n} \lim_{s \to 0} s^{6g-5+2n} V\Omega_{g,n}(L;s)$, where $V\Omega_{g,n}(L;s)$ are the 1-parameter family of hyperbolic TR amplitudes for the initial data (6.22).
 - (v) $a_{g,n} \int_{\mathcal{M}_{am}^{comb}(L)} \mathscr{B}_{\Sigma}^{comb} d\mu_{K}$ for any $L \in \mathbb{R}_{+}^{n}$.

- (vi) $a'_{g,n}$ times the constant term of the combinatorial TR amplitudes results by twisting the Kontsevich initial data by $f(\ell) = \frac{1}{e^{\ell} 1}$.
- (vi₊) $a'_{g,n} \lim_{s \to 0} s^{6g-5+2n} V\Xi_{g,n}(L;s)$, where $V\Xi_{g,n}(L;s)$ are the 1-parameter family of combinatorial TR amplitudes for the initial data (6.16).

In [15] one can find two more formulae for $MV_{g,n}$, proved starting from (i): the first one involves intersection theory on $\overline{\mathfrak{M}}_{g,n}$, and the second is another topological recursion distinct from (iv₊) and (vi₊). As they require very different methods (involving algebraic geometry of moduli spaces of curves), we do not discuss them.

There are many different paths and methods to establish all the equalities in Proposition 6.18. To clarify the picture, we list what currently is known:

(i) = (iii₀). Comes from the work of Bonahon [7] and Mirzakhani [34], see e.g. [17] or [1, Lemma 3.1].

(ii) = (iii₀). Proved by Mirzakhani in [33, Theorem 5.3], studying the Laplace transform of the counting function as reported here in Section 6.2. Proved differently in [17] by first relating (iii₀) with the asymptotic number of square-tiled surfaces, and combinatorial techniques to evaluate this number.

 $(iii_0) = (iii)$. Proved in [1, Corollary 3.6].

(ii) = (iii) = (iv). Proved in [1] via the study of (asymptotics of integrals of) statistics of hyperbolic lengths of multicurves and another Laplace transform method.

(iii) = (iv₊). Follows from Mirzakhani's work, together with the specialisation of [2, Theorem 10.1] to the Mirzakhani GR initial data twisted by $f(l; s) = \frac{1}{e^{sl} - 1}$ and the Laplace transform method.

(ii) = (v) = (vi) = (vi_+) are the results of the previous section. Notice that (vi) = (vi_+) is just a consequence of the polynomiality of the 1-parameter family of Masur–Veech polynomials.

One can also give an interpretation of the area Siegel–Veech constant via derivative statistics of combinatorial lengths by adapting [1, Section 4].

There are however some missing paths that would be desirable to complete in the future.

A first question consists in proving (iii) = (v) via the rescaling flow (Section 5). We can certainly write

$$\int_{\mathcal{M}_{g,n}(\beta L)} \mathcal{B}_{g,n} \, d\mu_{WP} = \int_{\mathcal{M}_{g,n}^{comb}(L)} \beta^{6g-6+2n} (\mathcal{B}_{g,n} \circ R_{\beta}) J_{\beta} \, d\mu_{K}, \qquad J_{\beta} = \frac{1}{\beta^{6g-6+2n}} \, \frac{R_{\beta}^* d\mu_{WP}}{d\mu_{K}}$$

We know by Theorem 5.2 due to Mondello that J_{β} converges pointwise to 1, and by Lemma 6.10 that $\beta^{6g-6+2n}(\mathcal{B}_{g,n} \circ R_{\beta})$ converges to $\mathcal{B}_{g,n}^{comb}$ uniformly on compacts. This is however not sufficient, as we would need an effective and integrable enough bound independent of β to conclude that (iii) = (v) by dominated convergence, and such a bound is not currently available. Note that one cannot hope for a bound by constant, because $\mathcal{B}_{\Sigma}^{comb}(\mathbb{G})$ can diverge when $sys_{\mathbb{G}} \to 0$. Getting effective and uniform bounds for the Jacobian J_{β} over $\mathcal{T}_{\Sigma}^{comb}$ is a question of broader interest: it would allow to study the behaviour for large boundary lengths L of the integral on $\mathcal{M}_{q,n}(L)$ of a larger class of functions.

So far the combinatorial expressions (v), (vi) and (vi₊) are only linked to (i) or (iii₀) indirectly, via the sum over stable graphs (ii). We believe it would be interested to find a direct geometric proof of the equality (i) = (v) in a similar way that (i) = (iii₀) was proved in [34], *i.e.* give a direct relation between the Masur-Veech measure on the moduli space of quadratic differentials and the combinatorial enumerative geometry of measured foliations. If such a link could be made precise, one may hope that the role of hyperbolic geometry for the study of flat surfaces could be substituted by combinatorial geometry, which may bring some useful simplifications.

A Topology on the combinatorial Teichmüller spaces

The combinatorial Teichmüller spaces have been endowed with a natural topology associated to a dual construction: the proper arc complex. The main references for this discussion are [29, 35].

Definition A.1. Fix a connected bordered surface Σ . Define the *arc complex* \mathcal{A}_{Σ} to be the simplicial complex whose vertices are isotopy classes α of arcs in Σ with endpoints on the boundary $\partial \Sigma$ which are homotopically non-trivial relatively to $\partial \Sigma$. A simplex in \mathcal{A}_{Σ} is a collection $\alpha = (\alpha_1, \ldots, \alpha_k)$ of distinct vertices such that the arcs α_i admit representatives which do not intersect. The non-proper subcomplex $\mathcal{A}_{\Sigma}^{\infty}$ of \mathcal{A}_{Σ} consists of those simplices α such that one connected component of $\Sigma \setminus \bigcup_{i=1}^{k} \alpha_i$ is not simply connected. The simplices in $\mathcal{A}_{\Sigma} \setminus \mathcal{A}_{\Sigma}^{\infty}$ are called proper.

Consider the geometric realisation spaces $|\mathcal{A}_{\Sigma}|$ and $|\mathcal{A}_{\Sigma}^{\omega}|$. The geometric realisation of a simplicial complex comes with two natural topologies: the coherent topology, namely the finest topology that makes the realisation of all simplicial maps continuous, and the metric topology, for which every k-simplex is isometric to the standard simplex $\Delta^k \subset \mathbb{R}^{k+1}$ and every attachment map is a local isometry. The metric topology on $|\mathcal{A}_{\Sigma}|$ is coarser than the coherent one, but they agree where the complex is locally finite, and this is the case for $|\mathcal{A}_{\Sigma}| \setminus |\mathcal{A}_{\Sigma}^{\infty}|$.



Figure 29: The simplicial structure of $|\mathcal{A}_P|$ and $|\mathcal{A}_P^{\infty}|$ and the set $|\mathcal{A}_P| \setminus |\mathcal{A}_P^{\infty}|$ associated to a pair of pants P. The latter is in natural bijection with the slice { $\mathbb{G} \in \mathbb{T}_P^{comb} \mid \ell_{\mathbb{G}}(\partial_1 P) + \ell_{\mathbb{G}}(\partial_2 P) + \ell_{\mathbb{G}}(\partial_3 P) = 1$ }.

Consider then the topological space $(|\mathcal{A}_{\Sigma}| \setminus |\mathcal{A}_{\Sigma}^{\infty}|) \times \mathbb{R}_+$, called *metrised arc complex*, whose points are of the form $x = \sum_{i=1}^{k} \ell_i \alpha_i$, where $\ell_i > 0$ and $(\alpha_1, \dots, \alpha_k)$ is a proper simplex. There is a bijection between the



Figure 30: Example of duality between a combinatorial structure (red) and an arc system (light blue).

spaces $\mathfrak{T}_{\Sigma}^{\text{comb}}$ and $(|\mathcal{A}_{\Sigma}| \setminus |\mathcal{A}_{\Sigma}^{\infty}|) \times \mathbb{R}_{+}$. Given a combinatorial structure \mathbb{G} , we define for each edge e the dual arc α_{e} connecting the two (possibly equal) boundary components on the two sides of e (see Figure 30). Thus, we define the map $\mathfrak{T}_{\Sigma}^{\text{comb}} \to (|\mathcal{A}_{\Sigma}| \setminus |\mathcal{A}_{\Sigma}^{\infty}|) \times \mathbb{R}_{+}$ by setting $\mathbb{G} \mapsto \sum_{e \in \mathbb{E}_{G}} \ell_{\mathbb{G}}(e) \alpha_{e}$. The map is clearly invertible and we topologise the combinatorial Teichmüller space $\mathfrak{T}_{\Sigma}^{\text{comb}}$ by pulling back the topology of the proper arc complex. Further, through the spine construction, it is possible to prove the following

Theorem A.2. [29, 35] There is an homeomorphism

$$sp: \mathfrak{T}_{\Sigma} \longrightarrow (|\mathcal{A}_{\Sigma}| \setminus |\mathcal{A}_{\Sigma}^{\infty}|) \times \mathbb{R}_{+}, \tag{A.1}$$

equivariant under the action of the mapping class group.

B Examples of cutting, gluing, and combinatorial Fenchel–Nielsen coordinates

CUTTING & GLUING. In order to make the presentation of the cutting and gluing algorithms clearer, we present some neat examples of these procedures. The first two (Figures 31 and 32) cover the case of a sphere Σ with four boundary components, the last two (Figures 33 and 34) cover the case of a torus T with one boundary component.

In all examples, the cutting algorithm is presented in lexicographic order, *i.e.* the images are labelled by (a), (b), (c), *et cetera*, while the gluing algorithm is presented with the same images, but in reversed lexicographic order. The combinatorial structure is depicted in red, the associated measured foliation in blue (only the singular leaves are reported). The cutting curve γ is depicted in green, and it coincide with the curve obtain in the gluing algorithm after identification of two boundary components $\gamma_{-} \sim \gamma_{+}$. The component γ_{-} is always located on the left side of the figure, while γ_{+} is located on the right side. The identification points $p_{\pm} \in \gamma_{\pm}$ are depicted in grey, and no twist is performed (that is, p_{-} is identified with p_{+}). Further, the letters a, b, c, ... are referring to edge lengths of the cutting process, while the letters r, s, t, ... are referring to edge lengths of the gluing process.

COMBINATORIAL FENCHEL–NIELSEN COORDINATES. We present two computations of combinatorial Fenchel–Nielsen coordinates on a torus T with one boundary component, relative to two different cells (Figure 35 and Figure 36). Furthermore, an illustration of the combinatorial Okai formulae (Proposition 5.8) is presented.

In all examples, the combinatorial structure is depicted in red, the associated measured foliation in blue (only the singular leaves are reported), the pants decompositions $\mathcal{P}, \mathcal{P}'$ in green and the collection $\mathcal{S}, \mathcal{S}'$ in yellow.

CUTTING. Consider $\mathbb{G} \in \mathfrak{T}_{\Sigma}^{comb}$ and γ a cutting curve as in Figure 31a. We have

$$\ell_{\mathbb{G}}(\gamma) = a + b + e + f.$$

After cutting, we obtain two pairs of pants P_\pm and two combinatorial structures $\mathbb{G}_\pm\in\mathbb{T}_{\mathsf{P}_\pm}^{comb}.$



(e)

GLUING. Consider two combinatorial structures $\mathbb{G}_{\pm} \in \mathbb{T}_{P_{\pm}}^{comb}$ and two boundary components γ_{\pm} of P_{\pm} of the same length as in Figure 31f. We have

$$\ell_{\mathbb{G}_{-}}(\gamma_{-}) = r + s, \qquad \ell_{\mathbb{G}_{+}}(\gamma_{+}) = u + v.$$

After gluing (with a choice of p_{\pm}), we obtain a sphere Σ with four boundary components and a combinatorial structure $\mathbb{G} \in \mathbb{T}_{\Sigma}^{comb}$.





CUTTING. Consider $\mathbb{G}\in \mathfrak{T}^{comb}_{\Sigma}$ and γ a cutting curve as in Figure 32a. We have

$$\lambda_{\mathbb{G}}(\gamma) = 2a + b + 2c + 2d + e.$$

After cutting, we obtain two pairs of pants P_{\pm} and two combinatorial structures $\mathbb{G}_{\pm} \in \mathbb{T}_{P_{\pm}^{mon}}^{comb}$. From Figure 32c to Figure 32d, a Whitehead move is performed.



GLUING. Consider two combinatorial structures $\mathbb{G}_{\pm} \in \mathbb{T}_{P_{\pm}}^{comb}$ and two boundary components γ_{\pm} of P_{\pm} of the same length as in Figure 32g. We have

$$\ell_{\mathbb{G}_-}(\gamma_-)=\mathfrak{r}+\mathfrak{s}+2\mathfrak{t},\qquad \ell_{\mathbb{G}_+}(\gamma_+)=\mathfrak{u}+\nu+2w.$$

After gluing (with a choice of p_{\pm}), we obtain a sphere Σ with four boundary components and a combinatorial structure $\mathbb{G} \in \mathfrak{I}_{\Sigma^{omb}}^{\mathbb{C}}$. From Figure 32d to Figure 32c, a Whitehead move is performed.

CUTTING. Consider $\mathbb{G} \in \mathfrak{T}_{\Sigma}^{comb}$ and γ a cutting curve as in Figure 31a. We have

$$\ell_{\mathbb{G}}(\gamma) = \mathfrak{a} + \mathfrak{c}.$$

After cutting, we obtain a pairs of pants P and a combinatorial structure $\mathbb{G}'\in\mathbb{T}_P^{comb}.$

GLUING. Consider a combinatorial structure $\mathbb{G}' \in \mathbb{T}_P^{comb}$ and two boundary components γ_{\pm} of P of the same length as in Figure 31f. We have

$$\ell_{\mathbb{G}'}(\gamma_{-}) = r, \qquad \ell_{\mathbb{G}'}(\gamma_{+}) = s.$$

After gluing (with a choice of p_{\pm}), we obtain a torus T with one boundary component and a combinatorial structure $\mathbb{G} \in \mathbb{T}_T^{comb}$.





Figure 33

CUTTING. Consider $\mathbb{G}\in \mathfrak{T}^{comb}_{\Sigma}$ and γ a cutting curve as in Figure 32a. We have

$$\ell_{\mathbb{G}}(\gamma) = 2a + b + c.$$

After cutting, we obtain a pairs of pants P and a combinatorial structure $\mathbb{G}'\in\mathbb{T}_P^{comb}.$

GLUING. Consider a combinatorial structure $\mathbb{G}' \in \mathbb{T}_P^{comb}$ and two boundary components γ_{\pm} of P of the same length as in Figure 32f. We have

$$\ell_{\mathbb{G}'}(\gamma_-) = r + s, \qquad \ell_{\mathbb{G}'}(\gamma_+) = r + t.$$

After gluing (with a choice of p_{\pm}), we obtain a torus T with one boundary component and a combinatorial structure $\mathbb{G}\in\mathbb{T}_{T}^{comb}$.













COMBINATORIAL FENCHEL–NIELSEN COORDINATES. Consider the combinatorial structure $\mathbb{G} \in \mathcal{T}_T^{comb}$ on a torus T with one boundary component as in Figure 35. We have

$$L = 2a + 2b + 2c.$$

Further, the combinatorial Fenchel–Nielsen coordinates with respect the seamed pants decomposition (\mathcal{P}, \mathcal{S}) = (γ, β) of Figure 35a are computed with the help of Figure 35b and are given by

$$\ell = a + c, \qquad \quad \tau = c,$$

while the combinatorial Fenchel–Nielsen coordinates with respect the seamed pants decomposition $(\mathcal{P}', \mathcal{S}') = (\gamma', \beta')$ of Figure 35c are computed with the help of Figure 35d and are given by

$$\ell' = b + c, \qquad \tau' = -c.$$

This is in accordance with the combinatorial Okai formulae, form which

$$\begin{split} \ell' &= |\tau| + \left[\frac{L-2\ell}{2}\right]_+ & \tau' = -\operatorname{sgn}(\tau) \left| \ell - \left[\frac{L-2\ell'(\ell,\tau)}{2}\right]_+ \right| \\ &= |c| + b & = -\left| (a+c) - a \right| \\ &= b+c, & = -c. \end{split}$$



Figure 35

COMBINATORIAL FENCHEL–NIELSEN COORDINATES. Consider the combinatorial structure $\mathbb{G} \in \mathcal{T}_T^{comb}$ on a torus T with one boundary component as in Figure 36. We have

$$L = 2a + 2b + 2c.$$

Further, the combinatorial Fenchel–Nielsen coordinates with respect the seamed pants decomposition (\mathcal{P}, \mathcal{S}) = (γ, β) of Figure 36a are computed with the help of Figure 36b and are given by

$$\ell = 2a + b + c, \qquad \quad \tau = a + b,$$

while the combinatorial Fenchel–Nielsen coordinates with respect the seamed pants decomposition $(\mathcal{P}', \mathcal{S}') = (\gamma', \beta')$ of Figure 36c are computed with the help of Figure 36d and are given by

$$\ell' = a + b, \qquad \tau' = -2a - b.$$

This is in accordance with the combinatorial Okai formulae, form which

$$\begin{split} \ell' &= |\tau| + \left[\frac{L-2\ell}{2}\right]_+ & \tau' = -\operatorname{sgn}(\tau) \left| \ell - \left[\frac{L-2\ell'(\ell,\tau)}{2}\right]_+ \right| \\ &= |a+b|+0 & = -\left| (2a+b+c)-c \right| \\ &= a+b, & = -2a-b. \end{split}$$



Figure 36

C Integral points in the moduli space and factors of 2

Let G be a ribbon graph of type (g, n), and $S \subset E_G$ such that #S = n and the dual graph G_S^* of S (considered as a subgraph of G) is connected and has a single cycle, of odd length. We label the edges so that $S = \{e_1, \ldots, e_n\}$ and $E_G = \{e_1, \ldots, e_{6g-g+3n}\}$. If **G** is a metric structure on G, we call $\ell_i = \ell_G(e_i)$. The multiplicity $A_{i,e} \in \{0, 1, 2\}$ of an edge $e \in E_G$ around the i-th face define an *adjacency matrix* A of size $n \times (6g - 6 + 3n)$. We recall the following well-known fact (see *e.g.* [16, Theorem 2.2]).

Lemma C.1. The restriction \hat{A} of the adjacency matrix R to the n first columns is invertible, and $|\det(\hat{A})| = 2$.

It is useful for § 4.2.2 to characterise the integral points in $\mathcal{M}_{q,n}^{\text{comb}}(L)$ in terms of integral points in $\mathcal{M}_{q,n}^{\text{comb}}$.

Lemma C.2.
$$\ell_{n+1}, \ldots, \ell_{6g-6+3n} \in \mathbb{Z}$$
 and $\sum_{i=1}^{n} L_i$ is even if and only if $\ell_1, \ldots, \ell_{6g-6+3n} \in \mathbb{Z}$.

Proof. It is sufficient to prove the implication, since the converse is obvious. The result is elementary and well-known, but as we did not identify a reference for it, we include a proof. Suppose that $l_{n+1}, \ldots, l_{6g-6+3n}$ are integers and $\sum_{i=1}^{n} L_i = 0 \pmod{2}$. By hypothesis, G_S^* is the union of trees rooted at a cycle with 2p + 3 edges for some $p \ge 0$, and its vertices are labelled from 1 to n. For $e \in S$ and $i \in \{1, \ldots, n\}$, let $d_{e,i}$ be the graph distance in G_S^* between e^* (the dual of e) and the vertex i. If e is adjacent to the face i, then $d_{e,i} = 0$. There is a natural notion of descendent vertices of an edge belonging to a tree of G_S^* , given that the trees are rooted at the cycles of G_S^* . The inverse of \hat{A} can now be made explicit.

• If e* does not belong to the cycle of G^{*}_S

$$\hat{A}_{e,i}^{-1} = \begin{cases} (-1)^{d_{e,i}} & \text{if } i \text{ is a descendent of } e^* \\ 0 & \text{otherwise.} \end{cases}$$

• If *e*^{*} belongs to the cycle of G^{*}_S,

$$\hat{A}_{e,i}^{-1} = \frac{(-1)^{d_{e,i}}}{2}.$$

If e^* does not belong to the cycle of G_S^* , for all $i \in \{1, ..., n\}$ we have $\hat{A}_{e,i}^{-1} \in \{-1, 0, 1\}$, so

$$\ell_{e} = \sum_{i=1}^{n} \hat{A}_{e,i}^{-1} \left(L_{i} - \sum_{k=n+1}^{6g-6+3n} A_{i,e_{k}} \ell_{k} \right)$$

belongs to \mathbb{Z} . If e^* belongs to the cycle of G_S^* , for all $i \in \{1, ..., n\}$ we have $2\hat{A}_{e,i}^{-1} = 1 \pmod{2}$. This implies the following:

$$2\ell_e = \sum_{i=1}^{n} 2\hat{A}_{e,i}^{-1} \left(L_i - \sum_{k=n+1}^{6g-6+3n} A_{i,e_k} \ell_k \right)$$
$$= \sum_{i=1}^{n} L_i - \sum_{k=n+1}^{6g-6+3n} \left(\sum_{i=1}^{n} A_{i,e_k} \right) \ell_k \pmod{2}$$
$$= 0 \pmod{2},$$

where the second to last line comes from the hypothesis $\sum_{i=1}^{n} L_i = 0 \pmod{2}$ and $\sum_{i=1}^{n} A_{i,e} = 2$. The result of this calculation is that $\ell_e \in \mathbb{Z}$.

Index of notations

G, G	Def. 2.2	Ribbon graph, metric ribbon graph
$Aut(G) \supset Aut(G)$	Def. 2.4	Their automorphism groups
$ \mathbf{G} $	Def. 2.10	Geometric realisation of a metric ribbon graph
(G , f), G	Def. 2.11	Combinatorial marking, combinatorial structure
S_{Σ}	Sec. 2.3.1	Set of homotopy classes of simple closed curves on Σ
$S_{\Sigma}^{\circ} \subset S_{\Sigma}$	Sec. 2.3.1	Subset of essential simple closed curves Σ
M_{Σ}, M'_{Σ}	Sec. 2.3.1	Set of multicurves of Σ , set of primitive multicurves on Σ
$\mathcal{P}_{\Sigma,m_0}, \mathcal{P}^{\varnothing}_{\Sigma,m_0}, \mathcal{P}^m_{\Sigma,m_0}$	Def. 2.22	Sets of homotopy classes of embedded pairs of pants in Σ : with m_0 as a boundary, with m_0 and an essential simple curve as boundaries, with m_0 and m as boundaries
$\mathfrak{C}_{\Sigma,\mathfrak{m}_0},\mathfrak{C}^{\varnothing}_{\Sigma,\mathfrak{m}_0},\mathfrak{C}^{\mathfrak{m}}_{\Sigma,\mathfrak{m}_0}$	Def. 2.23	Set of non-trivial homotopy classes of proper embeddings of arcs: with initial point on m_0 , with both endpoints on m_0 , with initial point on m_0 and final point on m
$Q\colon {\mathfrak C}_{\Sigma,{\mathfrak m}_0}\to {\mathfrak P}_{\Sigma,{\mathfrak m}_0}$	Eqn. (2.13)	Map associating to an arc the pair of pants obtained by going around the paths and the boundaries at its endpoints
$\operatorname{Mod}_{\Sigma}, \operatorname{Mod}_{\Sigma}^{\partial}$	Def. 2.8	Mapping class group of Σ , pure mapping class group of Σ
$\mathbb{T}^{comb}_{\Sigma},\mathbb{T}^{comb}_{\Sigma}(L)$	Def. 2.11	Combinatorial Teichmüller space of $\Sigma,$ with fixed boundary lengths $L \in \mathbb{R}^n_+$
$\mathcal{M}_{g,n}^{comb}$, $\mathcal{M}_{g,n}^{comb}(L)$	Def. 2.5	Corresponding moduli spaces
$\mathcal{M}_{g,n}^{\operatorname{comb},\mathbb{Z}}$, $\mathcal{M}_{g,n}^{\operatorname{comb},\mathbb{Z}}(L)$	Def. 4.8	Their subsets of integral points
$\mathfrak{T}_{\Sigma},\mathfrak{T}_{\Sigma}(L)$	Sec. 2.2.2	Teichmüller space of $\Sigma,$ with fixed boundary lengths $L \in \mathbb{R}^n_+$
$\mathfrak{M}_{g,n}, \mathfrak{M}_{g,n}(L)$	Sec. 2.2.2	Corresponding moduli spaces of bordered Riemann surfaces
$\mathfrak{M}_{g,n}, \overline{\mathfrak{M}}_{g,n}$	Sec. 2	Moduli space of genus g smooth complex curves with n marked points, Deligne–Mumford compactification
$\mathfrak{Z}_{\Sigma,G},\mathfrak{Z}_{\Sigma,G}(L)$	Sec. 2	Cell of \mathfrak{T}_Σ coming from a marked ribbon graph $(G,f),$ with fixed boundary lengths $L\in\mathbb{R}^n_+$
$sp\colon \mathfrak{T}_{\Sigma} \to \mathfrak{T}_{\Sigma}^{comb}$	Eqn. (<mark>2.13</mark>)	The spine homeomorphism
Φ_{β} , R_{β}	Defn. 5.1	The rescaling flows
MF^\star_Σ	Def. 2.15	Measured foliations on Σ with all types of behaviour at $\partial\Sigma$
MF_{Σ}	Sec. 6.2	Measured foliations on Σ with points at the boundary of parallel type
$\mathcal{F}_*\colon \mathfrak{T}^{comb}_{\Sigma} \to MF^\star_{\Sigma}$	Eqn. (2.11)	Inclusion of the combinatorial Teichmüller space into the space of measured foliations
$\ell_*\colon \mathfrak{T}_{\Sigma}^{comb} \to \mathbb{R}_+^{S_{\Sigma}}$	Thm. 2.20	Length spectrum map
$\omega_{\mathrm{K}}, \mu_{\mathrm{K}}$	Eqn. (3.2), (3.3)	Kontsevich 2-form and the associated measure
ω_{WP} , μ_{WP}	Sec. 4.2.1	Weil-Petersson symplectic form and the associated measure
Ξ _{g,n}	Sec. 4.1.2	Functions on $\mathfrak{M}^{comb}_{g,n}$ that are output of combinatorial GR
$\Omega_{g,n}$	Sec. 4.1.1	Functions on $\mathfrak{M}_{g,n}$ that are output of hyperbolic GR
$V\Xi_{g,n}(L)$	Eqn. (4.10)	The integrals $\int_{\mathcal{M}_{g,n}^{comb}(L)} \Xi_{g,n} d\mu_{K}$
$N\Xi_{g,n}(L)$	Eqn. (4.13)	The sums $\sum_{\mathbf{G} \in \mathcal{M}_{\mathbf{g},\mathbf{n}}^{\text{comb},\mathbb{Z}}(L)} \frac{\Xi_{\mathbf{g},\mathbf{n}}(\mathbf{G})}{\#\text{Aut}(\mathbf{G})}$
$V\Omega_{g,n}(L)$	Eqn. (4.10)	The integrals $\int_{\mathcal{M}_{g,n}(L)} \Omega_{g,n} d\mu_{WP}$
$\mathscr{B}_{\Sigma}, \mathscr{B}_{\Sigma}^{\mathrm{comb}}$	Def. 6.5, 6.9	Mirzakhani function in the hyperbolic and combinatorial setting
$\mu_{MV}, MV_{g,n}$	Sec. 6.5	The Masur–Veech measure, the Masur–Veech volumes

References

- J.E. Andersen, G. Borot, S. Charbonnier, V. Delecroix, A. Giacchetto, D. Lewański, and C. Wheeler, Topological recursion for Masur–Veech volumes, (2019), math.GT/1905.10352.
- [2] J.E. Andersen, G. Borot, and N. Orantin, *Geometric recursion*, (2017), math.GT/1711.04729.
- [3] J.E. Andersen, L.O. Chekhov, P. Norbury and R.C. Penner. Models of discretized moduli spaces, cohomological field theories. *Journal of Geometry and Physics*, 98.
- [4] J.E. Andersen, L.O. Chekhov, P. Norbury and R.C. Penner. Topological recursion for gaussian means and cohomological field theories. *Theor. Math. Phys.*, 185(3):1685–1717, 2015.
- [5] J. Bennett, D. Cochran, B. Safnuk, and K. Woskoff, *Topological recursion for symplectic volumes of moduli spaces of curves*, Michigan J. Math. **61** (2012), no. 2, 331–358, math.SG/1010.1747.
- [6] M. Bertola and D. Korotkin, Hodge and prym tau functions, Jenkins–Strebel differentials and combinatorial model of M_{g,n}, (2018), math-ph/1804.02495.
- [7] F. Bonahon, Shearing hyperbolic surfaces, bending pleated surfaces and Thurston symplectic form, Ann. Fac. Sci. Toulouse Math. 5 (1996), 233–297.
- [8] G. Borot, Topological recursion and geometry, to appear in Rev. Math. Phys. (2020), math-ph/1705.09986.
- [9] B.H. Bowditch and D.B.A. Epstein, *Natural triangulations associated to a surface*, Topology **27** (1988), no. 1, 91–117.
- [10] P. Buser, Geometry and spectra of compact Riemann surfaces, Springer Science & Business Media, 2010.
- [11] K.M. Chapman, M. Mulase, and B. Safnuk, *The Kontsevich constants for the volume of the moduli of curves and topological recursion*, Commun. Numb. Th. Phys. 5 (2011), no. 3, 643–698, math.AG/1009.2055.
- [12] L.O. Chekhov, Matrix model for discretized moduli space, Journal of Geometry and Physics 12 (1993), no. 3, 153–164, hep-th/9205106.
- [13] L.O. Chekhov and Y. Makeenko, A hint on the external field problem for matrix models, Phys. Lett. B 278 (1992), no. 3, 271–278, hep-th/9202006.
- [14] _____, The multicritical Kontsevich–Penner model, Mod. Phys. Lett. A 7 (1992), no. 14, 1223–1236, hepth/9201033.
- [15] D. Chen, M. Möller, A. Sauvaget, with an appendix by G. Borot, A. Giacchetto, and D. Lewański, Masur–Veech volumes and intersection theory on moduli spaces of quadratic differentials: the strata of odd zeroes and the principal strata, (2020), math.AG/1912.02267.
- [16] F. David and B. Eynard, *Planar maps, circle patterns and 2d gravity*, Ann. Inst. Henri Poincaré Comb. Phys. Interact. 1 (2014), no. 2, 139–183, math.ph/1307.3123.
- [17] V. Delecroix, E. Goujard, P. Zograf, and A. Zorich, Masur–Veech volumes, frequencies of simple closed geodesics and intersection numbers of moduli spaces of curves, (2019), math.GT/1908.08611.
- [18] R. Dijkgraaf, E. Verlinde, and H. Verlinde, Loop equations and Virasoro constraints in non-perturbative two-dimensional quantum gravity, Nucl. Phys. B 348 (1991), no. 3, 435–456.
- [19] N. Do, The asymptotic Weil–Petersson form and intersection theory on $\overline{\mathcal{M}}_{q,n}$, (2010), math.GT/1010.4126.

- [20] B. Eynard and N. Orantin, *Invariants of algebraic curves and topological expansion*, Commun. Numb. Th. Phys. 1 (2007), no. 2, math-ph/0702045.
- [21] B. Farb and D. Margalit, A primer on mapping class groups, Princeton University Press, 2011.
- [22] A. Fathi, F. Laudenbach, and V. Poénaru, *Thurston's work on surfaces*, Mathematical Notes, vol. 48, Princeton University Press, 2012.
- [23] J. Harer, The virtual cohomological dimension of the mapping class group of an orientable surface, Invent. Math. 84 (1986), no. 1, 157–176.
- [24] J. Harer and D.B. Zagier, The Euler characteristic of the moduli space of curves, Invent. Math. 85 (1986), no. 3, 457–485.
- [25] A. Hatcher and W. Thurston, A presentation for the mapping class group of a closed orientable surface, Topology 19 (1980), no. 3, 221–237.
- [26] A.J. Jenkins, On the existence of certain general extremal metrics, Ann. Math. (1957), 440–453.
- [27] R.M. Kaufmann, M. Livernet, and R.C. Penner, Arc operads and arc algebras, Geometry & Topology 7 (2003), 511–568, math.GT/0209132.
- [28] M. Kontsevich, Intersection theory on the moduli space of curves and the matrix Airy function, Commun. Math. Phys. 147 (1992), 1–23.
- [29] F. Luo, On Teichmüller spaces of surfaces with boundary, Duke Math. J. 139 (2007), no. 3, 463–482, math.GT/0601364.
- [30] H. Masur, Interval exchange transformations and measured foliations, Ann. Math. 115 (1982), no. 1, 169–200.
- [31] G. McShane, *Simple geodesics and a series constant over Teichmüller space*, Invent. Math. **132** (1998), no. 3, 607–632.
- [32] M. Mirzakhani, *Simple geodesics and Weil–Petersson volumes of moduli spaces of bordered Riemann surfaces*, Invent. Math. **167** (2007), no. 1, 179–222.
- [33] _____, Growth of the number of simple closed geodesies on hyperbolic surfaces, Ann. Math. 168 (2008), no. 1, 97–125.
- [34] Maryam Mirzakhani, *Ergodic theory of the earthquake flow*, International Mathematics Research Notices **2008** (2008), rnm116.
- [35] G. Mondello, Triangulated Riemann surfaces with boundary and the Weil–Petersson Poisson structure, J. Diff. Geom. 81 (2009), no. 2, 391–436, math.DG/0610698.
- [36] P. Norbury, Counting lattice points in the moduli space of curves, Math. Res. Lett. 17 (2010), 467–481, math.-AG/0801.4590.
- [37] T. Okai, Effects of a change of pants decompositions on their Fenchel–Nielsen coordinates, Kobe J. Math. 10 (1993), no. 2, 215–223.
- [38] R.C. Penner, *The decorated Teichmüller space of punctured surfaces*, Commun. Math. Phys. **113** (1987), no. 2, 299–339.
- [39] _____, Perturbative series and the moduli space of Riemann surfaces, J. Diff. Geom. 27 (1988), no. 1, 35–53.
- [40] I. Rivin, Simple curves on surfaces, Geom. Dedicata 87 (2001), no. 1-3, 345–360, math.GT/9907041.

- [41] K. Strebel, On quadratic differentials with closed trajectories and second order poles, J. Anal. Math. 19 (1967), no. 1, 373–382.
- [42] W.A. Veech, Gauss measures for transformations on the space of interval exchange maps, Ann. Math. 115 (1982), no. 2, 201–242.
- [43] E. Witten, *Two-dimensional gravity and intersection theory on moduli space*, Large N expansion in quantum field theory and statistical physics (Edouard Brézin and Spenta R Wadia, eds.), Surveys in Differential Geometry, vol. 1, International Press of Boston, 1990, pp. 243–310.
- [44] S. Wolpert, On the symplectic geometry of deformations of a hyperbolic surface, Ann. Math. 117 (1983), no. 2, 207–234.
- [45] _____, On the Weil–Petersson geometry of the moduli space of curves, Amer. J. Math. 107 (1985), no. 4, 969–997.
- [46] D. Zvonkine, Strebel differentials on stable curves and Kontsevich's proof of Witten's conjecture, (2002), math.AG/0209071.